

Hydrogeology, hydrochemistry and isotope hydrology of Palm Valley, Central Australia

Author: Wischusen, John David Henry

Publication Date: 2005

DOI: https://doi.org/10.26190/unsworks/17747

License:

https://creativecommons.org/licenses/by-nc-nd/3.0/au/ Link to license to see what you are allowed to do with this resource.

Downloaded from http://hdl.handle.net/1959.4/32925 in https:// unsworks.unsw.edu.au on 2024-05-05

HYDROGEOLOGY, HYDROCHEMISTRY AND ISOTOPE HYDROLOGY OF PALM VALLEY, CENTRAL AUSTRALIA.

John David Henry Wischusen

BSc, Grad. Dip. Appl. Geol., MSc

Submitted as a requirement in full for the degree of

Doctor of Philosophy

at the

THE UNIVERSITY OF NEW SOUTH WALES



SCHOOL OF BIOLOGICAL, EARTH AND ENVIRONMENTAL SCIENCES FACULTY OF SCIENCE

AUGUST 2005

TABLE OF CONTENTS

1.	INT	RODUCTION	1
	1.1.	BACKGROUND	1
	1.2.	OBJECTIVES	4
2	C A I	ADLING AND METHODOLOGY	6
2.	SAN	IPLING AND METHODOLOGY	0
	2.1.	INTRODUCTION	6
	2.2.	NTG DATABASE	7
	2.3.	SURVEY DATA.	8
	2.4.	GEOLOGY	8
	2.5.	PUMP TESTS	10
	2.6. 2.7	HYDROCHEMISTRY	10 11
	2.1. 28	TD A CE EL EMENTS	11 11
	2.8. 2.9	STADIE ISOTODES	. 11
	2.9.	CARBON 13 AND 14	12
	2.10.	CHLORINE 36	13
-			
3.	ENV	/IRONMENTAL SETTING	17
	3.1.	INTRODUCTION	17
	3.2.	CLIMATE	17
	3.2.1	!. Rainfall	17
	3.2.2	2. Temperature and Evaporation	. 23
	3.2.3	3. Summary	23
	3.3.	TOPOGRAPHY	25
	3.4.	LAND SYSTEMS	25
	3.5.	FLORA	27
	3.3.1	Introduction	27
	3.J.4	2. Livisionia mariae	30 21
	3.5.5	5. Genetic evidence of isolation	31
	3.1	5.3.2. Cycads	33
	3.:	5.3.3. Discussion	33
	3.:	5.3.4. Conclusion	34
	3.5.4	4. Water Requirements	. 34
	3.6.	ABORIGINAL CONNECTION	36
4.	HYI	DROGEOLOGY	42
	11	GEOLOGY	10
	4.1.	GEOLOGY	42 17
	4.1.1	Manning	42 12
	413	Stratioranhy	+2
	4.1.4	1 Lithology	49
	4.1.4	5. Structure	. 49
	4.2.	BACKGROUND - AOUIFER CHARACTERISTICS	52
	4.2.1	I. Introduction	52
	4.2.2	2. Fractured aquifers	53
	4.3.	HERMANNSBURG SANDSTONE - REGIONAL DATA	57
	4.3.1	Introduction	. 57
	4.3.2	2. Test pumping	. 58
	4.3.3	3. Previous assessments	. 61

4.3.4.	Hermannsburg Sandstone - aquifer characteristics	
4.3.5.	Groundwater flow	
4.4. L	OCAL HYDROGEOLOGY	
4.4.1.	Previous work	
4.4.2.	Groundwater features	
4.4.2	2.1. Introduction	
4.4.2	2.2. Discharge sites	
4.4.3.	Aquijer iype	د /
4.3. 3	UMMARY	
5. HYDE	ROCHEMISTRY	
5.1. In	NTRODUCTION	
5.2. P	REVIOUS WORK	
5.3. P	ALM VALLEY AREA	
5.3.1.	Major ion chemistry	
5.3.2.	Hydrochemical processes	
5.3.2	2.1. Saturation Indices	
5.3.2	2.2. Dissolution / Precipitation reactions	
Э.4. К	EGIONAL DATA SET	
5.4.1.	Description	۷۶ م
5.4.2.	Major Ion Chemistry	
J.4.5. 5 A 3	<i>Hydrochemical processes</i>	
543	3.2 Factors affecting pH	95 96
5.4.3	3.3. Redox potential	
5.4.3	3.4. Saturation indices	
5.4.3	3.5. Sulphate	
5.4.3	3.6. Ion exchange	
5.4.3	3.7. Discussion	
5.5. C	ONCLUSION	
6. STAB	LE ISOTOPE AND RADIOCARBON HYDROLOGY	
61 0	WYCEN 19 AND DELITEDHIM	107
611	Introduction	
6.1.2	Posulte	
613	Stable isotones: a proxy for groundwater age?	
62 P	Stable isolopes. a proxy for groundwater age:	
6.2. K	Rackaround	
622	Carbon 13	
623	Carrections	
624	Results - Radiocarbon age	
0.2.4.	Kesuiis - Kaalocarbon age	
7. CHLC	DRINE-36	
7.1. C		
7.1.1.	Introduction	
7.1.2.	Measurement	
7.1.3.	³⁶ Cl production	
7.1.3	3.1. Atmospheric fallout	
7.1.3	3.2. Changes in fallout rate in the past	
7.1.3	3.3. Epigene production	
7.1.3	3.4. Hypogene production	
7.1.3	1.3.5.1 Romb Peak	143 1/12
7. 7	1.3.5.2. Recirculation of bomb ³⁶ Cl	
7.	1.3.5.3. Localised anthropogenic ³⁶ Cl sources	
7.1.4.	Inputs	
7.1.5	Conclusions	
7.2. R	LESULTS	
7.2.1.	Introduction	
7.2.2.	³⁶ Cl analyses	
7.2.3	Factors affecting the use of ^{36}Cl as a chronometer.	
7.2.3	3.1. Introduction	

7.2.3.	2. Input ratio	
7.2.3.	 Past changes of ³⁶Cl fallout rate 	
7.2.3.	4. Epigene Production	
7.2.3.	5. Hypogene production	
7.2.3.	6. Anthropogenic Sources	
7.2.3.	7. Addition of "dead" chloride	
7.2.3.	8. Cl/Br ratios	
7.2.3.	9. Complementary data	
7.3. Co)NCLUSION	
8. RELEV	VANCE OF THIS STUDY	
8.1. BA	ACKGROUND	
8.2. Re	EVIEW OF GROUNDWATER ³⁶ CL DECAY STUDIES	
8.2.1.	Introduction	
8.2.2.	Renowned ³⁶ Cl decav groundwater studies	
8.2.2.	1. Great Artesian Basin	
8.2.2.	2. Milk River	
8.2.2.	3. Magadi-Natron Basin	
8.2.2.	4. Mazowsze Basin	
8.2.2.	5. Murray Basin	
8.2.3.	Other Studies that attempt ³⁶ Cl decay dating	
8.2.3.	1. Introduction	
8.2.3.	2. International studies	
8.2.3.	3. Australian Studies	
8.2	2.3.3.1. Attempted ³⁶ Cl decay dating	
8.2	4.3.3.2. Unsuitable for ³⁶ Cl decay dating	
8.2.4.	Discussion	
8.2.5.	Conclusion	
8.3. Co	MPARISON TO OTHER STUDIES	
8.3.1.	Discussion	
8.3.2.	Conclusion	
9. DISCU	SSION	
91 FL	OW	193
911	Flow threshold?	193
012	Diffusion from flow in fractured aquifors	195 104
9.1.2.	Elow model	104
9.1.3.	1 Introduction	194 104
913	 Toth unit basin model 	
913	3 Concentual model	193
9.2 Su	ISTAINABII ITV	190
921	Introduction	100
0.2.1.	Palaco climate considerations	202
0.2.2.	Discussion	202 202
9.2.3.	Discussion	
9.2.4.		
10. CON	CLUSION	
10.1.	MAIN CONCLUSIONS	
10.2.	RECOMMENDATIONS	
11. REF	ERENCES	

FIGURES

FIGURE 1-1 LOCATION MAP2
FIGURE 2-1 MONTHLY RAINFALL AT PALM VALLEY 1993-199416
FIGURE 3-1 THE AUSTRALIAN ARID ZONE AND RAINFALL SEASONALITY
FIGURE 3-2 MEAN MONTHLY RAIN (MM) FROM PALM VALLEY
FIGURE 3-3 YEARLY RAINFALL TOTALS AT PALM VALLEY
Figure 3-4 Mean monthly rainfall totals at Hermannsburg from 100 yrs of record
FIGURE 3-5 MONTHLY RAINFALL RECORDS AND VALUE OF THE SOUTHERN OSCILLATION INDEX (SOI) COMPARED FOR PALM VALLEY 1998–2001
FIGURE 3-6 MONTHLY RAIN RECORDS AND VALUE OF THE SOUTHERN OSCILLATION INDEX (SOI) COMPARED FOR PALM VALLEY
FIGURE 3-7 MEAN MONTHLY MAXIMUM AND MINIUM DAILY TEMPERATURES (^o C) AT ALICE SPRINGS AIRPORT. (BOM STATION 015590)
FIGURE 3-8 MEAN MONTHLY CLASS A PAN EVAPORATION WITH MEAN AND 9TH DECILE MONTHLY RAINFALL TOTALS (ALICE SPRINGS AIRPORT BOM STATION 015590) 24
FIGURE 3-9 TOPOGRAPHY OF THE AREA SURROUNDING PALM VALLEY
FIGURE 3-10 SCHEMATIC OF KRICHAUFF LAND SYSTEM
FIGURE 3-11 SCHEMATIC OF DISTRIBUTION OF ENDEMIC PLANT SPECIES AROUND PALM VALLEY
FIGURE 3-12 PRELIMINARY UNPUBLISHED PHYLOGENETIC TREE OF SOME AUSTRALIAN LIVISTONA SPECIES
FIGURE 3-13 DISTRIBUTION OF CYCADS IN THE AUSTRALIAN REGION
FIGURE 3-14 LOCATION AND ARANDA NAME OF ABORIGINAL DREAMING SITES NEAR PALM VALLEY
FIGURE 4-1THE AMADEUS BASIN, DISTRIBUTION OF RELEVANT LITHOLOGICAL UNITS43
FIGURE 4-2 GEOLOGY NEAR PALM VALLEY
FIGURE 4-3 MAP OF GEOLOGICAL STRUCTURE NEAR PALM VALLEY

FIGURE 4-4 STRATIGRAPHY OF THE NORTH EAST AND SOUTH CENTRAL AMADEUS BASIN. 48
FIGURE 4-5 AERIAL PHOTOGRAPH LINEAMENT INTERPRETATION, HERMANNSBURG SANDSTONE
FIGURE 4-6 LOCATION OF BORES DRILLED INTO THE HERMANNSBURG SANDSTONE AT SIMILAR MID AMADEUS BASIN POSITION TO PALM VALLEY
FIGURE 4-7 REGIONAL POTENTIOMETRIC SURFACE – HERMANNSBURG SANDSTONE 65
FIGURE 4-8 LOCATION OF DRILL HOLES IN THE VICINITY OF PALM CREEK
FIGURE 4-9 POTENTIOMETRIC SURFACE – HERMANNSBURG SANDSTONE
FIGURE 5-1 PIPER PLOT OF MAJOR CATION CHEMISTRY
FIGURE 5-2 (A-D), ASPECTS OF GROUNDWATER CHEMISTRY SAMPLED FOR THIS STUDY IN THE PALM VALLEY
FIGURE 5-3 PIPER PLOT OF MAJOR CATION CHEMISTRY FROM HERMANNSBURG SANDSTONE
FIGURE 5-4 (A-F) HERMANNSBURG SANDSTONE (TABLE 5.2) MAJOR ION VERSUS CHLORIDE PLOTS
FIGURE 5-5 (A-D) ASPECTS OF GROUNDWATER CHEMISTRY FROM THE REGIONAL HERMANNSBURG
FIGURE 5-6 (A-E) PLOTS HIGHLIGHTING SOME POSSIBLE HYDROCHEMICAL PROCESSES IN THE HERMANNSBURG SANDSTONE
FIGURE 5-7 (A-G) SATURATION INDICES VERSUS TDS
FIGURE 6-1 STABLE ISOTOPE DATA – GROUNDWATER SAMPLES, PALM VALLEY109
Figure 6-2 Relationship between $\delta^2 H$ and $\delta^{18} O$ for samples from the regional Hermannsburg Sandstone data set
FIGURE 6-3 RELATIONSHIP BETWEEN STABLE WATER ISOTOPES AND CHLORIDE
Figure 6-4 Relationship between $\delta^{13}C$ and dissolved inorganic carbon DIC. 120
Figure 7-1 A schematic of 36 CL in the hydrological cycle
Figure 7-2 The affect of various processes on 36 CL and CL ⁻ levels
FIGURE 7-3 RELATIVE PRODUCTION RATE OF ATMOSPHERIC ³⁶ CL OVER THE LAST 80 KA.

Figure 7-4 Variation of Earths Magnetic field strength over last 800 ka... 137

FIGURE 7-5 PLOT OF ³⁶ CL/CL VERSUS CHLORIDE FOR HERMANNSBURG SANDSTONE, PALM VALLEY
FIGURE 7-6 PLOT OF ³⁶ CL/CL VERSUS ³⁶ CL ATOMS FOR HERMANNSBURG SANDSTONE, PALM VALLEY
FIGURE 7-7 INLAND AUSTRALIAN ATMOSPHERIC CL/BR RATIOS
Figure 8-1 . $^{36}\rm{CL/CL}$ versus chloride content for the studies that have been identified as attempting $^{36}\rm{CL}$ decay dating of groundwater
FIGURE 8-2 THE SAME DATA SET AS FIG. 8.1 WITH ³⁶ CL/CL VERSUS THE LOG OF CL ⁻ CONCENTRATION
FIGURE 8-3 . ³⁶ CL/CL VERSUS THE LOG OF CL- CONCENTRATION FOR ALL OF THE AUSTRALIAN ³⁶ CL GROUNDWATER STUDIES
FIGURE 8-4 ³⁶ CL/CL VERSUS THE LOG OF CL ⁻ CONCENTRATION FOR STUDIES ATTEMPTING ³⁶ CL DECAY DATING OF GROUNDWATER (I.E. FIG 8.2) COMPARED TO THE RESULTS OF THIS STUDY AT PALM VALLEY
FIGURE 8-5 ³⁶ CL/CL VERSUS THE LOG OF CL ⁻ CONCENTRATION FOR THE AUSTRALIAN ³⁶ CL GROUNDWATER STUDIES COMPARED WITH THE RESULTS FROM PALM VALLEY.
FIGURE 9-1 COMPOSITE FLOW PATTERNS, UNIT BASIN FLOW SYSTEM WITH SIGNIFICANT TOPOGRAPHY
FIGURE 9-2 SCHEMATIC CONCEPTUAL FLOW MODEL FOR THE PALM VALLEY AREA200
FIGURE 9-3 REGIONAL TOPOGRAPHY OF THE PALM VALLEY AREA WITH VARIABLE FLOW PATH LENGTHS

TABLES

TABLE 2-1 Survey results Palm Valley area 9
TABLE 4-1 LOCATION AND AQUIFER CHARACTERISTICS OF BORES DRILLED IN THE HERMANNSBURG SANDSTONE
TABLE 4-2 PALM VALLEY AREA DRILLING DATA 71
TABLE 5-1 GROUNDWATER CHEMICAL DATA, PALM VALLEY STUDY
TABLE 5-2 GROUNDWATER CHEMICAL DATA, HERMANNSBURG SANDSTONE REGIONAL DATA SET. 90
TABLE 5-3 SELECTED SATURATION INDICES AND AVAILABLE δ^{18} O and δ^{2} H values, Hermannsburg Sandstone regional data set
TABLE 6-1 RADIOCARBON RESULTS AND CORRECTIONS, PALM VALLEY STUDY. 122
TABLE 7-1 ISOTOPE DATA, PALM VALLEY STUDY. 153
TABLE 7-2 TRACE ELEMENT CHEMISTRY FROM 3 PALM VALLEY BORES. 159
TABLE 7-3 KEYWOOD'S (1995) CENTRAL AUSTRALIAN ATMOSPHERIC CL/BR (MASS). 162
TABLE 8-1 36 Cl groundwater studies discussed in the text

PLATES

PLATE 3-1 LIVISTONA MARIAE, PALM CREEK, PALM VALLEY
PLATE 3-2 PALM VALLEY; THE MAIN STAND OF PALMS IN PALM CREEK
PLATE 3-3 PALM PLAIN AS SEEN LOOKING SOUTHWEST FROM THE KRICHAUFF RANGE41
PLATE 4-1 BEDDING RELATED SEEPS, HERMANNSBURG SANDSTONE
PLATE 4-2 JOINT RELATED SEEPS, HERMANNSBURG SANDSTONE
PLATE 4-3 CALCRETE DEPOSIT ASSOCIATED WITH A SEEP SITE77
PLATE 4-4 CALCRETE DEPOSIT ABOVE CURRENT POTENTIOMETRIC SURFACE
PLATE 4-5 WEST SPRING78
PLATE 4-6 ROAD SPRING
PLATE 4-7 OASIS SPRING (BOTTOM WATER), PALM VALLEY
PLATE 4-8 OASIS SPRING, MAIN AREA

ABSTRACT

The Palm Valley oasis in arid central Australia is characterised by stands of palm trees (Livistona mariae). How these unique plants, separated by nearly a 1000 kilometres of arid country from their nearest relatives persist, has long fascinated visitors. Defining the hydrogeology of the Hermannsburg Sandstone, a regionally extensive and thick Devonian sequence of the Amadeus Basin that underlies Palm Valley, is the major thrust of investigation. Appraisal of drilling data shows this aquifer to be a dual porosity fractured rock aquifer which, on a regional scale, behaves as a low permeability, hydraulically continuous resource. Groundwater is low salinity (TDS <1000 mg/L) and bicarbonate rich. Slight variations in cation chemistry indicate different flow paths with separate geochemical histories have been sampled. Stable isotope (δ^2 H, δ^{18} O) results from Palm Valley show groundwater to have a uniform composition that plots on or near a local meteoric water line. Radiocarbon results are observed to vary from effectively dead (< 4%) to 87 % modern carbon. To resolve groundwater age beyond the radiocarbon window the long lived radioisotope ³⁶Cl was also used. Ratios of 36 Cl/Cl range from 130 to 290 x 10⁻¹⁵. In this region atmospheric 36 Cl/Cl ratio is around 300×10^{-15} . Thus an age range of around 300 ka is indicated if, as is apparent, radioactive decay is the only significant cause of 36 Cl/Cl variation within the aquifer. A review of previous, often controversial, ³⁶Cl decay studies shows results are usually ambiguous due to lack of certainty when factoring subsurface Cl⁻ addition into decay calculations. Apparently, due to the thickness of the Hermannsburg Sandstone, no subsurface sources of Cl such as aquitards or halites, are encountered along groundwater flow paths, hence the clear ³⁶Cl decay trend seen. The classic homogenous aquifer with varying surface topography, the "Toth" flow model, is the simplest conceptual model that need be invoked to explain these isotope data. Complexities, associated with local topography flow cells superimposed on the regional gradient, signify groundwater with markedly different flow path lengths has been sampled. The long travel times (> 100 ka) indicate groundwater discharge would endure through arid phases associated with Quaternary climate oscillations. Such a flow system can explain the persistence of this arid zone groundwater-dependent ecosystem and highlight the possibility that Palm Valley has acted as a flora refuge since at least the mid-Pleistocene.

ORIGINALITY STATEMENT

'I hereby declare that this submission is my own work and to the best of my knowledge it contains no materials previously published or written by another person, or substantial proportions of material which have been accepted for the award of any other degree or diploma at UNSW or any other educational institution, except where due acknowledgement is made in the thesis. Any contribution made to the research by others, with whom I have worked at UNSW or elsewhere, is explicitly acknowledged in the thesis. I also declare that the intellectual content of this thesis is the product of my own work, except to the extent that assistance from others in the project's design and conception or in style, presentation and linguistic expression is acknowledged.'

Signed.....

COPYRIGHT STATEMENT

'I hereby grant the University of New South Wales or its agents the right to archive and to make available my thesis or dissertation in whole or part in the University libraries in all forms of media, now or here after known, subject to the provisions of the Copyright Act 1968. I retain all proprietary rights, such as patent rights. I also retain the right to use in future works (such as articles or books) all or part of this thesis or dissertation. I also authorise University Microfilms to use the 350 word abstract of my thesis in Dissertation Abstract International. I have either used no substantial portions of copyright material in my thesis or I have obtained permission to use copyright material; where permission has not been granted I have applied/will apply for a partial restriction of the digital copy of my thesis or dissertation.'

Signed Date

AUTHENTICITY STATEMENT

'I certify that the Library deposit digital copy is a direct equivalent of the final officially approved version of my thesis. No emendation of content has occurred and if there are any minor variations in formatting, they are the result of the conversion to digital format.'

Signed Date.....

ASSOCIATED PUBLICATION

Wischusen, J.D.H., Fifield, L.K. and Cresswell, R.G., 2004. Hydrogeology of Palm Valley, central Australia; a Pleistocene flora refuge? Journal of Hydrology 293, 20–46

ACKNOWLEDGEMENTS

This study has had the benefit of help from many people. I acknowledge the support of my partner Megan Halliday who gave birth to my two daughters, Lily and Thea in the second and third year of enrolment, an unexpected but welcome distraction. I gratefully acknowledge the financial support of a University of New South Wales Postgraduate Scholarship. I thank my student colleagues at UNSW: Maria Dubikova, Sarah Groves, Lange Jorstad, Wendy McLean, Karina Morgan, Jessica Northey, Beth O'Shea, Rachel Sijgers. Their enthusiasm and optimism was infectious and the camaraderie invaluable. The legendary Kaza O'Shea chocolate cakes and vibrant good cheer of the BEES department also helped. I thank Jerzy Jankowski for his faith in letting me blow in off the street and do my own thing. I thank Jim Jago, Ron Beckett and Don Armstrong for helping my scholarship application. Andy Herczeg, Fred Leaney, Glen Harrington of CSIRO are thanked for early discussion of results and providing useful references. This work initially went off on a few tangents and I am grateful for the assistance in GIS data and modifying software code provided by my former AGSO colleagues Robyn Gallagher, Stephen Hostetler and Gayle Young. Gerry Jacobson and Eleanor Astolfi of AGSO helped prepare the ³⁶Cl samples. The constant encouragement of Dennis Matthews, who let me roam in the Finke Gorge National Park and provided a wealth of local knowledge, is acknowledged. I have benefited from the cooperation of Magellan Petroleum staff in all my work over the years in the Palm Valley region. My colleagues at the Northern Territory Water Resources Branch are thanked for their assistance during the early fieldwork and in subsequently providing access to government data. In particular I acknowledge the help of Scotty Balfour, Alan Russ, Bob Paul, Bob Read, Ian Matthews, Peter Jolly, Jeff Fong, Phil Turner, Col Garner, Virginia Garner, Avis Wiegele, John Childs and Peter MacDonald. Keith Fifield at ANU undertook the ³⁶Cl analyses and encouraged an academic outcome from this project. My friends Julian Reid, Deryn Alpers, Julia Burke, Katrina Budrikis are thanked for encouragement and helping the later fieldwork. Prior to this study my brother Peter, sisters Jane and Mary, mother Sheila, and friends, all helped me back to health. Michael Green is thanked for the medical expertise in keeping me here; I am also grateful to Eddy Kondys and Francis O'Donnell for helping with finance and solving bureaucratic problems, John Owen many thanks for salvaging the computer crashes.

1. Introduction

1.1. Background

This thesis examines the hydrogeology of Palm Valley in central Australia. Palm Valley lies within the Western MacDonnell Ranges of the Northern Territory and is the major natural attraction of the Finke Gorge National Park, central Australia (Fig.1.1). In this region mean annual rainfall is around 250 mm/yr and potential evaporation is around 2800 mm/yr; hence the evaporation to precipitation ratio is over 10. The valley is named after the Red Cabbage Palms (*Livistona mariae*) that grow in this area. The nearest similar species are found around 1000 km away in the tropical regions of north Australia (e.g. at Mataranka, Lawn Hill shown on Fig. 1.1). The incongruity of this tropical looking palm growing in an arid environment is a source of fascination to visitors. The palms are described by Latz (1975) and are commonly assumed to represent a relict stand that has survived since a wetter climate allowed a greater distribution. Preliminary results of a comparative DNA gene sequencing study of the morphologically very similar palms at Palm Valley and Mataranka to the north, had indicated significant genetic divergence, a potential sign of long time separation between populations, (Bowman, pers. com 2002).

The onset of aridity in central Australia is dated by Chen and Barton (1991) at around 0.9 to 1.6 Ma. Since around 0.7 Ma world climates have been controlled by 100 ka glacial cycles (Colman et al., 1995). In Australia there is evidence that Pleistocene dry and wet episodes have oscillated, broadly following worldwide glacial and interglacial events respectively (Nanson et al., 1992). In central Australia up to 85% of a Quaternary glacial period is thought to comprise of dry (glacial) to hyper arid (glacial maximum) conditions with the remainder of cycle wet (pluvial) (White, 1994). The last hyper arid (glacial maximum) period in central Australia is thought to have occurred around 20 ka. The last full pluvial period in central Australia was around 100 ka (Kershaw and Nanson, 1993) though there is some evidence that wetter conditions prevailed around 35 – 50 ka (e.g. Croke et al., 1996).







Palm Valley is a part of the Finke Gorge National Park, central Australia, and lies about 130 km west of Alice Springs. Palm Valley, the main tourist attraction of this park, is a stand of *Livistona mariae* palm trees found along the Palm Creek, in the northwestern extension of the park. The nearest other populations of Livistona sp. in Australia are closer to the coast at the Hamersley Range, Mataranka and Lawn Hill locations marked. Given this likelihood of protracted arid conditions in central Australia during the Pleistocene and the fact that many other rare flora (Latz, 1996) and fauna (e.g. Davis, 1995, 1997) are also found at Palm Valley, it is interesting to speculate on how long these palms have been separated from other Livistona species in tropical Australia. It may in fact be that Palm Valley represents a refuge that has functioned since the onset of aridity in central Australia.

Because of the modern day and past aridity of climate in this region, it has long been suspected that apart from a sheltered habitat, tapping shallow groundwater is how these palms survive dry times. In this study, the feasibility of groundwater sustaining the flora refuge at Palm Valley is examined. Because significant recharge may have last occurred during Pleistocene pluvial events, some of this study is orientated to dating groundwater. As hydrogeological and radiocarbon data indicated that very old (> 30 ka) groundwater may exist in this aquifer the ³⁶Cl decay method was applied. This method, which allows dating well beyond the 30 ka range of the more commonly applied ${}^{14}C$ decay method to ages of around 1 Ma (Phillips, 2000), has been controversial in some past studies because results are ambiguous. This is chiefly because inconsistent chloride concentration levels along perceived flow paths (possible evidence of Cl⁻ addition post recharge) complicate the interpretation of ³⁶Cl decay results. This complication is due to an equivalence phenomenon i.e. there is no way of proving that chloride has been added slowly and consistently along a flow path or near instantaneously either immediately post recharge or at some other time during flow (Andrews and Fontes, 1992). Thus, while ³⁶Cl decay dating offers a valuable tool for assessing groundwater systems with its long time frame resolution capability, on the other hand, it is also a technique which has resulted in debate about its veracity. Given the controversy regarding this technique I have devoted a significant portion of the overall effort to reviewing previous ³⁶Cl decay groundwater studies and recent developments in the literature that may affect the application of the method

The use of ³⁶Cl is potentially significant as any new insights gained on the application of this controversial technique may help future groundwater studies. The study of groundwater systems is an important endeavour in central Australia as in this region all industries, communities and presumably many ecosystems are reliant on groundwater. The characteristic high interannular variability of rainfall along with a lack of

hydrological data due to sparse populations, makes the study of groundwater in arid regions difficult (Lerner et al., 1990). Thus, in the central Australian region, reliance on annual or seasonal precipitation is not an option. Indeed many groundwater supplies are thought to tap old water and to even be non renewable in that the water stored in some aquifers is believed to be replenished only during past wetter climate regimes (e.g. the Alice Springs town water supply; Jolly et al. 1991; Jacobson and Lau, 1991). Consequently, this study of an apparently long lived groundwater dependent ecosystem, is relevant to the understanding and managing of groundwater in central Australia and potentially also other arid zone groundwater systems elsewhere.

1.2. Objectives

Thus the principal research objectives of this study are:

1) to evaluate the hydrogeology, hydrochemistry and isotope chemistry of the aquifer underlying Palm Valley.

2) to review the background theory and previous applications of ³⁶Cl decay dating of groundwater systems.

3) to apply the ³⁶Cl decay groundwater dating method at Palm Valley and critically assess results in the context of objectives 1) and 2).

4) to determine the feasibility that groundwater maintains the flora refuge at Palm Valley

These objectives are addressed in the following chapters. Chapter 2 outlines the methodologies adopted for this study. Chapter 3 characterises the environmental setting of Palm Valley. Chapter 4 defines the geology and aquifer characteristics of the Hermannsburg Sandstone at and near Palm Valley. Chapter 5 outlines the hydrochemistry of the Hermannsburg Sandstone. Chapter 6 examines the stable isotope and radiocarbon hydrology at Palm Valley. Chapter 7 reviews the background and theory of ³⁶Cl decay dating and examines the results from Palm Valley against some

critical factors known to complicate ³⁶Cl decay age interpretations. Chapter 8 looks at the relevance of this study by way of a review of previous ³⁶Cl decay groundwater studies and comparison of results from Palm Valley. Chapter 9 discusses ramifications of all these results and develops a hydrogeological conceptual model of the groundwater flow system at Palm Valley. Chapter 10 assembles all the key results and provides a summary of the evidence that support the hypothesis that groundwater can, and has, sustained the flora refuge at Palm Valley for a significant period.

2. Sampling and methodology

2.1. Introduction

This study started as a request by the National Park rangers for the then Water Resources Branch of the Northern Territory Government to examine the water supply sustainability in the Palm Valley portion of the Finke Gorge National Park in 1993. I was then working as a hydrogeologist for the Water Resources Branch and this job was assigned to me. Due to the complexity of this task and competing work interests this job carried on for many years unfinished. So to finish this job and put the study within a scientific context I made the project the basis of this thesis. Given this origin some aspects of the methods used for this study are slightly different from some University initiated projects. The main difference is that as a government worker when much of the data was gathered I had the luxury of a budget, in house labour and laboratories to undertake many of the analytical tasks associated with this project. While this has meant I have missed the experience of preparing my own samples and running analytical apparatus for chemical and isotope data, the use of professional laboratory services for this work at least ensures the accuracy of chemical and isotope analyses reported here. Similarly my role as a Northern Territory Government hydrogeologist equipped me with a familiarity with the database; this greatly assisted me in obtaining and analysing data from other bores drilled in the Hermannsburg Sandstone in the vicinity of Palm Valley. This was crucial for putting data from Palm Valley into a regional context. This comparison with a compilation of regional data from the Hermannsburg Sandstone also adds to the integrity of data from Palm Valley.

The complexity and scale of the physical and biological environments encountered by water as it moves through the hydrologic cycle indicate that application of one scientific discipline alone is unlikely to be sufficient to fully characterise a groundwater system. Consequently this study, like many other groundwater investigations, relies on a variety of techniques. In a similar vein, assessment of a groundwater system often involves consideration of large areas of country, so it is rare for any one project to supply all the necessary data to define a groundwater system. Instead, most studies build on previous work that has been compiled from when an aquifer is first drilled and exploited. This

can mean an important part of any groundwater study involves collating, assessing and interpreting relevant pre-existing groundwater data that have been collected by others. This study is no exception; great reliance is placed on previous groundwater drilling and chemical analysis data held in the Northern Territory Government database and, to a lesser extent, some previously published stable isotope data sets. Due to the setting of Palm Valley within a National Park, the remote and rugged terrain, potential procedural difficulties with drilling on Aboriginal owned land, and a lack of a large budget (a constructed water production bore costs over \$10 000 in this region) no new drilling was possible for this project. Thus only existing bores and springs were sampled for this study.

To allay any potential confusion about originality of data presented because of my ready familiarity with using data sourced from the Northern Territory Government database, I have cited the source of all data and interpretations not my own throughout the text. In essence, I collected all the data presented from Palm Valley, apart from the water bore drilling data (bores in place before study began) and some major ion chemical data. I also collected chemical and isotope (including ³⁶Cl) data from a water supply production bore at each of the surrounding Hermannsburg, Areyonga and Injatnarrma Aboriginal communities.

For this study I collated and surveyed bore and water levels to produce the potentiometric surface diagram shown in Figure 4.9.

2.2. NTG database

Groundwater resources are owned and managed by the Crown in Australia. Thus the Northern Territory Government (NTG) like other Australian Governments, acts as custodian of groundwater resource information. This information is gathered at the taxpayers' expense and is therefore a public domain resource. The NTG database comprises vast amounts of digital and hard paper copy information. The files for each NT registered water bore contain paper records of most relevant information. This bore record information is filed under the registered number (RN) of each bore. Other information such as water chemistry and bore hydrograph data are held in various digital database systems and is often duplicated as a paper record on bore record files. Reports of all government groundwater investigations and many other pertinent external groundwater reports pertaining to the central Australian region are also held in the NTG database. These data are held by and are available from the NT Department of Infrastructure, Planning and Environment, Alice Springs, NT.

2.3. Survey data.

Bore elevations and heights (in metres of Australian Height Datum) of some springs and the main stand of palms were surveyed under my supervision by Scotty Balfour and Alan Russ of the NTG for this study in 1994. These data, which helped with construction of the potentiometric surface diagrams, are held in the NTG database in Alice Springs. Traditional dumpy level survey techniques were used for all elevations determined, apart from the elevation of bore RN14616 which was determined by differential GPS techniques. The dumpy level survey results were also crosschecked by the differential GPS method to ensure the level determined for RN 14616 was correct. These survey results are shown in Table 2.1

2.4. Geology

Apart from the cited published geological mapping, various other geological data were examined for this study. Results of various field and aerial photography mapping, conducted near Palm Valley for the petroleum exploration industry are available from company records held by the NT Geological Survey, Department of Minerals and Energy, Alice Springs. Data from petroleum wells drilled along the Palm Valley anticline are also available here. Some NTG water bore record files contain lithological logs compiled by various geologists from examination of drill cuttings and occasionally electric logging data. Nearly all bore records contain a lithological log compiled by the bore driller. Examination of these records and examination of drilling lithological samples held by the NTG were undertaken for this study. Field geology of the Hermannsburg Sandstone was examined at Palm Valley and elsewhere along the Krichauff Range to check the veracity of published and company record mapping.

Table 2.1 Survey results Palm Valley area

NB. GL = ground level; MP = measuring point; WL = groundwater level; AHD = Australian Heihgt Datum

2.5. Pump tests

Bore test pump data and interpretations are held as time versus drawdown charts on NTG bore record files. All tests were conducted by specialist NTG government field crews. Mono (positive displacement) pumps are set for these tests and discharge readings determined by orifice plates. Drawdown readings are taken by calibrated electrical "mega" lines. The tabulated readings taken during the pump tests are used to construct these charts and are held as paper records in the NTG database. Examination of these test pump records and previous groundwater studies conducted in the Hermannsburg Sandstone helped build an understanding of the aquifer characteristics. The supervision of various pump tests of bores sited in the Hermannsburg Sandstone at Areyonga and Palm Valley (Wischusen, 1994b, 1995a) also helped develop my understanding of this aquifer.

2.6. Hydrochemistry

Samples for major anion and cation analyses from bores and springs for this study were taken in new, one litre bottles, filled and rinsed 3 times with groundwater. For bore samples 3 bore casing volumes were purged and field pH and conductivity readings were steady before capping the sample bottle. Spring samples were taken in new bottles filled and rinsed 3 times; bottles were filled by submerging in the deepest part of the pool. In keeping with standard NTG protocols, samples are not fixed by acid or filtered in the field. Other bore chemical data used for this study from the NTG database are assumed to have followed a similar sampling protocol and analyses routine. However given the various vintages of analyses compared in this study it is possible that slight differences in methods used exist. Analyses of major cation and anion chemistry are determined by the water chemistry laboratory of the Department of Infrastructure, Planning and Environment in Darwin which uses the standard methods of the American Public Health Association (APHA, 1998). Iron (Fe²⁺), magnesium (Mg²⁺), potassium (K^+) and sodium (Na^+) ion concentrations were measured by flame atomic adsorption spectrometry (APHA, method 3111B). Calcium (Ca^{2+}) was determined by flame atomic absorption spectrometry (APHA, method 3111D). Chloride (Cl⁻) concentration was determined by argentometric titration (APHA, method 4500-Cl⁻ B), fluoride (F⁻) was measured by ion selective electrode technique (APHA, method 4500-F⁻ B). Total alkalinity (as CaCO₃), bicarbonate (HCO₃⁻), carbonate (CO₃²⁻) and hydroxide (OH⁻) were calculated by titration with acid (APHA method 2320 B). Sulphate (SO₄²⁻) was calculated by the colorimetric method of Goguel (1969). Nitrate (NO⁻₃) was calculated by Ultra Violet spectrophotometric screening technique (APHA method 4500-NO₃⁻, B). Silica concentrations were measured by colorimetry using a Molybdosilicate method (APAH method 4500-Si D). Total dissolved solids (TDS) were measured by evaporation of a sample at 180° C (APHA method 2540C). The pH (laboratory) was determined using instrumentation (APHA 4500-H⁺, B). Electrical conductivity (laboratory) was measured in microsiemens/cm at 25° C (APHA method 2510B). The uncertainty in analytical results for sodium, potassium, calcium and magnesium are ± 1 to ± 2 mg/L, for chloride, sulphate, bicarbonate, carbonate and nitrate ± 2 mg/L and for fluoride and iron ± 0.1 mg/L.

2.7. Chloride bromide ratios

Samples for chloride bromide ratio determination were taken in new 100 mL plastic bottles. The bottles were filled and rinsed three times before fixing pH with a few drops of concentrated nitric acid and capping. The Cl/Br ratios were determined by the University of New South Wales (UNSW) University Analytical Laboratory using standard ion chromatography to determine the chloride level and by Inductively Coupled Plasma Mass Spectroscopy (ICP/MS) technique for the determination of bromide from samples taken in 2002

2.8. Trace elements.

Samples for trace element analyses were taken in 100 mL plastic bottles. Bottles were filled and rinsed three times before fixing pH with a few drops of concentrated nitric acid and capping. Trace element chemistry was determined by Inductively Coupled Plasma Atomic Emission Spectra (ICPAES) at the School of Biological, Earth and Environmental sciences laboratory at the UNSW. Analyses at this laboratory are obtained by a run through a Perkin Elmer machine (Model: Optima 3000 DV).

Standards preparation and quality control follow the United Sates Environmental Protection Agency (USEPA) method 1620: Metals by Inductively Coupled Plasma Atomic Emission Spectroscopy and Atomic Absorption Spectroscopy.

2.9. Stable isotopes

The δ^{18} O over δ^2 H ratios for this study were determined by standard mass spectrometry at the Commonwealth Scientific and Industrial Research Organisation (CSIRO) Isotope Laboratory in Adelaide, South Australia. Samples were collected in new 30 mL glass McCartney bottles after filling and rinsing three times with groundwater. Deuterium for mass spectrometry analyses at this laboratory (Harrington, 1999) is prepared by reducing 25 µL of water to hydrogen gas over depleted Uranium turnings at 800° C in the method developed and described by Dighton et al. (1997). Preparation of an oxygen sample for oxygen-18 analysis entails equilibrating 1 µL of water sample with CO₂ gas for several hours at 30° C using the method described by Socki et al. (1992). Stable isotope measurements are obtained by running the gases produced from the water sample outlined above through special stable isotope mass spectrometer. CSIRO quote an accuracy of ± 1‰ for δ^2 H and ± 0.15‰ for δ^{18} O relative to the Vienna Standard Mean Ocean Water (VSMOW).

2.10. Carbon 13 and 14

Samples were collected in new 20 litre plastic bottles (to ensure enough carbonate was available for analyses) that were filled and rinsed three times from purged bores that recorded stable field conductivity. Samples from springs were collected in new 20L plastic bottles that were filled and rinsed with groundwater once before capping. All preparation of samples for analyses was undertaken by CSIRO in the Laboratory. In the laboratory carbonate is precipitated as $BaCO_3$ solid after the addition of $BaCl_2$ to the sample. At this laboratory the process (Harrington, 1999) involves adding 600 mL of 1.7 M $BaCl_2$ solution per every 20L of sample then raising the pH to between 9 and 10.5 by the addition of sufficient 10M NaOH. Overnight precipitation of the white $BaCO_3$ (and $BaSO_4$) is then facilitated by addition 5 - 10 mL of MAGNAFLOC-156.

The resulting precipitate is then acidified under vacuum using 9M HCl and the CO₂ gas produced is used for mass spectrometry analyses to determine the carbon-13 composition. CSIRO report an accuracy of ± 0.1 ‰ for stable carbon ¹³C results relative to the standard Pee Dee Belemnite Formation of South Carolina (PDB).

Much of the CO₂ evolved from the addition of acid on the BaCO₃ precipitate is also collected for radiocarbon analyses. The CO₂ is frozen using liquid N, water vapour is then removed via a series of ethanol/CO_{2(s)} traps prior to collecting CO₂ in gas flasks. The CO₂ gas is then stored for at least a month to ensure any ²²²Rn gas present in the sample has decayed. The CO₂ gas is then run through a Carbosorb/Permaflour mixture using the direct CO₂ adsorption method developed by Leaney et al., (1994). The resulting CO₂ saturated mixture is then collected in glass scintillation vials and an ultra low background Qantalus liquid scintillation counter used for measuring ¹⁴C activity.

2.11. Chlorine 36

To ensure that no mixing of recent rain and groundwater occurred, particularly at the springs, all the ³⁶Cl samples were collected in October 1994 following several months of no rain at Palm Valley (see Fig. 2.1). Samples were collected in nitric acid washed, thoroughly rinsed with distilled water, one litre bottles that were prepared by the water quality laboratory of the NT Department of Infrastructure, Planning and Environment, in Alice Springs. Bottles were filled and rinsed three times before sealing. Samples were not filtered or pH fixed.

The water laboratory of the Australian Geological Survey Organisation (AGSO) in Canberra then prepared samples for analysis (using silver nitrate to precipitate silver chloride). The AGSO laboratory was geared up for sample preparation, as this organisation was a significant user of ³⁶Cl analyses for groundwater studies at the time. After this preparation, samples were analysed for ³⁶Cl by the Nuclear Physics Department of the Australian National University (ANU) in Canberra.

Determination of ³⁶Cl of a water sample involves several necessary processes to prepare a suitable sample of silver chloride for analyses by an Accelerator Mass Spectrometer.

Particulars of the steps necessary to prepare a sample, as followed by AGSO and also as detailed by Keywood (1995) for ³⁶Cl analyses by the ANU are listed below.

In a laboratory, under low light to prevent photo degradation of AgCl and using spectroscopically pure grade AgNO₃ and BaNO₃ and analytical grade NH₄ and HNO₃, chloride from the water sample is extracted as AgCl and then purified to create a sulphur free sample. This is done by heating 100 to 150 ml of water sample on low heat for 1 to 2 hours, adding 5 ml of AgNO₃ and heating until a precipitate is formed. These samples were then cooled and left to stand overnight covered in parafilm. These samples are then centrifuged at around 3000 rpm for at least ten minutes. After repeating this process on all the supernatant, a bead of AgCl is formed in the bottom of the centrifuge tube.

This bead of AgCl is then dissolved using 1 ml of concentrated NH₄ solution and diluted by addition of 40 ml of Milli-Q[®] water and 5 ml of BaNO₃ added after 5 minutes heating in a water bath. After a further 10 minutes in a water bath these samples are taken out and left to stand overnight. The supernatant of these samples is then drawn off using cleaned disposable pipettes into a clean 50 ml centrifuge tube and the BaSO₄ and possible BaCO₃ precipitate discarded. This supernatant solution is then added to a prepared and heated mixture of 15 ml 2M HNO₃, 10 ml Milli-Q[®] water and 5 ml 10% AgNO₃, a precipitate of AgCl is then formed. After an hour of heating at a low level the samples are cooled, covered in parafilm and left to stand overnight. The precipitate is then transferred to a centrifuge tube and centrifuged at 3000 rpm for around five minutes. The bead of AgCl formed in the bottom of the centrifuge tube is then rinsed and centrifuged two more times. The centrifuge tube and bead of AgCl is dried overnight in an oven. This purified bead of AgCl is then ready for ³⁶Cl analyses in the ANU accelerator mass spectrometer. In each batch of samples prepared, a solution of sodium chloride of known ³⁶Cl concentration (Weeks Island Halite) and Milli-Q[®] is prepared and also run through the preparation process to provide a control for the procedure.

Details of the method of analysis, accuracy of results and a description of the Accelerator Mass Spectrometer (AMS) used at the ANU for ³⁶Cl determinations are outlined by Fifield et al. (1987, 1990, 1994). The process begins at ANU when the

prepared AgCl sample is imbedded into a substrate of high purity (Merck, AR grade) AgBr that coats a copper sample holder used for AMS analysis. This reduces the amount of rouge ³⁶S that can be produced if the caesium ion beam mistakenly strikes the copper sample holders used (Fifield et al., 1990). In simple terms the AMS involves the sample of AgCl being bombarded with a caesium ion beam from a sputter source and the beam of chloride ions produced being accelerated in the ANU 14UD AMS machine. En-route to the heavy ion detector for counting, the chloride ions are stripped to Cl¹⁰⁺ ions and accelerated by a terminal voltage of 14 MV and then accelerated again (tandem acceleration) back to ground voltage. A Faraday cup is used to count stable ³⁵Cl¹⁰⁺ and ³⁷Cl¹⁰⁺ ions, the Faraday cup can be retracted and an in line propane filled ionisation chamber (heavy ion detector) then used to determine ³⁶Cl¹⁰⁺ ion counts. This system, where stable and radioactive isotopes are transported in the same beam, allows the ³⁶Cl/Cl ratio to be determined without reference to a standard (Fifield et al., 1990). The high energies of ions produced also allow clear separation of ions in the beam so that the ³⁶S isobar and stable ³⁷Cl isotope do not complicate the ³⁶Cl count. Each suite of samples prepared by the AGSO laboratory includes one sample prepared from the Weeks Island Halite which has a known 36 Cl/Cl ratio of below 1 x 10 $^{-15}$ (Fifield et al., 1987). An analysis of this sample provides a quality assurance on the sample preparation process. In addition, for each run through the AMS, one of the 12 sample holders contains AgCl prepared from a standard of known ³⁶Cl/Cl ratio (GEC standard with weighted mean value of $436 \pm 2 \times 10^{-15}$ ³⁶Cl/Cl). ANU reported no deviations from the expected ³⁶Cl/Cl ratios for the Weeks Island Halite blank sample or the GEC standard during any of the analyses performed for this study, consequently no analytical errors are expected for the ³⁶Cl results reported here. Fifield et al. (1994) report reproducibility of ³⁶Cl measurements is within $\pm 2\%$ based on regular measurements of the GEC standard.



Figure 2-1 Monthly rainfall at Palm Valley 1993-1994

When sites at Palm Valley were sampled for ³⁶Cl analyses during October 1994 it can be seen that no rain had fallen for months, which reduced the likelihood of contamination of spring water with modern rainfall.

3. Environmental setting

3.1. Introduction

Palm Valley is located in central Australia, 130 km west of Alice Springs. Palm Valley lies within, and is the major tourist attraction, of the Finke Gorge National Park, central Australia (Fig. 1.1). The valley is renowned for a unique arid zone floral assemblage with Red Cabbage Palms (*Livistona mariae*) (see Plate 3.1) being the obvious unique vegetation in the valley (see Plate 3.2). A quarter of the plant species recorded from central Australia occur here. The valley also supports 37 rare, relict and restricted central Australian distribution plant species (CCNT, 1992). The Palm Valley Gas Field (Do Rozario, 1991), which supplies Darwin and Alice Springs with natural gas, lies immediately north of the park. A first management priority of the park is the conservation of the *Livistona mariae* palm tree population.

3.2. Climate

3.2.1. Rainfall

Palm Valley lies within the low rainfall arid area defined by the Australian Bureau of Meteorology (BOM) shown in Figure 3.1. This region is described by the BOM (www.bom.gov.au) as being dry for much of the year, and having an erratic rainfall pattern, with a slight summer maximum. Rainfall can occur in any month but is more likely in summer months. Summers are hot and winters cool in this region. Based on a limited period of available rain records (1981 – 2002) mean annual rain at Palm Valley is 325 mm yr⁻¹. Monthly mean rainfalls for this period are shown in Figure 3.2.

This chart shows summer rainfall to be in excess of winter rainfall, which is typical of the central Australian region (e.g. Hesse, 2004). This rainfall mean may be skewed because the short rainfall record period at Palm Valley which encompasses the extremely wet years of 2000 and 2001, as seen in Figure 3.3 below.

The mean annual rainfall data based on over 100 years of record for the Hermannsburg community, only 15 kms north of Palm Valley, is 234mm. Thus it may be that the

pattern for monthly rainfall data where summer rain is in excess of winter rains, based on over 100 years of record for the Hermannsburg community, shown in Figure 3.4 gives a truer representation of long term rainfall pattern at Palm Valley. Figure 3.3 illustrates the variable nature of rainfall in central Australia where rainfall in 1996 (78 mm) is an order of magnitude less than in 2000 when over 1100 mm of rain was recorded for Palm Valley. Spatial variation of rainfall in this region is also evident as the unusually wet year of 2000 resulted in around 300 mm less annual rainfall at other rain stations within a 125 km radius of Palm Valley: Wallace Rockhole, 50 km east, 780mm; Ormiston Gorge National Park, 65 km north, 820mm; Watarrka National Park, 125 km west, 824mm. This difference could result from Palm Valley catching one or two more heavy thunderstorms during 2000 than at some other rain stations. The wettest month at all these stations was February 2000 when heavy rain was widespread across central Australia. This event was caused by a rain-bearing tropical depression, the former Tropical Cyclone Vincent, moving south into central Australia.



Figure 3-1 The Australian Arid Zone and rainfall seasonality

The arid zone of Australia lies within the solid black line. Palm Valley lies near the middle of the Arid Zone where summer rain is usually 1.3 - 3 times winter rainfall. Adapted from Quaternary International, Vol 118-119, Hesse et al. 2004, "late quaternary climates of the Australian arid zone: a review" p 87-102, ©2004 with permission from Elsevier.



Figure 3-2 Mean monthly rain (mm) from Palm Valley Mean monthly rain (mm) at Palm Valley, Finke Gorge National Park, Northern Territory (BOM, station number DR015663).



Figure 3-3 Yearly rainfall totals at Palm Valley Yearly rainfall totals at Palm Valley, January 1982 to December 2002. (BOM, station DR015663).



Figure 3-4 Mean monthly rainfall totals at Hermannsburg from 100 yrs of record Mean monthly rainfall totals at Hermannsburg from 100 yrs of record (BOM, station DR015531).

The Bureau of Meteorology website gives an extensive general overview of climate and rainfall pattern in central Australia. Much of the discussion below on central Australian rainfall is sourced from the extensive data contained on this website (www.bom.gov.au). The rare large widespread summer rainfall events, such as February 2000 discussed above, are attributed to southerly incursions of the northern monsoon, bringing decaying tropical cyclones or the monsoon trough south into central Australia.

Variability in rainfall is probably also partly linked to variations in the Southern Oscillation Index (SOI), a measure of major air pressure shift between the Asian and east Pacific regions that is calculated from the monthly or seasonal fluctuations in the air pressure difference between Tahiti and Darwin. Sustained positive values of the SOI (La Nina events) are associated with stronger Pacific trade winds and warmer sea temperatures to the north of Australia. Together these give a high probability that eastern and northern Australia will be wetter than normal and an earlier than normal start to the northern monsoon season will occur. The best-known extremes associated with the SOI are the opposite to La Nina and are known as El Niño events which are characterised by sustained negative SOI values. These events impact most strongly on southern and eastern Australia and are associated with drought conditions. (www.bom.gov.au).

Analyses of years with positive SOI from 1910 to 1988 by the Bureau of Meteorology, show that winter- spring rainfall is at the 8th decile on average for these years in central Australia. It is noted, however, that there is not a direct correlation between high SOI years and central Australian rainfall as not all La Nina years have resulted in above average rainfall years.

When the high rainfall years at Palm Valley of 2000 and 2001 (see Fig. 3.3) are compared to records of the SOI, as shown in Fig. 3.5, it can be seen that these extreme rainfall years do correlate with a sustained period of positive SOI. When total monthly rainfall record at Palm Valley is compared to monthly SOI values as shown in Figure 3.6, it can be seen that not all periods of sustained positive SOI have resulted in above average rainfalls. Thus while the La Nina phenomenon may influence central Australia summer rain patterns, there is not a direct correlation with positive SOI (La Nina) events. La Nina years are more likely to result in a longer and more intense northern monsoon, which in turn then increases the opportunity for favourable synoptic conditions to occur that allow a southerly incursion of decaying tropical cyclones and/or the monsoon trough. This fact probably best explains the occasional correlation of extreme summer rain and La Nina events in central Australia. The occasional nature of this relationship is shown on Figure 3.5, where it can be seen that even though the SOI was strongly positive from the spring of 1998 to autumn 1999, no significant summer rain occurred. In contrast, it can be seen that during the following summer, positive SOI values are associated with the heavy rains of February and April 2000.

Thus while the causes of heavy rainfall in central Australia are understood, confident prediction of rainfall is not possible and the Bureau of Meteorology classification of the rainfall pattern in central Australia as being erratic seems the most apt description.



Figure 3-5 Monthly rainfall records and value of the Southern Oscillation Index (SOI) compared for Palm Valley 1998–2001. It can be seen that although a positive SOI existed for three summer periods in a row (1998–1999; 1999–2000; 2000–2001) extreme rainfall occurred only during the 1999–2000 summer. Data from BOM (www.bom.gov.au).



Figure 3-6 Monthly rain records and value of the Southern Oscillation Index (SOI) compared for Palm Valley. The majority, but significantly not all of the wettest months (> 150mm), seem to correlate with long periods of positive SOI. Data from BOM (www.bom.gov.au).

3.2.2. Temperature and Evaporation

Along with this low and variable rainfall, temperature and evaporation rate are high during summer in this region. Figure 3.7 shows the mean maximum and minium temperatures for Alice Springs; large seasonal and diurnal variations are evident. Mean monthly maximum temperatures in January are over 36 °C while the minium mean temperature is around 21°C. Conversely, in July mean maximum temperature is less than 20 °C (19.6) and mean minimum temperature is 4 °C (www.bom.gov.au; weather summary, Alice Springs Airport.).

Evaporation rate is also high in this region. Figure 3.8 shows the mean monthly class A pan evaporation for Alice Springs. When the monthly evaporation is compared to the monthly rainfall data, it can be seen that evaporation can be expected to far exceed rainfall even in winter months. The fact that the 9th decile monthly rain totals shown on Figure 3.8 are also well below evaporation rates indicate it is a rare month indeed for rainfall to exceed evaporation. Mean annual evaporation (3068mm) exceeds mean annual rainfall (285 mm) by an order of magnitude (i.e evaporation / precipitation is 10.8).

3.2.3. Summary

In summary the climate of central Australia is characterised by hot summers and cool winters. Rainfall is low and erratic with rain more likely in summer months. Rare incursions of the northern monsoon can result in widespread rain and thunderstorms in summer months.


Figure 3-7 Mean monthly maximum and minium daily temperatures (°C) at Alice Springs Airport (BOM station 015590).



Figure 3-8 Mean monthly class A pan evaporation with mean and 9th decile monthly rainfall totals at Alice Springs Airport (BOM station 015590).

3.3. Topography

Palm Valley is surrounded by rocky ranges (Plate 3.3). These ranges rise up to 400 metres above the main stand of palms. The topography of the area surrounding Palm Valley is shown in Figure 3.9. Palm Valley is situated in the flatter area surrounded by ranges (known locally as Palm Paddock), as shown by the area of wide spaced topographic contours. The surrounding ranges are steep, as shown by the closeness of the 20m contours. Simpson et al. (1991) note the range country to be steep to moderately undulating with slopes of 15-30° common and some erosion gullies to have sandstone walls with over 200m of relief. The majority of the area, shown in Figure 3.9, drains to the Finke River. This river runs for hundreds of kilometres south eastwards towards Lake Eyre from Palm Valley (Fig. 1.1).

3.4. Land Systems

Land systems are areas or groups of areas throughout which there is a recurring pattern of topography, soils and vegetation (Low et al., 2004). Perry et al. (1962) have mapped and defined the land systems of the Alice Springs region. The Palm Valley region of the Finke Gorge National Park is mapped as being Krichauff land system. This system is noted by Perry et al. (1962) to occur in sandstone range country throughout central Australia. They describe the geomorphology of this system as eroded and dissected sandstone plateaux with superimposed drainage of narrow gorges and joint-controlled tributaries. The dominant vegetation is described as sparse shrubs and low trees over Spinifex grasses. River Red Gums, Ghost Gums and Ironwoods can also occur along river channels that dissect the sandstone plateaux. Dames and Moore (1982) determined the soils of this land system as Lithosol shallow (<10 cm depth) skeletal sands with little pedological development. These soils are noted to be coarse-textured, noncalcareous, lacking in coherence and chemically and physically infertile. In larger drainages such as the Finke River, coarse sands are common. Simpson et al. (1991) describe soils near the apex of the Palm Valley anticline to be comprised of quartz sands and sandstone fragment framework and note these red to reddish yellow soils are thin and rich in secondary iron oxides. The soil matrix is described as kaolinite with traces of illite, montmorillonite and iron oxides.



Figure 3-9 Topography of the area surrounding Palm Valley. The upper map shows the topographic contours at 20m intervals. The close spacing of these contours prevents annotation with contour heights at this scale but an appreciation of the steepness of terrain and relative flatness of the plain, known as Palm Paddock, is discernible. The path of the Finke River as it cuts through the ranges is clearly marked as a line of no contours in the upper right corner of this map. The lower map is a plot of only the 100m (AHD) contour data with heights annotated. The Finke River and Palm Creek drainage lines are seen to cut below the 600m AHD contour. Contour data are from AusLig, via Finke Gorge National Park.

Given the name of this land system is taken from the Krichauff Ranges that border Palm Valley, it is likely the description of this land system well matches the country around Palm Valley. Figure 3.10 shows a schematic diagram of this land system and highlights some of its component land units. Land unit number 6 on this diagram, described by Perry et al. (1962) as narrow valley floors of major through-going drainage, is noted by Low et al. (2004) to contain a broad diversity of plant species and represents the habitat within the Finke Gorge National Park where the majority of rare and relict plant species occur. Figure 3.11 shows a map of the number and distribution of endemic species near Palm Valley. From this figure, it can be seen that the sheltered gullies running south from the Krichauff Ranges hold the greatest number of endemic species. Details of the flora and fauna of the land units of the Krichauff Land system are given by Dames and Moore (1982) and Low et al. (2004).

3.5. Flora

3.5.1. Introduction

Latz (1996) discusses and rates the relict nature of many of the plant species found in Palm Valley. The palms are described in some detail by Latz (1975) and are commonly assumed to represent a relict stand that has survived at Palm Valley since a wetter climate allowed a wider distribution. In recognition of this relictuality *Livistona mariae* is listed by the Australian and New Zealand Ecological Conservation Council as a vulnerable species worthy of conservation (Morton et. al., 1995). The NT government information sheets on threatened species series note this palm is listed by the Federal Environment Protection and Biodiversity Conservation (EPBC) Act (1999) as a threatened species with a vulnerable status.

Figure 3-10 Schematic of Krichauff land system. Numbers mark the various units that make up this land system. Land unit number 6 is the narrow valley floor of through-going drainage and represents the habitat in which most rare and relict plant species occur in the Palm Valley region. (After Perry et al., 1962).



Figure 3-11 Schematic of distribution of endemic plant species around Palm Valley. (Image from Dennis Matthews, Finke Gorge National Park.)

3.5.2. Livistonia mariae

This palm (Family: Arecaceae) was first described, by the naturalist F. von Mueller in the 1870s, from specimens collected from the Finke River by the explorer Ernest Giles. Rodd (1998) reviews several of these descriptions and notes there is some confusion as to which publication actually defines the *mariae* species, but notes that by 1878 this name was associated with the central Australian palm. For a long time (first described in the 1870s by Mueller, Rodd, 1998) perhaps encouraged also by the geographical separation, the Palm Valley Livistona mariae were regarded as a distinct species. Similar palms to the north at (e.g. Mataranka and Lawn Hill, Fig. 1.1) were classified as a separate species, Livistona rigida. In a comprehensive review of the Livistona species in Australia, Rodd (1998) has now reclassified some of these palms. Due to the very similar morphology of the Livistona rigida found at Mataranka, eastern Arnhem Land and far north west Queensland and some remote palms in the Kimberly region of Western Australia (King Leopold and Durack Ranges) with the Palm Valley Livistona mariae, Rodd (1998) regards these three geographical populations as subspecies of Livistona mariae. The Palms at Palm Valley are designated Livistona mariae subsp. mariae, the northern palms (formerly Livistona rigida) as Livistona mariae subsp. rigida and the Kimberly palms as Livistona mariae subsp. occidentalis.

Rodd (1998) acknowledges that the assignment of the central Australian palms as being conspecific with the various northern Australian populations is potentially controversial. However, apart from the slight but consistent differences in leaves and fruits he notes between these populations, he concludes that the striking resemblance of most organs is overwhelming evidence that these populations are the one species. Detailed description and definition of these three subspecies are given by Rodd (1998).

How long the palms have been separated from their nearest relatives 1000 km away in the wet-dry tropics of northern Australia is uncertain. Given the arid climate of central Australia and the fact that Rodd (1998) notes *Livistona mariae sp.* to always occur near permanent water even when found in northern Australia suggest groundwater is crucial to the palms survival at Palm Valley.

3.5.3. Genetic evidence of isolation

3.5.3.1. Livistona species

Preliminary results of a comparative DNA gene sequencing study of the morphologically very similar palms at Palm Valley and Mataranka to the north indicated significant genetic divergence and thereby gave an indication of long time separation between populations (Bowman, pers. comm., 2002). This seemed to support the hypothesis that a refugium operates at Palm Valley. However, subsequent analyses of the chloroplast DNA by Japanese researchers have been unable to detect a genetic difference between these two populations (Isagi, pers. comm., 2003). In fact phylogenetic differences between *Livistona mariae* and other morphologically diverse northern Australian species (*Livistona humilis* and *L. inermis*) could not be recognised from these analyses of chloroplast DNA. (see Fig. 3.12). Thus, given this indication that in some instances speciation may not be registered by changes in chloroplast DNA, further work, perhaps utilising DNA from other types of cells, is needed to determine which, if any, genetic differences exist between the central and northern Australian *Livistona mariae* subspecies.

A comparison of allozymes from an isolated stand of *Livistona australis* in south eastern Australia with the nearest major stand 180km further north by Orscheg and Parsons (1996) had similar results in that no genetic differences were discernible and hence no insight into how long these two populations may have been separated was gained.

Thus, as apparently standard genetic testing methods do not give clear information on genetic differences between populations and sometimes species, more work on the genetic characteristics of the Livistona species may be needed for useful comparisons between separate populations to be made.

Figure 3-12 Preliminary unpublished phylogenetic tree of some Australian Livistona species based on analyses from 6 loci in chloroplast DNA. Reliability of this tree is considered uncertain by its creators, as genetic variations are very low based on these data. Additional analyses of these species genetics is ongoing; however, this diagram illustrates the possibility that the *Livistona mariae* (Palm Valley) and *Livistona rigida* (Mataranka) species are genetically indistinguishable. Data from Isagi, pers comm. 2003.

3.5.3.2. Cycads

The other well known endemic, seemingly wet climate plant species that exists in central Australia and at Palm Valley, the cycad fern (*Macrozamia macdonnellii*) occurs in shaded apparently moist gorge habitats throughout the MacDonnell ranges (Morton et al., 1995). Other Australian cycads occur only in wetter regions along the north, east and south west coasts (see distribution map Fig. 3.13) so this species, like the *Livistona mariae*, has also been considered a relict from when a wetter climate prevailed in central Australia. Sharma et al. (2004) in a study of genetic variation of cycads in northern New South Wales noted that even some morphologically distinct species of Macrozamia had very low genetic differences, which makes it very difficult to genetically characterise these morphospecies as distinct species. Thus even if cycads were adopted as a surrogate for the central Australian palms in an attempt to use genetic differences between central Australian and coastal populations to determine how long seemingly wet climate relict fauna have existed in the region, results may also prove inconclusive. Sharma et al. (2004) note that a range of molecular genetic techniques may be needed to adequately gauge genetic differences between some species of cycad.

3.5.3.3. Discussion

Until rigorous genetic analyses are conducted, other explanations for the existence of the incongruent tropical flora in central Australia are possible. For instance transportation of plant seeds by water birds travelling along chains of natural waterholes less than 200 km apart that link Palm Valley to the south and north of the continent is proposed by Latz (1996) as a plausible explanation for the presence of tropical (non relict) like vegetation at Palm Valley. However, the lack of palms in the intervening country or at other known refugia (Morton et al., 1995) in the central Australian MacDonnell ranges does not lend support to a bird-borne propagation of this palm. Orscheg and Parsons (1996) also note the possibility of bird or even mammal transportation of seed may explain the existence of the isolated stands of *Livistona australis* in western Victoria (southeastern Australia). The cycad fern (*Macrozamia macdonnellii*) has a similar distribution to the palms, in that no occurrences link the coastal and central Australian cycad populations, hence a bird-borne propagation

mechanism, if applicable, has also failed to leave a trail in the intervening country between cycad occurrences. In fact Sharma et al. (2004) note cycads have no mechanism for long distance dispersal. Given these points, the explanation of these "wet climate" plant types being relicts marooned in the arid interior since a previous wetter time seems more likely. Whether this isolation has operated since the onset of aridity in central Australia or since some Quaternary pluvial period when perhaps climate allowed connection to coastal populations along river corridors is yet to be determined.

3.5.3.4. Conclusion

Application of genetic studies has the potential to give an insight on the genetic divergence between the so called relict isolated central Australian "wet climate" flora of palms and cycads with distant analogues that grow in modern day wetter climates. Unfortunately, straightforward techniques such as comparison of allozymes (Orscheg and Parsons, 1996; Sharma et al., 2004) and DNA sequencing (Isagi, pers. comm., 2003) do not necessarily allow ready determination of genetic divergence for these two types of plants. Thus, a study of the relict nature of palms (and cycads) in central Australia from genetic characteristics will seemingly require a specifically orientated comprehensive study using various techniques. The results of any such study would be of interest as apart from period of separation an indication as to how long shallow groundwater has discharged at Palm Valley may also be gained.

3.5.4. Water Requirements

It has long been inferred that groundwater sustains these palms (Chapter 1), actual proof that this is so and what the water requirements of this family of palm are yet to be determined. There is circumstantial evidence that supports a groundwater dependency for the *Livistona mariae*, the palms seemingly survive protracted dry spells here and exist no where else in the arid zone. The populations of relict fish that exist in nearby spring pools (Davis, 1997) support the notion of a permanent groundwater supply being available to the palms at Palm Valley. Protracted dry periods of over 200 days are occasionally recorded in central Australia (e.g. at Tempe Downs, 130 km south west of Palm Valley, Bureau of Meteorology records show that no rain at all was recorded for over 250 consecutive days during both 1894 and 1944).

The fact that date palms thrive at mound springs but no where else in the desert surrounding the Great Artesian Basin fed Dalhousie springs in the driest part of the continent in northern South Australia shows at least one other species of palm that also is inferred need a groundwater source to survive in the Australian arid zone. In central Australia there are some data on Date Palm tree water usages, at the Northern Territory Government Arid zone research institute block at Alice Springs where Date Palms are irrigated and closely monitored for research purposes Nesbitt (pers. comm., 2005) reports that drip irrigated mature Date Palms use an average of 200 litres of water each per day.

The investigation of *Livistona* palm plant water use physiology is apparently yet to be done. Indeed two of the leading experts on *Livistona* Palm species, John Dransfield at the Royal Botanic Gardens, Kew, England and the Australian expert John Dowe at James Cook University in Townsville, Queensland both acknowledge that the link of groundwater to palm growth in desert regions around the world has never been proved to their knowledge, even though the postulated water requirements of these species of plant suggest that this is so. For instance John Dransfield (pers. comm. 2005) notes " ...there are several palms that occur along water courses in areas of very low rainfall, and it has always been assumed that it is availability of near surface groundwater that allows these palms to grow in otherwise very hostile environments." John Dowe (pers. comm. 2005) notes "I dont know of any studies or research projects that have assessed the water requirements of *Livistona mariae*."

There are, however, some studies in tropical Australia where plant water use has been studied e.g. Eamus et al., 2001; Cook et al., 1998; Lamontagne et al., 2005. These studies have not, however, concentrated on any *Livistona* species. Interestingly the recent study by Lamontagne et al. (2005) managed to determine deuterium values for xylem water from several tree species and by comparison to soil, river and groundwater deuterium values in the area estimate which tree species where groundwater dependant during the dry season. Thus it may be that given sufficient data some future adaptation of this technique may yet confirm groundwater dependency of the *Livistona mariae* at Palm Valley.

Given the above it seems that, like the studies of *Livistona sp*. DNA mentioned above, much is yet to be learnt on the physiology of the *Livistona mariae* or indeed any other *Livistona* species.

3.6. Aboriginal connection

The northern section of Finke Gorge National Park in which Palm Valley lies is surrounded by Aboriginal Land. Aboriginal people lived in central Australia before European settlement in the 19th century. It is not known exactly how long people have inhabited the Palm Valley area; however, human occupation has been dated to around 22 ka, 150 km west of Palm Valley in the Cleland Hills by Smith (1987, 1989). Latz (1995) notes that the *Livistona mariae* palm features strongly in Aboriginal mythology throughout the central Australian region.

When the township at Hermannsburg (15 km north of Palm Valley) was settled in 1877 as a Lutheran mission, the local Aranda speaking Aboriginal people slowly began to congregate at this site. Descendants of these "Western Aranda" people (Stead, 1982) have strong traditional links to the country around Hermannsburg where most of them now live. In the late 1990's a small portion of the Finke Gorge National Park, which included the main stand of palms in Palm Creek (Palm Valley), was subject to an Aboriginal Land rights claim under the Federal Government Land Rights (Northern Territory) Act, 1976. In a report of this land claim, Gray (1999) details the strong affiliation many Aboriginal people have with Palm Valley, which is known as *Ntarea* in the Aranda language. In order to determine what group had claim to this land under the Act, Justice Gray took evidence from interested Aboriginal parties on their spiritual connections with sites on the *Ntarea* estate.

Figure 3-13 Distribution of cycads in the Australian region (after Jones, 2002). The *Macrozamia macdonnelii* of central Australia, found at Palm Valley and elsewhere in nearby ranges, is seen, like the *Livistona mariae* palms, to be an isolated mid-continent dwelling species.

The main "dreaming" sites connected to Palm Valley are the *Ratapa*, or Twins Dreaming, the *Lakabara*, or Black Falcon Dreaming and the *Yelka*, or Bush Onion Dreaming. At one of the main dreaming sites *Itaratara*, which lies in the bed of the Palm Creek near the upstream extent of the main stand of palms (see Fig. 3.14), marks in the bedrock represent the twins of the *Ratapa* Dreaming. Custom has it that lying on one of the twins will enable a person to become fat or thin, depending on which twin is selected (Gray, 1999).

One of the outcomes of the Palm Valley land claim was that a group 978 Aboriginal people, the Renkeraka/Ratara group who mainly come from Hermannsburg, were determined to be the traditional Aboriginal owners of Palm Valley. The connection of Aboriginal people to Palm Valley has long been recognised by Finke Gorge National Park management. Consultations on park management issues are routinely made. In fact one of the senior Aboriginal custodians for Palm Valley lives on an outstation 8km south of Palm Valley. This allows for ready communication between park staff and traditional owners.



Figure 3-14 Location and Aranda name of Aboriginal dreaming sites near Palm Valley as listed during a Land Claim hearing. The site marked Itaratara (2.1) is located near the main stand of palms at Palm Valley and represents a significant place in the Ratapa or Twins Dreaming. Image adapted from Gray (1999).



Plate 3-1 *Livistona mariae*, Palm Creek, Palm Valley. Red Cabbage Palms (*Livistona mariae*) growing in Palm Creek at Palm Valley. The Hermannsburg Sandstone forms the gorge wall in the background.



Plate 3-2 Palm Valley; roughly half of the main stand of Palms in Palm Creek looking east towards the Finke River from the top of the northern gorge wall. Vibrant community with all stages from seedlings to mature trees present.



Plate 3-3 Palm Plain as seen looking southwest from Palm Valley 8 gas well site on top of the Krichauff Range from the eastern side of the Finke River (cream coloured sand and river gums along the Finke are seen in the bottom right of the photo.). The contrast between the Palm Plain and surrounding topography is evident. The vast extent of Hermannsburg Sandstone is also illustrated by this photo, as all hills shown are outcrops of this sandstone.

4. Hydrogeology

4.1. Geology

4.1.1. Introduction

The first systematic government geological mapping program to cover Palm Valley was reported by Ranford and Cook (1964) in a set of notes and map covering the Henbury 1:250 000 sheet area. A detailed account of prior geological work is given in these notes; reconnaissance appraisals by various geologists and exploration mapping for minerals and petroleum comprise most of these earlier reports. These notes represent a part of the Australian Government's Bureau of Mineral Resources (BMR) systematic mapping program of the Amadeus Basin, a sediment filled intracratonic depression that covers a large area of central Australia. Mapping and photo-geological studies by various petroleum companies prior to this mapping program had identified various potential petroleum traps based on surface structural expression. One of these features is named the Palm Valley anticline due to its proximity to Palm Valley. In 1965 a successful petroleum (gas) exploration well (Palm Valley No. 1) was drilled on the Palm Valley anticline (a few kilometres north east of Palm Valley on the banks of the Finke River). This well has subsequently led to the development of a commercial gas field along this anticline (Do Razzio, 1991).

4.1.2. Mapping

In 1968 the first geological map of the Henbury 1 : 250,000 sheet which covers the Palm Valley area was published (Cook, 1968). In 1970 detailed results of the BMR mapping program of the whole Amadeus Basin was also published (Wells et al., 1970). Figure 4.1 shows the extent of the Amadeus Basin, distribution of some relevant lithological units, and the location of Palm Valley within the Amadeus Basin.

Figure 4-1 The Amadeus Basin, distribution of relevant lithological units and the location of Palm Valley. (Reproduced from Simpson et al., 1991)

Figure has been removed due to copyright restriction

Figure 4-2 Geology near Palm Valley. On this map the unit Pertnjara group (annotated as Pzp on the diagram) coincides with the distribution of the Hermannsburg Sandstone in the area shown. The Pertnjara Group had not been formally separated into members such as the Hermannsburg Sandstone at the time this geology map was compiled. After BMR, 1970. Figure 4.2 shows the geology near Palm Valley. On this more detailed map a unit called the Pertnjara Group is annotated as Pzp on the diagram. The Pertnjara Group was first proposed by Chewings (1931), with the name pertnjara which means 'many stones' in the local aboriginal Aranda language being an appropriate name for a unit which contains conglomerates. The Hermannsburg Sandstone is a member of this group and the outcrop shown surrounding Palm Valley is in fact solely comprised of Hermannsburg Sandstone (see Figs 4.2 and 4.1). The Pertnjara Group has been differentiated on later larger scale maps. For instance the Henbury 1: 250 000 geological map (Cook, 1968) clearly shows Palm Valley to be surrounded by Hermannsburg Sandstone. It should be noted that as the Palm Valley area lies near the boundary of two 1:250,000 map sheets the older smaller scale mapping image has been presented (Fig 4.2) for clarity. The Pertnjara Group was divided into three members (Parke Silstone, Hermannsburg Sandstone, Brewer Conglomerate) by Wells et al., (1970). The Hermannsburg Sandstone is named after Mount Hermannsburg which lies about 7 km north of Palm Valley in the Krichauff Ranges (see Fig. 1.1).

The geology of the Hermannsburg Sandstone near Palm Valley has been of ongoing interest to petroleum companies, and other mapping covering the Palm Valley area has been presented by Thigpen (1973) and Pancontinental (1984). In all this mapping the Palm Valley area is shown as an anticline with gentle dips. Palm Valley lies on the southern flank of Palm Valley anticline and a syncline axis is mapped just south of the Palm Creek. Dips are very flat at around 5 or less degrees along the southern flank as seen on examination of Figure 4.3. Detailed structural mapping of the Hermannsburg Sandstone, while potentially being of interest as a clue to the structure of deeper gas bearing strata to petroleum companies, has been hampered by a lack of distinct marker horizons (Magellan, 2003).

4.1.3. Stratigraphy

Palm Valley lies on a vast outcrop of the Hermannsburg Sandstone (Figs 4.1, 4.2). The Palm Plain area generally has only superficial Cainozoic cover with the Hermannsburg Sandstone usually within 1 m of the ground surface. Palm Creek, in which the main stand of palms is found, is incised into this sandstone and drains into the Finke River,

Figure 4-3 Map of geological structure near Palm Valley. The Palm Valley anticline axis and shallow dips of the limbs are clearly seen. After Do Rozario et al., 1991.

one of the major rivers of central Australia. The Devonian Hermannsburg Sandstone is part of the Amadeus Basin stratigraphic succession as illustrated in Figure 4.4. The middle to Upper Devonian event, the Pertnjara Movement, initiated the deposition of the Hermannsburg Sandstone in the Amadeus Basin. A subsequent Late Devonian event, the Alice Springs Orogeny, resulted in the folding of the Hermannsburg Sandstone (Wells et al., 1970).

Jones (1991) has further refined the relationship of lithology to tectonics and links variation in distribution, facies and grainsize of the whole Pertnjara Group with several distinct tectonic events or pulses during the Alice Springs Orogeny. Initiation of the Hermannsburg Sandstone deposition is linked to the Henbury Movement. The predominantly medium grained, red-brown quartzose sandstone with very few siltstone lenses suggest deposition on an extensive braid plain. The lack of coarse scale detritus indicates either a low relief or distant source area (Jones, 1991).

The upper Ljiltera Member of the Hermannsburg Sandstone defined by Jones (1972), which is distinguishable from the lower undifferentiated Hermannsburg Sandstone, by the presence of medium and coarse grained pebbly sandstones, was deposited in response to the subsequent Brewer Movement. The presence of larger detritus indicates a more intense uplift along the northern edge of the Amadeus Basin is associated with the Brewer Movement than with the Henbury Movement. The following deposition of 3000+ metres of coarse Brewer Conglomerate above the Ljiltera Member (see Fig. 4.4) indicates the intensity of this movement continued to increase in the Early Famennian. Major folding of the Pertnjara Group that produced numerous intra-basinal anticlines (e.g. the Palm Valley anticline) separated by broad synclines in the Amadeus Basin is inferred by Jones (1991) to have taken place during the Late Carboniferous.

Figure 4-4 Stratigraphy of the north east and south central Amadeus Basin. After Oaks et al., 1991.

4.1.4. Lithology

The red brown Hermannsburg Sandstone near Palm Valley is observed to be the upper Ljiltera Member (Warren and Shaw, 1995). This upper sequence is described by Jones (1972) as orange to reddish brown litharenite and rare sublitharenite, fine to very coarse sand which is commonly trough cross stratified in co-sets varying from 1 to 5 m in thickness. Pebbles and mudclasts up to 7 cm in size are common at the base of these cosets. The cosets generally become finer grained upwards and may merge into smaller scale tabular cross stratified beds or flat beds. The sequence can therefore be divided into numerous partially complete sedimentary cycles.

The Hermannsburg Sandstone is described by Jones (1972) as cross stratified and minor flat bedded medium to very coarse grained sandstone. Fine grained sandstones are rare and sandy siltstone beds are very rare. The main difference of the Ljiltera Member from the rest of the section is that generally pebbles or clasts are rare below the Ljiltera Member.

4.1.5. Structure

The Hermannsburg Sandstone is described by Lindsay and Korsch (1991) as a synorogenic molasse-like succession. This sandstone thins southwards away from the northern Amadeus Basin margin, though Jones (1972) infers the Hermannsburg Sandstone to be over 1000 m thick with the upper Ljilitera Member, characterised by the presence of occasional pebbly facies, being over 200 m thick near Palm Valley.

The Hermannsburg Sandstone is gently folded in the Palm Valley region (Ranford and Mikolajczak, 1963). The Palm Creek area is on the southern flank of the Palm Valley anticline with a syncline axis mapped through Palm Plain. The anticline is mapped as an arcuate (convex to the north) structure at the surface (Hermannsburg Sandstone) and from seismic data with steeper deeps on the northern flank (Do Rozario and Baird, 1987). The anticline axis can be traced for over 40 km at the surface.

In outcrop, this sandstone is jointed near Palm Valley. Thigpen (1973) mapped the joints/lineaments from airphoto interpretation in this area as part of early petroleum exploration by Magellan Petroleum. These lineaments/joints were studied to help understand the nature of the fractured gas reservoir at depth in the Pacoota Sandstone. Figure 4.5 shows a portion of this fracture mapping (with other features) near Palm Valley. Analysis of the fracture patterns identified two major trends, a north westerly and a north easterly. Do Rozario and Baird (1987) interpret these patterns as evidence of one or more conjugate systems of shear fractures formed by compression during folding associated with Alice Springs Orogeny. Other trends, as shown on Figure 4.5, are thought to possibly reflect other episodes of folding and/or younger fracturing due to post folding horizontal rotation or tensional unloading of the anticline.

Mah et al. (1995) using edge enhancement techniques of satellite images and some field mapping along the crest of the Palm Valley anticline have classified some east-west and north-south fracture trends as additional major lineaments. On the basis of these additional trends, some of which seem to have been previously mapped (Fig. 4.5) by Thigpen (1973), Mah et al. (1995) propose a possible two stage deformation model for the formation of the Palm Valley anticline. The first stage, with principle north south stress direction is believed responsible for an east west trending anticline. The second stage with an east to west principal stress direction is then inferred to have transformed the east west Palm Valley anticline into an arcuate anticline. Most other workers (Lindsay and Korsch, 1991; Warren and Shaw, 1995) favour several pulses of north south compression during the Alice Springs Orogeny to have caused the structural features such as the Palm Valley anticline.

While the exact structural history is an important consideration to oil companies interested in the timing of gas entrapment etc, the present day structural and fracture configurations are the most relevant aspects of structural geology to groundwater studies.





4.2. Background - Aquifer characteristics

4.2.1. Introduction

The aquifer characteristic usually calculated from test pumping is aquifer transmissivity. When observation bores in addition to the pumped bore are available aquifer storage parameters can also be calculated. For an unconfined aquifer, the storage parameter sought is known as specific yield, and is defined by Hazel (1975) as "the volume of water which will drain under gravity from a unit volume of aquifer" and thus is expressed as a dimensionless ratio. It is usually noted that this parameter, while obviously associated with the aquifer's porosity, refers only to the drainable volume of stored water; surface tension and lack of interconnected voids will result in some portion of water being retained within the aquifer once the specific yield is released. Transmissivity is a measure of the aquifer's ability to transmit water and is related to the fundamental equation of groundwater flow, known as Darcy's law (Freeze and Cherry, 1979). The derivation of Darcy's mid-nineteenth century empirical law is described in most hydrogeology texts. Krusemann and de Ridder (1970) define Darcy's law thus "the rate of flow through a porous medium is proportional to the head loss, inversely proportional to the length of the flow path, and proportional to a coefficient k, and may be expressed by

$$Q = kiA \tag{4.1}$$

where Q is the flow rate (m^3/day) , k is a constant (m/day), *i* is the hydraulic gradient, i.e. the loss of head h over a distance l, A is the total cross section perpendicular to the flow (m^3) or

$$Q/A = v = ki \tag{4.2}$$

where Q is the flow rate (m^3/day) , k is a constant (m/day), *i* is the hydraulic gradient, i.e. the loss of head h over a distance l, A is the total cross section perpendicular to the flow (m^2) and v is the Darcy velocity (m/day)".

Transmissivity (T) is related to the Darcy flow equation coefficient k by the relation;

$$T = kb \tag{4.3}$$

where b is the aquifer thickness (Hazel, 1975). The Darcy flow equation coefficient k is known as the hydraulic conductivity and is defined by Hazel (1975) "as the rate at which water is transmitted through a unit area under a unit gradient." It should not be confused with velocity, as the dimension of length / time is a shortened version of length³ / length² x time. Transmissivity is then the rate of flow, under a hydraulic gradient equal to one, through a cross-section of unit width over the whole thickness of the aquifer. It is commonly designated by the symbol T and has the dimensions of length³ / (time x length) or length² / time and is usually expressed in the units m²/day (Krusemann and de Ridder, 1970).

4.2.2. Fractured aquifers

The extensive mathematical equations used to determine aquifer characteristics under various hydraulic conditions are mostly based on the Darcy flow equation (Hazel 1975). However, the vast majority of test pumping equations presented in the literature, where the analysis of groundwater flow to bores allows the determination of aquifer characteristics, are based on many assumptions of ideal aquifer performance and usually restricted to groundwater flow through granular porous media only (Freeze and Cherry, 1979). These assumptions include that the aquifer is homogeneous, isotropic and of uniform thickness and that flow to the well is radial, laminar and horizontal (Krusemann and de Ridder, 1970).

The porosity and permeability of the Hermannsburg Sandstone is well documented in studies of this unit over 100 km east of Palm Valley as part of investigations associated with the Alice Springs borefield. Read and Paul (2000) present data from cores within the upper Hermannsburg sandstone where helium porosity of around 20% and permeability to water around 50 milli Darcy i.e. a hydraulic conductivity of 0.04 m/day). Lau, (1989) measured an average permeability and porosity of 80 milli Darcy

(0.07 m/day) and 24 % respectively in the Ljiltera member of the Hermannsburg Sandstone in a cored hole at the Alice Springs town water supply borefield at Roe Creek.

For fractured sandstone aquifers such as the Hermannsburg Sandstone where some of the flow to the well is via fractures of potentially various orientations and dimensions, the usual assumptions made for test pumping analysis and therefore any aquifer characteristics derived, may not be valid for this aquifer type. For example, groundwater flow in any sort of fractured rock aquifer is potentially very complex to analyse as flow may be non-laminar (Mackie, 1982) or non-radial (Milne-Home, 1988) near to pumping bores. This excludes the determination of transmissivity from test pumping methods commonly used to analyse granular porous media. Some alternative testing and interpretation methods for tests conducted in fractured aquifers have been proposed by some workers (e.g. Barker, 1988; Acuna and Yortos, 1995; Novakowski and Bickerton, 1997).

As the Hermannsburg Sandstone can be expected to behave as a double porosity aquifer (fracture and matrix permeability), due to the porous sandstone matrix, still other types of test pumping analyses of data may be more applicable for this type of aquifer. Lods and Gouze (2004) review some of the detailed studies that have previously been undertaken in an effort to combine fractured flow and double porosity concepts (e.g Hamm and Bidaux, 1996) in analysing test pumping results; they present a computer model that may help characterise flow in some double porosity aquifers. However, the use of averaged hydraulic properties is still a requirement of the method presented. Furthermore, Jourde et al. (2002) demonstrate that in some instances where flow is only within two different fracture systems, results may erroneously be interpreted as double porosity flow behaviour (delayed yield like response) and consequently warn that test data from a single well should be used cautiously when assessing flow properties of fractured aquifers.

In a straightforward approach Boehmer (1993) notes that double porosity aquifers should respond in a similar manner to tests conducted in porous media that display a delayed yield response.

Some test data from the Hermannsburg Sandstone (e.g. RN012026, NTG bore file) certainly show a decrease in the rate of drawdown some time (log of time scale) after a constant rate test has begun (i.e. delayed yield). This suggests use of double porosity analyses techniques (e.g. Kruseman and de Ridder, 1991) may provide some additional information (e.g. storage data) on aquifer characteristics for some bores. However, other test pumping results do not show an obvious double porosity delayed yield effect (e.g. at the Hermannsburg community borefield, Childs (1990) notes most tests show a straight line response on drawdown against log of time plots). Thus it seems some bores within the Hermannsburg Sandstone only display a porous medium like response when test pumped. This is further evidence that a dual porosity aquifer system characterises flow in the Hermannsburg Sandstone.

In their classic text book on test pumping analysis and evaluation, Krusemann and de Ridder (1991) warn that "in spite of the intense research that fractured rocks have undergone in the last two decades, the problem is still the subject of much debate." A decade on, the problem of interpreting flow in fractured aquifers was noted to still be at the frontier of research in hydrogeology (Berkowitz, 2002). Indeed Berkowitz (2002) notes that the inherent structural and hydraulic complexity of fractured formations drastically limits the nature and quality of data that can be obtained from field and test pump results, which in turn limits the certainty of any hydrological conclusions drawn from such data.

Most textbooks (e.g. Singhal and Gupta, 1999) and reviews of flow in fractured media (e.g Berkowitz, 2002) note the analysis of flow in fractured rock aquifers can be approached either as a discrete problem where the hydraulics of flow in individual fractures is examined or by a continuum approach where the fractured media is taken as a representative continuum in which spatially defined values of hydraulic conductivity, porosity, storage, etc are assigned, and it is assumed the fractured media then acts in a hydraulically similar fashion to granular porous media. It is worth noting that some detailed experiments that have compared the discrete and continuum approaches to the analyses of flow in fractured rocks have provided evidence of the validity of the continuum approach (e.g., Cacas et al., 1990; Troisi and Vurro, 1988). The work in a deep mine in Switzerland (Cacas et al., 1990) required setting packers to isolate discrete fractures in boreholes and then testing hydraulic properties for each fracture. Such

detailed work to gain hydraulic information on discrete fractures is rarely practical, particularly at remote Northern Territory water bores drilled in the Hermannsburg Sandstone.

This problem of characterizing groundwater flow in fractured media has inspired attempts to simulate flow by developing apt computer models of fractured systems. Many recent studies adopt stochastic approaches as this method can better simulate the expected wide variations (orders of magnitude, Cacas et al, 1990) of hydraulic properties in fractured media compared to the deterministic models commonly applied in granular porous terrains. This strategy of modelling still requires a choice of whether a discrete or continuum approach is adopted. For example Tsang et al. (1996) develop a stochastic continuum model of a fractured medium and account for double porosity affects by assigning a small fraction of the medium higher hydraulic conductivities concentrated along the perceived fracture planes. In contrast, Masciopinto (in press) uses a concept of discrete parallel fractures to develop a stochastic model of flow in carbonate terrain in the Saleno region of Italy. Berkowitz (2002) gives a thorough background on the theory and types of modelling techniques deployed in fractured rock media.

The relation of matrix and fracture flow in fractured rock aquifers has also been examined by finite element modelling techniques. Matthai and Belaynech (2004) by way of field data-based finite element simulations have examined the flow interaction between matrix and fractures in double porosity double permeability systems. They found that stagnation of flow within matrix blocks (a concept used to simulate fractured flow in some earlier studies) would not be expected to occur unless the fracture to matrix hydraulic conductivity ratio exceeded 10^5 and note that this result implies matrix permeability below that expected in hydrocarbon reservoirs. This result would suggest that, as the Hermannsburg Sandstone is porous and permeable, groundwater can be expected to flow both within the matrix and along fractures in this aquifer.

Detailed field studies like some of those mentioned above are seldom possible for groundwater investigations. In the absence of readily useable models to mimic flow through generally unknown fracture patterns, representing the fractured media as an equivalent effective volume of porous media (continuum approach) is often the only available approach to studying some groundwater systems (Freeze and Cherry, 1979). Berkowitz (2002) notes this approach is applicable when the interaction between the fracture network and the porous/permeable host rock allows sufficient interaction to establish a local equilibrium. This then suggests if the scale of study allows such conditions then a continuum approach is a valid concept to apply. Selecting the appropriate scale to match the available data is stressed as being an important consideration when evaluating fractured rock aquifers (Cook, 2003; Berkowitz, 2002). For example Cook (2003) notes that calculating regional groundwater flow rates by multiplying regional head data by aquifer hydraulic conductivity determined from bore test pump test results may overestimate the rate of flow in some fractured terrains. This may arise due to the different scales from which the two data sets are obtained. For instance, if the nature of fracture patterns is such that fractures are not well connected away from the individually tested bores, then the regional flow rates would better be calculated by considering only the matrix hydraulic conductivity rather than the fracture-enhanced characteristics determined from well testing.

4.3. Hermannsburg Sandstone – Regional data

4.3.1. Introduction

Aspects of the hydrogeology of the Hermannsburg Sandstone have previously been described by McQueen and Knott (1982), Lloyd and Jacobson (1987), Jacobson et al. (1989) and Lau and Jacobson (1991). Palm Valley lies in the Amadeus Basin folded rock aquifer province defined by Lloyd and Jacobson (1987). The Hermannsburg Sandstone is a reliable source of potable groundwater (Lau and Jacobson, 1991). Bores are usually deep (>50 m) with modest yields. This aquifer is generally regarded as unconfined, though higher permeability zones such as fractures / joints can result in marked water strikes at depth during drilling. Lau and Jacobson (1991) ascribe the aquifer characteristics of the Hermannsburg Sandstone to both fracture and porous permeability.

4.3.2. Test pumping

When Darcy flow conditions are assumed, tests in the Hermannsburg Sandstone can be analysed as for standard porous media. When analysing unconfined aquifers most workers note the problem of adequately compensating for vertical components of flow to the well, as the aquifer is dewatered during the test. However, later during the test the flow pattern around the well is commonly considered to be a fair approximation of radial flow in a confined aquifer and consequently the standard methods developed for analysing flow in confined aquifers can be used (Freeze and Cherry, 1979; Hazel, 1975; Krusemann and de Ridder, 1991).

One of the commonest methods used to analyse single bore test pump data is the graphical procedure known as Jacob's straight-line approximation (Mackie, 1989) where the method of Cooper and Jacob (1946) is approximated and transmissivity is determined by the equation:

$$T = \frac{2.3\,Q}{4\pi\,\delta\!S}\tag{4.4}$$

where T = transmissivity, Q = discharge per day (m³/day) and $\delta S =$ slope of graph (drawdown per log cycle of time).

Values of transmissivity determined by Northern Territory Government technical officers using this (semi-log) graphical technique are presented in Table 4.1 and cited by other studies (e.g. Lau and Jacobson, 1991; McQueen and Knott, 1982). It should be noted that most of these results are computed from pump tests of eight to 24 hour duration on individual bores; few observation bores are available. Values are determined for the latest times of constant rate tests in an effort to meet the confined aquifer behaviour assumption outlined above. The Jacob drawdown correction for thin unconfined aquifers (Hazel, 1975) is usually not applied.

SampleID	Easting	Northing	Area	Name / type (NT database)	Depth (m)	(m) SWL	T (m ² /day)	k (m/day)	Yield (L/s)	Conduc- tivity (<i>u</i> S/cm)
G0055007	269461	7349064	H'burg area	Kaporilja Spring						740
G0055029	279846	7335932	Palm Valley	Little Palm Spring						444
G0055030	269249	7336285	Palm Valley	Oasis (South) Spring						892
G0055057	264180	7336891	Palm Valley	PV04/200 Shot Hole .	8.0	6.0				811
G00Shotline1	254640	7340422	Palm Valley	PV01/608 Shot Hole	8.0	3.0				
GOROAD	269850	7336853	Palm Valley	Road Spring						774
GOWEST	259550	7337669	Palm Valley	West Spring						802
RN000052	222706	7335104	Areyonga	Production No. 1	55.8	3.70			0.5	670
RN000291	222671	7335209	Areyonga	Production No. 2	137.2	5.50			0.9	930
RN002830	222533	7335586	Areyonga	Production No. 3	108.8	09.0	10	0.092	1.8	824
RN002934	267561	7349576	H'burgtown	Production No. 2	97.5	4.30	3.4	0.036	1.4	885
RN003280	269494	7348731	H'burg area	Station No. 1	54.9				1.3	1140
RN003618	273198	7348641	H'burgtown	Production No. 3	193.0	8.80	1.5	0.008	1.3	1590
RN004508	270745	7336876	Palm Valley	Camp Ground Bore	91.5	1.50	0.14	0.002	0.2	1190
RN005740	282620	7341616	H'burg area	Old Station	36.6	6.40			0.5	1090
RN006949	332731	7346883	Owen Springs	5 Mile	85.3	46.00			1.3	723
RN007175	270160	7349106	H'burgtown	Production No. 4	171.0	1.20	1.5	0.009	3.1	1130
RN007291	271031	7348963	H'burg area	Ironwood	165.3	0.52	З	0.018	3.0	1050
RN007292	267110	7349354	H'burg area	Rainbow	140.7	00.00	14	0.100	8.8	747
RN010209	353853	7340751	Owen Springs	Southern Cross	243.8	121.90			1.0	1018
RN010774	222537	7335609	Areyonga	Production No. 5	182.0	1.49	2.4	0.013	4.5	808
RN011338	282496	7341704	H'burg area	Apmakuaputa	122.0	5.50	0.34	0.003	1.0	1800
RN011341	280243	7344111	H'burg area	Tukarara	129.0	7.00	0.43	0.004	0.7	1160
RN011342	281423	7342791	H'burg area	llpiulalknarrintja	129.0	6.70	0.59	0.005	0.8	1695
RN011343	244171	7345840	H'burg area	Lyalaltum	61.0	11.40	6.5	0.131	2.3	1220
RN011439	241806	7347200	H'burg area	Ilkarilalama	61.0	2.68	29	0.497	2.5	840
RN011440	244171	7345840	H'burg area	Ltltaltoma	83.5	15.36	4.3	0.063	1.0	1050
RN011446	280158	7344050	H'burg area	Coultards	137.0	11.40	0.6	0.005	1.0	1130
RN011447	282069	7342228	H'burg area	Ilpiulalknarrintja	152.5	6.70	0.4	0.003	1.0	1705
RN011448	286478	7337658	H'burg area	Yankee Page 1 of 2	122.0	41.83	0.4	0.005	0.9	1000

Table 4.1 Location and aquifer characteristics of bores drilled in the Hermannsburg Sandstone
Sandstone	
Hermannsburg	
n the	
drilled ir	
pores (
cation and aquiter characteristics of	
able 4.1 Lou	
10	

SampleID	Easting	Northing	Area	Name / type	Depth	SWL	, T	X	Yield	Conduc-
				(NI database)		E)	(m²/day)	(m/day)	(L/S)	u S/cm)
RN011831	377000	7341597	Owen Springs	Railway Bore	432.9	108.40	2.5	0.008	2.0	1450
RN012022	270128	7337171	Palm Valley	Rangers House 1 (1978)	200.0	10.30			0.1	950
RN012023	269128	7337171	Palm Valley	Rangers House 2 (1978)	100.0	1.80			0.1	1120
RN012024	270128	7337171	Palm Valley	Production Bore	67.0	7.70	22	0.371	1.9	971
RN012025	279128	7347171	H'burg area	Arkankaputa	200.0	12.22		0.000	0.1	920
RN012026	278924	7345393	H'burg area	Arkankaputa	168.0	10.00	0.2	0.001	1.1	1190
RN012549	277172	7346094	H'burg area	Tjanukura	150.0	29.70	0.7	0.006	1.0	1360
RN012650	319092	7343058	Owen Springs	Desert	152.0	91.00			2.3	970
RN013652	343489	7318839	Owen Springs	McGinnis	110.0	51.00			1.0	1227
RN013653	363515	7339374	Owen Springs	Rock Hole	330.0	99.70			1.5	500
RN014164	263278	7349604	H'burg area	Tnyimipirta	128.0	14.42	2.3	0.020	2.3	875
RN014165	271018	7348931	H'burgtown	Production 7	166.0	00.0	2.77	0.017	3.6	1075
RN014166	304759	7329716	Wallace	Production	220.0	91.30	3.77	0.029	3.2	1050
RN014616	264224	7334708	Palm Valley	Palm Paddock	121.0	0.80	0.4	0.003	0.8	802
RN014952	288338	7337271	Injarrtnarma	Production	194.6	50.00	4	0.028	2.0	704
RN014956	253180	7348321	Ljiljera	Antjakwera	78.8	5.55	8	0.109	10.0	1150
RN014957	258672	7349160	Ljiljera	Production	112.2	6.50	3.1	0.029	2.0	832
RN015006	271268	7349371	H'burgtown	Monitoring	250.0	5.03			2.0	1280
RN015007	270538	7349000	H'burgtown	Monitoring	220.0	5.98	4	0.019	4.0	1084
RN016359	288338	7337291	Injarrtnarma	Standby	192.7	41.20			1.5	860
	-	MEAN VA	ILUES of WAT	TER BORE DRILLING	148	34	4.6	0.06	1.8	1065

N.B. SWL = standing water level measured below top of casing or ground level. Hydraulic conductivity (k) calculated by dividing tramssmissivity (T) by assumed saturated thickness, i.e. Depth minus SWL.

Page 2 of 2

A shortfall in aquifer testing data is recognised for the aquifer characteristics determined for the Hermannsburg Sandstone from Northern Territory Government testing because:

(1) the aquifer is a fractured sandstone and may not behave as a Darcian continuum.

(2) observation bore data are limited.

(3) delayed yield effects from the double porosity aquifer may have been missed due to brevity of test.

(4) hydraulic boundaries may have affected drawdown graphs.

Nevertheless, the similarity of transmissivities recorded in the Hermannsburg Sandstone (Table 4.1) is an indication of the suitability of late time transmissivity calculations from the single bore tests. This also then implies that taking mean aquifer characteristics for the purpose of regional analyses should be justified. That is, the continuum or equivalent porous media approach is a valid method to assess groundwater flow in the Hermannsburg Sandstone.

It should be noted that such assumptions are commonly made when assessing groundwater resources at a regional scale. For example Nastev et al. (2004) apply a continuum approach when assessing the fractured rock aquifers of south western Quebec, Canada. It should also be noted that the method used by the NT government for determining transmissivity in a fractured rock aquifer is not without precedent. Krusemann and de Ridder (1991) outline the Warren-Root method where transmissivity is also calculated by use of the Jacob straight line method on late time drawdown data from a single well test, although in this case a confined fractured aquifer was tested.

4.3.3. Previous assessments

Lau and Jacobson, (1991) compiled drilling data from many water bores drilled in the Amadeus Basin and tabulated drilling statistics. This work and other regional studies have included an appraisal of aquifer characteristics for the Hermannsburg Sandstone. Lau and Jacobson (1991) note transmissivity (T) in the Hermannsburg Sandstone averages around 100 m²/day (includes an anomalous T value of 1500 m²/day). Analyses of test pumping in the north eastern Amadeus Basin region (McQueen and Knott, 1982) gave a typical transmissivity (T) range of around $0.2 - 20 \text{ m}^2$ /day and hydraulic conductivity (k) range of 0.04 to 1.0 m/day. Nearby to Palm Valley, drilling along the northern flank of the Krichauff Range generally shows less permeable aquifer characteristics for the Hermannsburg Sandstone. Childs (1990) notes that all but one bore drilled for the Hermannsburg community borefield have a T below 4.5 m²/day. These bores are all deep (97 - 220 m) and water levels are at or near the surface, thus hydraulic conductivity must also be low, (i.e. 100 m saturated thickness with T= 4.5 m²/day represents k = 0.045 m/day). Elsewhere along the north side of the Krichauff Range bores drilled into the Hermannsburg Sandstone for outstations show similar aquifer characteristics to the Hermannsburg borefield, that is, the value of k ranges from 0.04 to 1.0 m/day and T ranges from 4 to 20 m²/day (Salas, 1987).

4.3.4. Hermannsburg Sandstone - aquifer characteristics

A compilation of bores drilled into the Hermannsburg Sandstone at a similar latitude to Palm Valley is presented in Table 4.1. These bores from the Northern Territory Government database were determined to tap the Hermannsburg Sandstone by examination of drilling records and published geological mapping. This compilation was undertaken to put the data from Palm Valley in a regional context and represent bores from a similar mid-Amadeus Basin position. The location of the bores presented in Table 4.1 is shown in Figure 4.6.

At Palm Valley bore RN 12024 was tested at $T = 22 \text{ m}^2/\text{ day}$, k = 0.3 m/day in 1994 (Wischusen, 1995a). Other bores near Palm Valley have recorded lower hydraulic characteristics. From Table 4.1 it can be seen that RN 14616 and RN 4508 had T values of 0.4 and 0.14 m²/day and k of 0.003 and 0.002 m/day respectively when tested.

The discrepancy of these aquifer characteristics from the regional compilation of Lau and Jacobson (1991) may reflect variation in facies and/or structural (joints fractures) features occurs elsewhere in the Amadeus Basin. It should be noted, however, that the highest yielding bore in the Hermannsburg Sandstone cited by Lau and Jacobson (1991) RN 11169, is now regarded by Northern Territory Government geologists (Read and



Figure 4-6 Location of bores drilled into the Hermannsburg Sandstone at similar mid Amadeus Basin position to Palm Valley. The hydraulic characteristics of these bores are presented in Table 4.1

Paul, 2000) to be drilled into the Mereenie Sandstone, a much higher yielding aquifer in the Amadeus Basin. Given this reclassification of Bore RN 11169 and the small sample (n = 12) used to compile Hermannsburg Sandstone aquifer characteristics by Lau and Jacobson (1991), it is likely that without this bore, aquifer characteristics would be computed to be closer to that shown on Table 4.1, where n = 29. (The complete list of bores used to compile the Lau and Jacobson (1991) aquifer characteristics data is not known, so mean values cannot be re-calculated excluding RN 11169).

Given these aquifer characteristics are usually derived only from the testing of successful production bores and not all bores drilled, estimates of hydraulic characteristics of the aquifer, as a whole, are likely to be at or below the lower end of the range.

4.3.5. Groundwater flow

Lloyd and Jacobson (1987) estimated head in the Hermannsburg Sandstone and constructed a regional potentiometric surface diagram in the folded rock province of the Amadeus Basin. This diagram with some additional radiocarbon information from Jacobson et al. (1989) is shown in Figure 4.7. Regional groundwater flow is clearly seen to be from west to east in this diagram.

Brown et al. (1990) used a horizontal and vertical k of 0.03 and 0.003 m/day respectively when modelling flow through this aquifer. If such aquifer characteristics are representative, then slow flow through the Hermannsburg Sandstone can be expected. If aquifer hydraulic conductivity and porosity are known, Darcy's Law can be used to estimate the travel time of water over a specified distance under a given gradient by the formula:

$$T = \frac{x\theta}{ki} \tag{4.5}$$

Where T = travel time (in days) over distance x, x = distance in metres, θ = porosity, i = gradient and k = hydraulic conductivity in m/day.



REGIONAL POTENTIOMETRIC SURFACE -HERMANNSBURG SANDSTONE

Figure 4-7 Regional Potentiometric Surface – Hermannsburg Sandstone.

This Figure shows the potentiometric surface over a large area of central Australia (After Lau and Jacobson, 1991). A west to east gradient is clearly seen in the Palm Valley area. This diagram was constructed without the benefit of water level data from Palm Valley but highlights how the local data shown in Fig. 4.9 fits with the regional pattern. The locations of communities where bores that tap the Hermannsburg Sandstone have been sampled for this study are also shown. If a regional gradient of a 100m drop every 50km (Fig. 4.7, $i = 2 \ge 10^{-3}$) and a porosity of 6% (Magellan Petroleum, 1966)) along with a horizontal hydraulic conductivity of 0.03 (Brown et al. 1990) are used to determine travel time assuming Darcy flow conditions *T* is calculated (Eq. 4.5) as $T = 9.85 \ge 10^{-4}$ m/day. The average linear velocity *V* can be calculated by use of the formula:

$$V = 365 \frac{ki}{\theta} \tag{4.6}$$

Where V= average linear velocity, k = hydraulic conductivity in m/day, i = gradient and θ = porosity (i.e. effectively 365 x T.)

From the regional data discussed above average linear velocity is then calculated to be 0.36 m year⁻¹. When the *k* values near those estimated for bores RN 4508 and RN 14616 (Table 4.1) near Palm Valley are used (i.e. k = 0.0025 m/day), calculated average linear velocities are around 0.03 m yr⁻¹. Both these average linear velocity estimates and the modelling of Brown et al. (1990), with an average linear velocity of 0.06 m year⁻¹ through the Hermannsburg Sandstone, highlight the potential for very slow groundwater movement through this aquifer.

4.4. Local Hydrogeology

4.4.1. **Previous work**

The first bore at Palm Valley was drilled near the Campground to service a chalet that then existed in the area. This bore (numbered out of sequence in the Northern Territory data base as RN011490) was drilled in 1961 to depth of 51 m in the Hermannsburg Sandstone. Records are scarce for this bore but it apparently flowed at 0.2 L/s after initially being thought a dry hole. In 1964, the current campground bore RN 4508 (Fig. 4.8) was drilled, a yield of 0.2 L/s and water level of 1.5 m below ground was recorded. In 1978, following the drilling of several exploratory bores, the bore RN 12024 was drilled and constructed as a production bore. In 1986 this bore was deepened in an attempt to increase yield, and a bore for an outstation (Conrads bore for Intyiamangama outstation) in Palm Paddock RN 14616 was drilled (Fig. 4.8).

An investigation was then undertaken in 1988 (Warne and Prowse, 1988) to look for additional water supplies for planned expansion of the campground at Palm Valley. Eleven shallow (40-60m) holes were drilled within 2 km to the west of RN 4508 (Fig. 4.8) but no supply greater than 0.1 L/s was encountered, and all these holes were backfilled (bores RN 14897-14907, Table 4.2).

Based on the results of this and nearby drilling, Warne and Prowse (1988) considered joint/fracture development was limited. They went on to develop a concept that only a limited shallow band of fractured Hermannsburg Sandstone, locally recharged from creek flow, stored groundwater in this region. The potential for intergranular storage was discounted. This concept raised concern about the possibility of competition between the unique vegetation of this park and tourists for the then perceived locally restricted groundwater supplies.

Given these concerns the existing bore RN 12024 was examined to see if its use presented any danger to the palms groundwater supply (Wischusen, 1995a). This report concluded that interference to the palms' water supply from park domestic extractions from bore RN 12024 would be most unlikely. The reasons were considered to be the large distance (7 km), potentiometric surface drop between the bore and main stand of palms of more than 30 m, and limited propagation of drawdown effects in low permeability aquifers (Theis, 1940) indicated interference to be unlikely. Furthermore, even discounting additions from aquifer throughflow and induced drainage, something in the order of 30 m of 6% porosity aquifer over a radial area of 7 km² would need to be dewatered (i.e. around 270 ML) before affecting the palms. This interpretation readily allayed fears that the modest groundwater extraction for park use would adversely affect palm tree health. From analysis of test pump data, it was concluded a yield of 30 m³/day is sustainable from RN 12024 (Wischusen 1995a). This yield, along with the occasional use of the bore at the campground (RN 4508), easily meets the requirements of park visitors and rangers.



with the Palm Creek stand of palms. The creeks drain to the Finke River, a major river of the Lake Eyre Basin that runs southwards for (Livistona mariae) growth and creeks in the north-western part of the Finke Gorge National Park. The name Palm Valley is associated Figure 4-8 Location of drill holes in the vicinity of Palm Creek (data from these holes are shown in Table 4.2) and area of Palm many hundreds of kilometres (Fig. 1.1).

4.4.2. Groundwater features

4.4.2.1. Introduction

Figure 4.9 shows local groundwater features and bores drilled in the Palm Valley region. A summary of drilling data from these bores is shown in Table 4.2. The regional west to east gradient shown through the Hermannsburg Sandstone in Figure 4.7 is corroborated by the potentiometric data available in the Palm Valley region shown in Figure 4.9.

Figure 4.5 shows the distribution of joints and fractures in the Hermannsburg Sandstone mapped from air photos near Palm Valley by Thigpen (1973). The position of production bores and springs are seen to have some correlation with these mapped fractures. Most of the dry or low yield backfilled bores shown plot away from mapped fractures. The lack of supply noted for Bores RN 14897 – 14907, drilled in the vicinity of Road Spring and Bore RN4508, illustrate how the low k of the Hermannsburg Sandstone can limit the observed yield of shallow rotary air blast drilled holes when no fractures are encountered.

4.4.2.2. Discharge sites

Groundwater seepage sites are commonly observed at many locations along the creek beds in the vicinity of Palm Valley. Low yielding seepages, some of which dry up in the heat of the day, appear to be bedding related. That is, groundwater apparently oozes (Plate 4.1) from small lenses of slightly coarser, cleaner sandstone (i.e. at the base of some crossbed cosets which Jones (1972) notes can be coarser grained). Groundwater is also seen to seep from joints (Plate 4.2). The line of calcrete deposits associated with the discharge mark the joint plane.



Figure 4-9 Potentiometric surface – Hermannsburg Sandstone in Palm Valley and Hermannsburg community region. Bore and springs are shown with radiocarbon and standing water level (SWL; reduced to Australian Height Datum (AHD)) data, where available. Potentiometric surface is a composite of readings derived from the different most recent dates available in the NTG database.

BORE RN	LOCATION	Depth (m)	SWL (m)	Supply (L/s)	Status
4508	Palm Valley	92	1.5	0.2	Production
10774	Areyonga	180	3	8	Production
11940	Palm Valley	55	surface	0.2	Backfilled
12022	Palm Valley	200	10.3	0.13	Open hole
12023	Palm Valley	100	1.8	0.13	Backfilled
12024	Palm Valley	67	8.4	1.9	Production
14616	Palm Valley	121	0.8	0.8	Production
14897	Palm Valley	60		0.1	Backfilled
14898	Palm Valley	40			Backfilled
14899	Palm Valley	40			Backfilled
14900	Palm Valley	30			Backfilled
14901	Palm Valley	40		0.01	Backfilled
14902	Palm Valley	30			Backfilled
14903	Palm Valley	30			Backfilled
14904	Palm Valley	30			Backfilled
14905	Palm Valley	30			Backfilled
14906	Palm Valley	30			Backfilled
14907	Palm Valley	30			Backfilled
14952	Injarrtnama	195	41	2	Production
14165	Hermannsburg	166	surface	3	Production
1/608	Palm Valley	8	3		Shot hole
4/200	Palm Valley	8	6		Shot hole

The groundwater appears saturated with calcite, as most seep sites are associated with calcrete deposits. These deposits may be precipitated when the solubility of the calcite decreases as water temperature rises during the heat of the day. In many instances active seepages are surrounded by dry calcrete deposits, (Plate 4.3). These dry deposits appear to be blocked, former seep sites. Calcrete deposits are also common elsewhere out of the creeks on the surface of the Hermannsburg Sandstone at Palm Valley. Some of these deposits may represent where seeps from an ancient, higher, potentiometric surface have occurred (Plate 4.4). Apart from these observable discharge features, it is also possible that unseen diffuse discharge through the less permeable sandstone matrix occurs and/or resultant mineral precipitation from evaporation clogs near-surface fine matrix pores.

The springs shown in Figure 4.9 are evident as $20 - 100 \text{ m}^2$ pools of water around 1m deep. Thick vegetation grows near these springs. Only West Spring has no *Livistona mariae* palms nearby. West Spring (Plate 4.5) and Road Spring (Plate 4.6) are known as permanent springs (Finke Gorge Park rangers, pers. comm. 1992). These two springs are evident as several square metres of pooled water over 1 m deep within bare sandstone outcrop along creek lines. Small active seepage sites surround each of these springs and the main pools of water are intersected by vertical joint planes in the sandstone. Oasis (bottom water) spring (Plate 4.7) is also positioned near a vertical joint plane. This spring, which has a population of fish, may also be permanent, though the heavily vegetated upstream main spring area (Oasis Spring, Plate 4.8) could provide a permanent refuge for fish. It should be noted that this main spring area is so heavily vegetated that no significant pools of water to obtain samples from were found here.

Another reputed permanent spring is in the Little Palm Creek. This spring is observed to vary markedly in size and, when sampled, had water of much lower conductivity than at the other springs and bores in this region (Table 5.1). In contrast to other springs it was sampled after summer rains in March 1993. The lower conductivity was taken to indicate that mixing with recent rain had occurred. For this reason and also due to budget constraints, a full suite of isotope analyses was not run for samples from this site. Davis (1995, 1997) regards all three of these springs as permanent water bodies on the basis of limnological studies that describe relict fauna at each site.

All of these springs are taken to represent groundwater discharge from enhanced permeability conduits, such as joint planes or cleaner sand lenses within the low permeability Hermannsburg Sandstone

4.4.3. Aquifer type

If the Hermannsburg Sandstone is taken to be a low permeability saturated sandstone with intermittent joints/fractures providing conduits of enhanced permeability, as evidenced by occasional springs and higher yielding bores, then the drilling results of Warne and Prowse (1988) at Palm Valley can be interpreted differently. That is, rather than showing no groundwater in the Hermannsburg Sandstone, it may be that the shallow drilling encountered no significant joints/fractures and that insufficient aquifer area was exposed for groundwater seeping to the hole be noticed during rotary air blast drilling. All these holes were backfilled immediately after drilling so no recovery of water level data could be observed. Bore RN 6503, 3 km south of Hermannsburg and RN 11490 at the campground (Fig. 4.8) when drilled, were thought to be dry but subsequently flowed. Two bores, RN 7292 and RN 14165 at the Hermannsburg borefield also flow.

The drilling data either side of the Krichauff Range at Hermannsburg and Palm Valley can be interpreted as evidence that the Hermannsburg Sandstone - while having a low hydraulic conductivity unless enhanced by joints/fractures - is a saturated intergranular aquifer, often with potentiometric levels at or near the ground surface. The porosity, areal extent and thickness (up to 1300 m; Wells et al., 1970) of the Hermannsburg Sandstone indicate that large volumes of groundwater may be stored within this aquifer. Such an interpretation is in keeping with regional hydrogeological studies of the Amadeus Basin (Fig. 4.7).

4.5. Summary

The Hermannsburg Sandstone aquifer is a double porosity aquifer with low matrix k (Table 4.1). Permeability is occasionally enhanced by fractures. In unfractured regions deeper bores are often required for viable yield. At Palm Valley occasional permanent pools of water (springs) are found on Hermannsburg Sandstone outcrops. These springs result from increased discharge associated with localised permeability enhancement.

On face value (consistent geology and potentiometric trend), it appears this shallow groundwater originates from a vast resource of water moving slowly through the regionally extensive Hermannsburg Sandstone. As groundwater is likely to be crucial to the ongoing survival of the Palm Valley ecosystem, examination of other aspects of the hydrogeology is warranted.



Plate 4-1 Bedding related seeps, Hermannsburg Sandstone.

Water seeping from the Hermannsburg Sandstone (dark areas within the paler calcrete regions, ruler in photo centre is 30 cm long). These weeps seem bedding related with associated calcrete deposits showing both active and former discharge sites. Features such as this show that the watertable is at, or above, the surface in areas of Palm Valley.



Plate 4-2 Joint related seeps, Hermannsburg Sandstone Water seep site from a joint in the Hermannsburg Sandstone within Palm Creek. The pale calcrete deposits associated with discharge clearly mark the joint plane.



Plate 4-3 Calcrete deposit associated with a seep site. Seep site in Hermannsburg Sandstone near West Spring where a plant has taken root in an associated calcrete deposit.



Plate 4-4 Calcrete deposit above current potentiometric surface.

Thick calcrete layer on the Hermannsburg Sandstone along the edge of Palm Creek near West Spring. This calcrete (seen as a pale bed on top of the darker Hermannsburg Sandstone) which is now above the potentiometric surface, may have formed by groundwater discharge when the potentiometric surface was higher some time in the past.



Plate 4-5 West Spring.

Some of the ponds of water at West Spring, Palm Valley. No palms grow near to this Spring.



Plate 4-6 Road Spring.

viewed looking westwards from high on a hill. The lone *Livistona* palm marks the location of the permanent spring. At the time of photograph (August 2002) some flow from recent rain can be seen along creek bed in distance. All land surface seen in photo is outcrop of Hermannsburg Sandstone.



Plate 4-7 Oasis Spring (bottom water), Palm Valley.

A lone Palm (*Livistona mariae*) grows at this part of the spring system. Fish are recorded here. A distinct joint plane forms one boundary of the water pool. This accessible pool of water was sampled for chemistry and stable isotope analyses.



Plate 4-8 Oasis Spring, main area.

This heavily vegetated area is the main Oasis Spring area. No readily accessible pools of water were found for groundwater sampling during visits to the area. The vibrant vegetation, however, suggests a significant resource of groundwater.

5. Hydrochemistry

5.1. Introduction

Many factors influence the make up of groundwater chemistry. At the ground surface ionic concentration can be increased by evaporation of rain or stream water and/or by dissolving surface evaporites. From the time rain or stream water enters the soil, the chemical reactivity of water ensures a rapid change in the original water composition. The chemistry of a host rock aquifer then influences subsequent developments of water chemistry, with additional variables of rock surface area, water residence times, solute transport rate away from reaction sites and temperature all further influencing water chemistry (Freeze and Cherry, 1979).

Evaporation of shallow groundwater can influence groundwater chemistry by increasing ionic concentration. However, direct evaporation of groundwater is restricted to a theoretical maximum depth below ground surface of 4 m (De Weist, 1965), and is usually only a factor when groundwater is less than 2 m below ground surface (Schoeller, 1959).

Transpiration by plant root uptake of water can also be expected to concentrate soil moisture at the root zone (Allison, 1988), which can then be flushed or dissolved into groundwater, thereby increasing ionic concentration. Transpiration by direct water uptake from the saturated zone by tree roots, as postulated by Pidsley et al. (1994) may also concentrate groundwater.

The chemical equilibrium reactions possible with dissolved soil gases and various host aquifer compositions and the change of individual constituent solubilities as a result of various combinations of these chemical factors are extensively examined in texts such as Hem (1985), Lloyd and Heathcote (1985) Freeze and Cherry (1979), Stumm and Morgan (1996). Schoeller (1959) gives a comprehensive overview of groundwater chemistry, especially in relation to arid zone conditions where respiration by plants is noted to produce significant soil CO_2 gas which plays an important part in arid zone groundwater chemistry.

The most obvious result of these chemical reactions between water and the subsurface is the amount of dissolved solids within the water, which controls salinity and therefore the quality of groundwater. Apart from the health considerations of individual ions that may be present in groundwater, hydrochemical evaluation of groundwater is usually based on six major inorganic ions: sodium (Na⁺) magnesium (Mg²⁺), calcium (Ca²⁺), chloride (Cl⁻), bicarbonate (HCO₃⁻) and sulphate (SO₄²⁻). The total concentration of these six major ions normally comprises more than 90% of the total dissolved solids in water regardless of whether the water is dilute or has salinity greater than seawater (Freeze and Cherry, 1979). Standard chemical analyses conducted by the Northern Territory Government often also include potassium (K⁺), fluoride (F⁻), nitrate (NO⁻₃), iron (Fe²⁺) and silica (SiO₂) determinations in addition to the six major inorganic ions.

5.2. Previous work

The hydrochemistry of the Hermannsburg Sandstone in the Missionary Plain area near Hermannsburg community (12 bores) was briefly examined by McQueen and Knott, (1982). Groundwater was noted, based on classification of water types according to the domain in which they plotted on the major anion and cation triangles of a Piper Diagram (see Fig. 5.1), to range between a magnesium bicarbonate chemical type and to a lesser extent calcium bicarbonate type. On a Piper trilinear diagram the Hermannsburg Sandstone analyses generally plotted near the middle of the cation triangle with some plots showing Mg^{2+} as the dominant cation. Anions plotted as bicarbonate water with 25 - 50 % chloride. Sulphate concentration showed only a slight increase with chloride content. The Hermannsburg Sandstone data were noted to be a distinct group compared to the Mereenie Sandstone (Deep Well area) where sodium and chloride were more dominant. Jacobson et al. (1989) also present chemical data from 11 bores thought to tap the Hermannsburg Sandstone in the folded rock belt of the Amadeus Basin. No chemical distinction was drawn between these bores and the four other Mereenie Sandstone bores and one Bitter Springs Formation bore that made up their folded rock belt data group. Cation chemistry showed variation in that calcium, sodium or magnesium could be the dominant cation in this sample set. Anion data showed low relative sulphate and either bicarbonate or chloride to be dominant.

5.3. Palm Valley area

5.3.1. Major ion chemistry

The major ion chemistry of sites sampled for this study is shown in Table 5.1. The Northern Territory Government Water Laboratory in Darwin conducted all these analyses. The Standard methods of the American Public Health Association are used at this laboratory (APHA, 1998). All groundwater is, by central Australian standards, low salinity with TDS less than 800 mg/L.

Figure 5.1 shows a piper plot of data in Table 5.1. From this figure, it can be seen that apart from Injarrtnama (RN 14952) these waters have no dominant cation discernable when using this graphical technique. There is, however, a trend from bicarbonate dominant to no dominant anion type water. This limited range of graphically classified water type can then be taken to indicate only slight variation in evaporation concentration and/or water – rock reaction histories has occurred. For example, Oasis Spring plots nearest to chloride type water ($Cl^- > HCO_3^-$ Table 5.1), probably a result of evaporation from this spring pool (see Chapter 6).

The fact that Areyonga, Injarrtnama and Hermannsburg groundwater (RNs 10774, 14952, 14616 respectively) all plot in a similar position to the Palm Valley samples indicates that no distinct spatial trend in Hermannsburg Sandstone groundwater chemistry is seen along the regional outcrop (Fig. 4.7). By contrast many previous studies chart the evolution of anion chemistry from HCO_3^- dominant to either Cl⁻ or SO_4^{2-} dominant along flow path, particularly in large sedimentary basins (Freeze and Cherry, 1979; Schoeller, 1959; Jankowski and Jacobson, 1989). These results suggest comparisons based on anion chemistry may not be helpful here.

Table 5.1 Groundwater chemical data, Palm Valley study

0.2 13
-
0.4 14
0.4 14
0.6 33 147
0.5 22
0.4 18 160
0.3 15
0.4 19
0.5 18

equivalents exceeds 10% of the molar concentration of the solution.^b Cl/Br ratio determined by UNSW University Analytical Lab using standard ion chromatography to determine the chloride level and by ICP/mass spectrometry technique for the determination of bromide from samples taken in 2002. Depth of boreholes sampled shown in Table 4.2. All samples, apart from RN004508, collected by John Wischusen.



Figure 5-1 Piper plot of major cation chemistry. Samples taken from the Hermannsburg Sandstone at Palm Valley and selected communities (See Table 5.1).

Given the similarity in relative major cation concentrations seen in Figure 5.1, it is likely that some commonly used water type classification schemes, such as the Szczukariew and Priklonski scheme (Alekin, 1970) where water type classification is based on listing only the ions that occur in concentrations above 20 % of the molar concentration of the solution, are not suitable for examining the subtle differences in cation chemistry seen for this aquifer. Consequently, the more sensitive water type classification scheme derived from the computer program Aquachem (1999), where major anions and cations are listed only when concentration in milli equivalents exceeds 10% of the molar concentration of solution is used throughout this study.

Examination of the water types classified in Table 5.1 shows all sites analysed for this study (apart from Oasis Spring discussed above) are bicarbonate dominant type waters. This is typical of low salinity arid zone groundwaters where recharge through vadose zones enriched in CO_2 gas from heavy plant respiration, is expected (Mazor, 1997). Water type classification based on cation chemistry, however, is seen to vary for these samples. Variation in major cation chemistry is common in groundwater flow systems; however, detailed consideration of specific hydrochemical processes is often needed for observed concentrations to be accounted for. A simple examination of cation chemistry below outlines some of the hydrochemical processes that may occur within the Hermannsburg Sandstone.

5.3.2. Hydrochemical processes

5.3.2.1. Saturation Indices

Figure 5.2a shows the saturation indices (SI) with respect to calcite and dolomite for each sample. Only Little Palm Spring water plots as undersaturated, though it is still close to the range where equilibrium to one or both of these phases might reasonably be expected (e.g. SI \pm 0.3). Saturation indices were calculated using the PHREEQC model (Parkhurst and Appelo, 1999). As no field HCO₃⁻ determinations were made and field pH measurements were not recorded at all sites, laboratory results were used to calculate the SI data shown in Figure 5.2a. While field pH and HCO₃⁻ measurements are desirable for accurate SI calculation (Hem 1985) it is believed that these data still give a reasonable indication of SI. This is because field pH, when recorded at 8 sites, is never



Figure 5-2 (a-d), aspects of groundwater chemistry sampled for this study in the Palm Valley area. Plot of (A); saturation indices of calcite and dolomite (B); magnesium against bicarbonate (C); calcium against bicarbonate (D); sodium against chloride.

less than 90% (usually >95%) of the laboratory determined pH. Given the similarity of field and laboratory pH and the fact that field HCO3- is often higher than laboratory HCO3- (Hem, 1985), using laboratory parameters to determine SI most likely gives a fair approximation here.

Given this saturation with respect to carbonates, variations of Ca^{2+} , Mg^{2+} and HCO_3^- concentration in groundwater, should not be further influenced by aquifer carbonate dissolution as equilibrium conditions are indicated. Freeze and Cherry (1979) note that Ca^{2+}/Mg^{2+} ratio can vary widely due to a number of different processes in calcite and dolomite terrains, even without cation exchange reactions. It should be noted that a positive SI does not necessarily mean that phase exists within the system.

5.3.2.2. Dissolution / Precipitation reactions

The majority of samples show magnesium to be the dominant cation (Table 5.1). When magnesium is plotted against bicarbonate (Fig.5.2b) and compared to a 1:2 ratio line (representing dissolution of pure dolomite), it can be seen that a simple dissolution of dolomite;

$$CaMg(CO_3)_2 + 2H_2O + 2CO_2 \leftrightarrow Ca^{2+} + Mg^{2+} + 4HCO_3$$
(5.1)

cannot account for all the magnesium concentration in this aquifer. While many different processes may produce magnesium in excess of that expected from dolomite dissolution, the make up of the Hermannsburg Sandstone gives a clue to possible processes. Chlorite and micas are noted secondary minerals in the Hermannsburg Sandstone (Lau and Jacobson, 1991), thus the high Mg²⁺ content of groundwater could be a result of the weathering of detrital magnesium rich chlorites. For example chlorite in its most simple form (Drever, 1988) may react as:

$$Mg_{6}Si_{4}O_{10}(OH)_{8} + 12H^{+} \leftrightarrow 6Mg^{2+} + 4H_{4}SiO_{4} + 2H_{2}O$$

$$(5.2)$$

Groundwater calcium is commonly derived from dissolution of calcium rich carbonates (e.g. calcite):

$$CaCO_3 + H_2O + CO_2 \leftrightarrow Ca^{2+} + 2HCO_3^{-}$$
(5.3)

Figure 5.2c shows the expected 1:2 calcite and 1:4 dolomite (Eqs. 5.1& 5.3) ratio lines, based on the stoichiometry if calcium were derived from dissolution of calcite or dolomite respectively. This plot shows dissolution of calcium carbonates could account for most of the calcium in solution.

Figure 5.2d shows a sodium against chloride plot. While a positive correlation consistent with evapo-concentration of recharge water is observed, two points plot off the linear trend. This may indicate some other reactions can interfere with the Na/Cl ratio of recharge water once it enters the aquifer. As chloride is typically conservative, reactions with sodium rich feldspars within the feldspathic Hermannsburg Sandstone may well account for slight anomalies in Na/Cl ratio. For example dissolution of sodium feldspars would increase Na⁺ relative to Cl⁻ by way of the reaction:

$$2NaAlSi_{3}O_{8} + 11H_{2}O + 2CO_{2} \leftrightarrow Al_{2}Si_{2}O_{5}(OH)_{4} + 2Na^{+} + 4H_{4}SiO_{4} + 2HCO_{3}^{-} (5.4)$$

Similar reactions of calcium rich plagioclase feldspars and/or back reactions with weathering products may account for the slight variation of calcium concentration from the calcite dissolution line (Fig. 5.2c). For example;

$$CaAl_2Si_2O_8 + 2CO_2 + 3H_2O \leftrightarrow Al_2Si_2O_5(OH)_4 + Ca^{2+} + 2HCO_3^{-}$$
(5.5)

Cation chemistry may also be influenced by the change of conditions experienced once groundwater discharges at a spring or seep. Degassing of dissolved CO_2 and or temperature rise can be expected to lower the solubility of carbonates, which may then precipitate. As calcite precipitates more readily than dolomite (Hem, 1985), a decrease in the Ca^{2+}/Mg^{2+} ratio in spring water when compared to nearby groundwater might be expected. No such pattern, however, is discernible when spring and bore chemistry is

compared (Table 5.1). The fact that all sites are saturated with respect to carbonates (Fig. 5.2a) also means observed variation in Ca^{2+}/Mg^{2+} ratios cannot be readily attributed to preferential precipitation of calcite. The one site not supersaturated with respect to calcite and dolomite, Little Palm Spring, has Ca^{2+} in excess of Mg^{2+} (Table 5.1). Despite these data, the readily observed calcrete stains and deposits throughout the area indicate carbonates are precipitated at Palm Valley.

5.4. Regional data set

5.4.1. Description

To further examine groundwater chemistry of the Hermannsburg Sandstone, chemistry data from the regional data set of bores used to examine the aquifer characteristics in the folded rock belt of the Amadeus Basin, presented in Table 4.1, were gathered. Chemistry data from these bores is presented in Table 5.2. These major ion data are from the Northern Territory Government database.

Figure 5.3, compares the chemistry of water at Palm Valley with around 40 other bores (data from Tables 5.1 and 5.2) drilled elsewhere in the Hermannsburg Sandstone within the folded rock province of the Amadeus Basin. This plot, which also shows the data from Palm Valley discussed above, confirms that the groundwater at Palm Valley is chemically typical of the Hermannsburg Sandstone. There is a slightly wider range of anion chemistry than that discussed above for the Palm Valley data set. For instance one bore RN 13652 plots (see also Table 5.2) showing chloride to comprise more than 50% of anion chemistry, while another RN 6949 shows bicarbonate to comprise 80% of anion chemistry. Apart from these anion outliers, the chemistry represented by this graphical technique is much the same as shown for the Palm Valley data set above. Cations are again seen to exist in roughly equal proportions.

SampleID	Depth	Date	A	TDS	Charge	TYPE*	Na	Mg	Ca	¥	Fe	HCO3	ច	SO₄	ő	SIO ₂	Alkali-	ы
	(m			(mg/L)	balance	(>10% meq)	(mg/L) (mg/L)	(mg/L) (mg/L) ((J/gm	(mg/L)	(J/gm	(mg/L)	(mg/L) ((mg/L)	nity (mg/L)
					%												(mg/L)	
Kaporilja Sp.	0	26/08/1984	7.6	500	0.21	Mg-Na-Ca-HCO3-Cl	62	39	51	∞	0.2	261	114	50	4	19	214	54
Lt. Palm Sp.	0	4/02/1993	7.5	255	-1.54	Na-Ca-Mg-HCO3-Cl	37	14	29	9	0.3	155	55	18	-	18	127	32
Oasis (sth) Sp.	0	4/02/1993	7.9	522	-0.01	Mg-Na-Ca-Cl-HCO3	76	43	37	7	0.1	227	152	39	2	14	186	45
PV 4/200	8	7/10/1994	8.4	462	-1.33	Na-Mg-HCO3-Cl	93	25	32	15		243	106	48	19	19	206	47
PV 1/608	8	27/09/1994			-100	CI							212					0
Road Sp.	0	19/09/1994	7.7	429	2.61	Mg-Ca-Na-HCO3-Cl	54	37	51	2		243	97	45			199	49
West Sp.	0	20/09/1994	8.4	436	-1.25	Mg-Na-HCO3-Cl	71	42	20	13		217	121	53			178	42
RN000052	56	25/09/1986	7.5	555	-1.37	Na-Mg-Ca-HCO3-Cl	73	38	59	10	2.0	275	135	67	9	12	225	57
RN000291	137	21/10/1981	7.7	460	-2.37	Ca-Mg-Na-HCO3-Cl	51	39	99	13	0.1	360	06	49	-	15	295	73
RN002830	109	10/03/1994	7.1	469	2.41	Ca-Mg-HCO3-Cl	39	37	84	11	0.4	334	20	63	-	15	274	76
RN002934	98	12/08/1998	œ	498	-1.97	Mg-Ca-Na-HCO3-Cl	64	36	57	7	0.1	275	110	72			226	54
RN003280	55	7/06/1971	7.3	640	-0.84	Mg-Ca-HCO3-Cl	55	65	85	7	0.1	434	141	54	8	19	356	93
RN003618	193	21/12/1982	7.5	970	0.38	Na-Ca-Mg-Cl-HCO3-SO4	189	48	92	13		307	260	216	5	12	252	64
RN004508	92	19/01/1988	7.1	748	3.22	Mg-Ca-Na-HCO3-Cl	06	56	06	13	0.1	406	150	11	-	13	333	92
RN005740	37	17/07/1970	8.1	610	-0.92	Mg-Ca-Na-HCO3-Cl	59	62	75	7	0.2	381	144	64	6	19	312	74
RN006949	85	11/11/1997	6.8	417	-3.82	Ca-Mg-HCO3	20	29	84	9	0.1	407	34	25			334	105
RN007175	171	21/12/1982	7.7	670	-2.66	Mg-Ca-Na-HCO3-C1-SO4	84	57	76	11	0.1	384	140	132	5	13	315	78
RN007291	165	12/08/1971	7.4	620	-0.69	Mg-Ca-Na-HCO3-Cl	59	59	63	6	2.1	351	134	99	5	15	288	74
RN007292	141	12/08/1998	7.8	419	-1.51	Mg-Ca-Na-HCO3-Cl	50	31	50	7	0.4	236	97	49			194	47
RN010209	244	1/05/1998	6.6	581	-1.94	Na-Mg-Ca-HCO3-Cl	97	41	52	7	0.1	319	134	81			262	94
RN010774	182	10/03/1994	7.5	458	6.01	Ca-Mg-HCO3-Cl	40	38	79	13	21.5	354	70	47	-	15	290	70
RN011338	122	10/10/1988	7.9	1010	-0.16	Mg-Na-Ca-Cl-HCO3	158	88	88	6	0.8	390	327	138	19	23	320	17
RN011341	129	17/03/1986	7.6	650	-0.57	Mg-Na-Ca-HCO3-Cl	80	20	52	2	0.5	387	135	56	47	19	318	79
RN011342	129	10/10/1988	7.8	930	0.22	Mg-Na-Ca-Cl-HCO3	128	06	60	9	0.4	392	287	109	45	16	321	78
RN011343	61	15/08/1994	7.6	729	2.26	Ca-Na-Mg-HCO3-Cl	06	39	117	7	0.2	389	163	75			319	80
RN011439	61	11/10/1988	7.8	475	-0.76	Ca-Mg-HCO3-Cl	41	40	71	8	0.8	361	17	41	-	6	296	72
RN011440	84	7/12/1976	7.5	792	-0.97	Ca-Na-Mg-HCO3-Cl	72	38	86	8	0.4	371	119	64	1	22	304	<u></u>
RN011446	137	10/10/1988	7.7	665	2.91	Mg-Ca-Na-HCO3-Cl	78	20	20	5	0.4	413	124	52	42	22	339	84
RN011447	153	10/10/1988	7.7	1030	-0.58	Na-Mg-Ca-Cl-SO4-HCO3	184	56	92	80	10.7	305	278	245	2	10	250	60
RN011448	122	25/08/1988	7.6	645	-0.32	Mg-Ca-Na-HCO3-C1	56	63	57	8	0.1	347	110	58	45	30	285	71
			0	-		-												

NOTE: Data from NTG database, ions and ${\rm SiO}_2$ and DIC all measured in mg/L. .

pH is Laboratory measure. Site refers to location area. Bore locations shown on Fig. 4.6: All sites interpreted to tap the Hermannsburg Sandstone. Highlighted sample IDs represent sites used for Palm Valley study (see Table 5.1). Additional site data shown in Table 4.1. * Water type classification is derived from the program Aquachem (1999) where major cations and anions are listed only when concentration in meq/L exceeds 10% of total.

SampleID	Depth	Date	P	TDS	Charge	э ТҮРЕ*	Na	Ng	ca	¥	Ъе	ő HCO	ច	So₄	ő	Slo,	Alkali-	ы С
	(u)			(mg/L)	balance	s (>10% meq)	(mg/L) (mg/L) ((mg/L) (I) (J/gr	mg/L)	(mg/L)	(mg/L)	(mg/L) ((mg/L) ((mg/L)	nity	mg/L)
					\$.0											(mg/L)	
RN011831	433	5/10/1986	7.5	770	2.6	Na-Mg-Ca-Cl-HCO3	135	58	71	15	1.0	369	230	61	ო	16	303	77
RN012022	200	2/11/1978	7.6	570	0.63	Ca-Mg-Na-HCO3-C1	54	43	75	16	0.6	348	66	51	22	22	285	71
RN012023	100	3/11/1978	7.4	710	0.14	Na-Mg-Ca-HCO3-Cl	105	47	67	14	0.3	320	176	88	e	18	262	67
RN012024	67	1/12/1994	7.8	559	-0.33	Mg-Ca-Na-HCO3-Cl	73	47	64	9	0.3	361	113	64			296	72
RN012025	200	15/11/1978	7.8	758	0	Mg-Ca-Na-HCO3-Cl	51	64	54	12	3.8	397	92	57	19	11	326	79
RN012026	168	10/10/1988	7.8	695	-0.44	Mg-Na-Ca-HCO3-Cl	11	75	62	S	1.3	431	134	58	51	20	353	86
RN012549	150	18/08/1994	8.3	743	3.08	Mg-Na-Ca-HCO3-Cl	105	76	72	7	0.1	424	182	11			348	81
RN012650	152	12/11/1997	7.6	568	-4.36	Mg-Ca-Na-HCO3-Cl	55	50	63	8	0.1	370	118	63			303	76
RN013652	110	30/04/1998	6.6	783	-1.98	Na-Mg-Ca-Cl-SO4	94	46	65	16	0.1	138	243	138			113	44
RN013653	330	12/04/1983	7.5	270	1.63	Mg-Ca-HCO3	24	25	41	7	1.2	218	36	22	7	12	179	45
RN014164	128	1/08/1997	7.2	475	0.77	Ca-Na-Mg-HCO3-C1-SO4	71	34	68	6	1.9	301	91	92			247	99
RN014165	166	12/08/1998	7.9	607	-2.57	Mg-Ca-Na-HCO3-C1	67	52	75	7	0.2	408	113	87			335	81
RN014166	220	6/12/1995	7.4	614	-0.95	Na-Ca-Mg-HCO3-Cl	110	29	64	10	1.9	287	152	89			235	60
RN014616	121	10/08/1994	8.1	438	3.25	Ca-Mg-Na-HCO3-C1	52	34	73	7	0.2	321	17	42			263	63
RN014952	195	6/10/1994	8.4	414	2.36	Mg-Na-HCO3-Cl	52	49	15	7	0.3	218	83	46			185	42
RN014956	. 62	15/08/1994	7.8	657	-0.73	Mg-Ca-Na-HCO3-C1	81	50	80	6	0.8	338	165	89			277	68
RN014957	112	16/08/1994	7.5	471	-0.6	Ca-Mg-Na-HCO3-C1	53	33	71	10	1.3	297	97	65			244	62
RN015006	250	3/03/1994	7.6	747	2.48	Na-Ca-Mg-Cl-HCO3-SO4	123	41	92	11	0.4	279	178	160	-	15	229	57
RN015007	220	12/08/1998	7.6	631	-2.59	Mg-Ca-Na-HCO3-C1	75	48	75	7	3.9	393	115	104			322	80
RN016359	193	16/02/1994	7.2	491	0.61	Mg-Ca-Na-HCO3-Cl	44	48	68	80	0.6	357	73	47	28	25	293	79

Table 5.2 Groundwater chemical data, Hermannsburg Sandstone regional data set

NOTE: Data from NTG database, ions and SiO $_2$ and DIC all measured in mg/L.

PH is Laboratory measure. Site refers to location area. Bore locations shown on Fig.4.6. All sites interpreted to tap the Hermannsburg Sandstone. Highlighted sample IDs represent sites used for Palm Valley study (see Table 5.1). Additional site data shown in Table 4.1. * Water type classification is derived from the program Aquachem (1999) where major cations and anions are listed only when concentration in meq/L exceeds 10% of total.



Figure 5-3 Piper plot of major cation chemistry from samples taken from the Hermannsburg Sandstone. Plot shows the Palm Valley area data from Table 5.1 (Fig. 5.1) compared with the Hermannsburg Sandstone regional data set compiled from the NTG database and presented in Table 4.2.

5.4.2. Major Ion chemistry

Figure 5.4 shows plots of some major ions against chloride concentration. Again, the analyses from Palm Valley are seen to be typical of Hermannsburg sandstone. No distinct correlation between chloride (commonly assumed to behave conservatively in groundwater) and the major ions, as would be expected if water chemistry were derived solely from the concentration of rainfall, is shown in Figure 5.4. The plot of sulphate against chloride (Fig. 5.4c) does, however, show a positive correlation trend. This may indicate that redox conditions are roughly similiar throughout the Hermannsburg Sandstone. Such relationships are commonly observed in groundwater from central Australia (e.g. Harrington, 1999; Jankowski and Jacobson, 1989; Wischusen, 1995b, 1998) and other arid regions (e.g., Schoeller, 1959; Mazor, 1991) and are taken to reflect the influence of water-rock interactions on groundwater chemistry.

5.4.3. Hydrochemical processes

5.4.3.1. Dissolution / Precipitation reactions

As previously discussed, (Section 5.3.1.), the uptake of CO_2 in the vadose zone and dissolution of carbonates and silicates also adequately accounts for the chemical variations seen in Figure 5.4. This point is illustrated in Figure 5.5a-d where the plots of Figure 5.2a-d previously discussed are shown with the additional regional data from Table 5.2. From this Figure it can be seem that saturation with respect to dolomite and calcite (Fig. 5.5a), magnesium to carbonate ratio in excess of dolomite dissolution stoichiometry (Fig. 5.5c), and sodium to chloride correlation roughly along a concentration of seawater trend (Fig. 5.5d), are regional phenomena within the Hermannsburg Sandstone.



Figure 5-4 (a-f) Hermannsburg Sandstone (Table 5.2) major ion versus chloride plots. (A); $Mg^{2+}v$. Cl⁻ (B); $Ca^{2+}v$ Cl⁻ (C); $SO_4^{2-}v$ Cl⁻ (D); $HCO_3^{-}v$ Cl⁻ (E); $NO_3^{-}v$ Cl⁻ (F); $SiO_2 v$ Cl⁻ (note Na⁺ v Cl⁻shown in Fig. 5.5d).



Figure 5-5 (a-d) aspects of groundwater chemistry from the regional Hermannsburg Sandstone data set plotted with Palm Valley area data. Plot of (A); saturation indices of calcite and dolomite (B); magnesium against bicarbonate (C); calcium against bicarbonate (D); sodium against chloride. Compare with Fig. 5.2.
There is a slight anomaly in Figure 5.5a where it can be seen that in contrast to the consistent regional trends, three bores plot as being under saturated with respect to calcite and dolomite. These bores, RN 13652, 10209, 6949 are unusual in that they alone record acidic water (pH < 7) in this data set, Table 5.2. The relationship to pH and carbonate saturation is shown in Figure 5.6a where these three bores are readily identified. As these bores have all been sampled relatively recently (Table 5.2) and all occur on Owen Springs Station (around 60km east of Palm Valley), this under saturation probably reflects variation in chemical conditions rather than incorrect pH being entered in the database. The reason for these anomalous lower pH readings is not readily apparent from examination of Table 5.2, however, as noted above, two of these bores (RN 13652, 6949) plot as outliers at either end of the HCO₃⁻ / Cl⁻ range on the piper diagram (Fig. 5.3). Given this atypical chemistry and pH, a detailed site and bore lithology investigation may be needed to confirm that these bores do indeed tap the Hermannsburg sandstone and not some other Amadeus Basin sequence.

5.4.3.2. Factors affecting pH

It is also possible that poor sampling protocol has caused a problem in pH calculation for the three Owen Springs Station bores, RN 13652, 10209, 6949. In most instances, water samples collected in the Northern Territory for standard major ion analyses are not filtered and acidified and rarely are field parameters recorded. While this system allows non-specialists to collect water so that the groundwater's suitability for stock or human consumption can readily be determined, such a sampling protocol may limit the accuracy of chemical information. For instance, iron is common in central Australian groundwater; levels of total iron are routinely assessed for water quality purposes (e.g. high iron can cause staining of washing or unsightly clouding etc.) in samples submitted to the laboratory, but virtually never at the time of sampling.

Hem (1985) notes the oxidation of ferrous iron to precipitate ferric hydroxide post sampling can lower pH prior to laboratory analyses. Thus, it is possible that some peculiarity of aquifer chemistry produces a more reducing environment with a consequent anomalous amount of dissolved ferrous iron in this region. Such an environment may have resulted in Ferric hydroxide precipitation post sampling. If so, this would lower pH and influence saturation calculations based on carbonate equilibria.



Figure 5-6 (a-e) Plots highlighting some possible hydrochemical processes in the Hermannsburg Sandstone. Plot (A); pH v calcite saturation indices (B); Alkalinity v dissolved inorganic carbon, a Deffeyes type plot (C); calcium v sulphate (D); divalent cations v sulphate plus bicarbonate plot, potential ion exchange trends annotated (E) plot Na⁺-Cl⁻ of against Ca²⁺+Mg²⁺-SO₄²⁻-HCO₃⁻.

For example, a plot (Fig. 5.6b) of DIC against alkalinity shows most of the Hermannsburg Sandstone data to follow a linear trend. Such a tight grouping is expected for samples within the pH range above 7 - 8.5 (Deffeyes, 1965). The Hermannsburg Sandstone data set (excluding the anomalous Owen Springs bores) has mean pH of 7.7, standard deviation of 0.31 and a range of 7.1 to 8.4 (derived from pH data Table 5.2). If the three anomalous bores are taken to be vertically displaced from the main Hermannsburg Sandstone group (Fig. 5.6b), this may indicate that a reduction in alkalinity due to a proton yielding reaction such as nitrification or sulphide oxidation has occurred (Stumm and Morgan, 1996).

Alternatively, it is also possible that this displacement is caused by addition of CO_2 and these three anomalous bores have been shifted horizontally from the main group due to an increase in DIC content (Deffeyes, 1965). As no ready source of CO_2 in addition to atmospheric flux is known for this area, it seems more likely that if these anomalous bores do represent a deviation from the main group, then some form of pH changing reaction is the likely cause.

5.4.3.3. Redox potential

Investigation of redox potential of these waters is not possible due to a lack of data; furthermore, proxy indicators of redox conditions such as nitrate concentration (e.g. Edmunds, 1995) are also not available. One potential indicator of a more reducing environment is water depth (i.e. the greater the depth the less connection with atmospheric oxygen). From Table 4.2 it can be seen that these three Owen Springs bores have deeper water levels than the mean of 34m below ground level, in fact bore RN 10209 has the deepest water level recorded at 120+ m. Thus, careful re-sampling and analysis of chemical parameters may also help explain why these three bores are anomalous compared to the rest of the Hermannsburg Sandstone data set.

5.4.3.4. Saturation indices

Saturation indices determined by the chemical modelling program PHREEQC (Parkhurst and Appelo, 1999) are shown in Table 5.3 for all the samples from the Hermannsburg Sandstone. These saturation indices are shown plotted against TDS in

Figure 5.7. Apart from the generally positive carbonate (Fig, 5.7a) and dolomite (Fig 5.7b) saturation indices discussed above it can be seen that all samples are under saturated in respect to both gypsum (Fig. 5.7c) and anhydrite (Fig. 5.7d). There is, however, a trend of increase in saturation indices with increase in TDS for these sulphate rich phases, which probably reflects the rough positive correlation between chloride and sulphate observed in Figure 5.7c above. Figure 5.7e shows that a small minority of bores in the Hermannsburg Sandstone are saturated with respect to siderite. The reason for this saturation is not known, however it is interesting to note that the bore at Areyonga RN 10774 (shown to be saturated with respect to siderite on Fig. 5.7e) was identified to have an iron fouling problem due to unfavourable well hydraulic characteristics allowing iron compounds to be precipitated in response to pumping (Wischusen, 1994b). Thus these siderite saturated bores may reflect disturbance of chemical equilibrium due to excessive pumping rates rather than natural variations in the chemical characteristics of the Hermannsburg Sandstone.

Figures 5.7f and 5.7g show the saturation indices of chalcedony and quartz respectively where it can be seen that while most samples are saturated in respect to quartz they are under saturated in respect to the amorphous chalcedony silica form, a situation that is not unusual for groundwater (Hem, 1985). While the source of silica is not known reactions such as the breakdown of feldspar shown in equation 5.4 are among the likely sources of silica input to groundwater (Hem, 1985).

SITE	Lable 5.3 5 CALCITE	bolomite	In Indices and av	/allable ∂`'O, ð'F GYPSUM	ANHYDRITE	es, Hermannsburg CHALCEDONY	Sandstone reg QUARTZ	ional data set δ^{18} O‰ δ^{2}	"М	5 ¹³ C‰
I			SA	TURATION INDE	x			VSMOW		PDB
Kaporilja Sp.	0.21	0.65	-1.08	-2.10	-2.32	90.0-	0.37			
Lt. Palm Sp.	-0.28	-0.53	-0.94	-2.64	-2.86	-0.09	0.34	-3.75	-28.9	-9.2
Oasis (sth) Sp.	0.31	1.03	-1.98	-2.34	-2.56	-0.20	0.23	-3.59	0.6	
PV 4/200	0.76	1.76		-2.28	-2.50	-0.08	0.35	-7.25	-45.8	
PV 1/608								-6.72	-46.7	
Road Sp.	0.29	0.79		-2.13	-2.35			-6.89	-44	-11.8
West Sp.	0.50	1.68		-2.46	-2.68			-4.29	-36.2	-11
RN000052	0.18	0.52	0.09	-1.93	-2.15	-0.26	0.17	-7	-47.7	-9.9
RN000291	0.55	1.22	-1.41	-2.02	-2.24	-0.17	0.26			
RN002830	0.02	0.04	-0.30	-1.81	-2.03	-0.16	0.26	-6.7	-52.5	-10.1
RN002934	0.66	1.47	-2.10	-1.91	-2.13					
RN003280	0.31	0.85	-0.85	-1.95	-2.17	-0.06	0.37			
RN003618	0.34	0.74		-1.36	-1.58	-0.26	0.17			
RN004508	0.10	0.34	-0.85	-1.77	-1.99	-0.23	0.20	-7.51	-49	-10.8
RN005740	0.98	2.23	-1.87	-1.92	-2.14	-0.07	0.36			
RN006949	-0.17	-0.45	-0.99	-2.18	-2.40			-7.94	-52.2	
RN007175	0.58	1.39	-1.40	-1.61	-1.83	-0.23	0.20			-10.6
RN007291	0.20	0.72	0.32	-1.95	-2.17	-0.16	0.26			
RN007292	0.37	0.88	-1.17	-2.09	-2.31					-9.3
RN010209	-0.71	-1.18	-1.27	-1.91	-2.13			-6.39	-44.1	-7.9
RN010774	0.40	0.84	1.21	-1.96	-2.18	-0.17	0.26			
RN011338	0.81	1.96	-0.87	-1.61	-1.83	0.02	0.45			
RN011341	0.35	1.18	-0.52	-2.13	-2.35	-0.06	0.37			
RN011342	0.73	1.82	-0.98	-1.69	-1.91	-0.14	0.29			
RN011343	0.69	1.25	-0.92	-1.66	-1.88					
RN011439	0.68	1.46	-0.69	-2.07	-2.29	-0.39	0.04			
RN011440	0.46	0.91	-0.49	-1.82	-2.04	00.0	0.43			

Page 1 of 2

SITE	CALCITE	DOLOMITE	SIDERITE	GYPSUM	ANHYDRITE	CHALCEDONY	QUARTZ	δ ¹⁸ O‰ δ ² H9	‰ 8 ¹	¹³ C‰
			SA	TURATION IND	EX			VSMOW		PDB
RN011446	09.0	1.54	-0.76	-2.05	-2.27	0.00	0.43			
RN011447	0.51	1.14	0.51	-1.32	-1.54	-0.34	0.09			
RN011448	0.35	1.10	-1.27	-2.05	-2.27	0.14	0.56			
RN011831	0.35	0.96	-0.10	-1.97	-2.19	-0.14	0.29	-7.48	-52	-6.9
RN012022	0.48	1.07	-0.48	-1.97	-2.19	0.00	0.43			
RN012023	0.18	0.55	-0.56	-1.80	-2.02	-0.09	0.34			
RN012024	0.62	1.45	-1.12	-1.94	-2.16			- 7.7-	-51.3	-9.4
RN012025	0.58	1.58	0.02	-2.08	-2.30	-0.30	0.13			
RN012026	0.65	1.74	-0.42	-2.06	-2.28	-0.04	0.39			
RN012549	1.17	2.71	-2.56	-1.90	-2.12					
RN012650	0.43	1.10	-1.24	-1.96	-2.18			- 7.8	-54.3	-9.17
RN013652	-1.00	-1.80	-1.60	-1.62	-1.84			-7.32	-53.3	-7.9
RN013653	-0.01	0.11	-0.21	-2.46	-2.68	-0.26	0.17	-7.31	-52.1	-7.31
RN014164	-0.02	0.00	0.34	-1.74	-1.96					
RN014165	0.82	1.83	-1.44	-1.77	-1.99					
RN014166	0.12	0.25	0.20	-1.78	-2.00			-7.24	-48	-8.74
RN014616	0.93	1.89	-1.94	-2.04	-2.25			-7.16	-45.8	ၐ
RN014952	0.39	1.64	-2.56	-2.64	-2.86					
RN014956	0.67	1.48	-0.73	-1.74	-1.96					
RN014957	0:30	0.62	-0.06	-1.86	-2.08					
RN015006	0.43	0.85	-0.76	-1.44	-1.66	-0.17	0.26			
RN015007	0.50	1.16	0.37	-1.69	-1.91					
RN016359	0.06	0.32	-0.09	-2.04	-2.26	0.06	0.49			

Page 2 of 2



Figure 5-7 (a-g) Saturation indices versus TDS: A) calcite SI; B) dolomite SI; C) gypsum SI; D) anhydrite SI; E) siderite SI; F) chalcedony SI; F) quartz SI.

5.4.3.5. Sulphate

Further chemical characteristics are discernible from these regional data shown in Table 5.3. While lack of field based measurement as discussed above may limit the accuracy of these data some regional trends are still apparent. For instance, all samples are under saturated with respect to gypsum. Given the low sulphate concentration in this aquifer (Table 5.2) and the fact that gypsum is more soluble than carbonates (Freeze and Cherry, 1979), it seems that dissolution of gypsum plays less of a role than carbonates in the hydrochemistry of the Hermannsburg Sandstone.

For instance, Figure 5.6c shows the relationship between calcium and sulphate. Nearly all analyses show that calcium exists in greater concentrations than would be expected were dissolution of gypsum the sole source of calcium to the system:

$$CaSO_4. 2H_2O \iff Ca^{2+} + SO_4^{2-} + 2H_2O$$
(5.6)

As mentioned above, dissolution of carbonates and silicates is a potential source of calcium in this aquifer. If gypsum and or anhydrite are not major salts in this system then additional sources of sulphate may also be a factor.

Sulphate forms a significant portion, often in excess of chloride, of inland rainfall composition (Hem, 1985). Indeed in central Australia Hutton (1983) and Keywood (1995) found sulphate to be in excess of chloride in rainfall analysed from Alice Springs. Furthermore, both these studies considered suspended dust provided a significant proportion of ions to rainfall and the land surface as dry fallout. Therefore, concentration of rainfall by evapotranspiration and dissolution of dry fall and root zone salts in the vadose zone could also provide a source of sulphate to this aquifer. Other sources are also feasible, for example, oxidation of pyrite can also produce SO_4^{2-} in groundwater.

5.4.3.6. Ion exchange

Another possible influence on cation chemistry is ion exchange between ions in solution and cation receptors such as clays. Figure 5.6d shows a composite ion plot of the divalent cations against carbonate and sulphate. The divalent cations are plotted as these ions are nearly always exchanged for monovalent cations like sodium or potassium held on ion absorbers such as clays (Freeze and Cherry, 1979, Hem, 1985). Cerling et al. (1989) have previously used this graphical method to investigate ion exchange processes. A 1:1 relationship is evident if dissolution of gypsum, dolomite and calcite are the dominant processes. If data plots below the 1:1 line an excess of alkalinity from silicate weathering (e.g. alkali feldspar weathering in the presence of CO_2 can also produce bicarbonate) reactions and/or a reaction that consumes Ca^{2+} and Mg^{2+} (e.g. ion exchange) is implied (Cerling et al., 1989). As Figure 5.5d shows Na^+ is not in excess of Cl⁻ there is no ready indicator of silicate weathering producing significant excess alkalinity.

Another method for examining ionic exchange processes is to plot Na⁺-Cl⁻ against $Ca^{2+}+Mg^{2+}-SO_4^{2-}-HCO_3^{-}$ where Na^+-Cl^- represents the amount of Na gained or lost relative to halite dissolution and $Ca^{2+}+Mg^{2+}-SO_4^{2-}-HCO_3^{-}$ represents the amount of Mg²⁺ and Ca²⁺ lost or gained relative to dissolution of calcite, gypsum and dolomite (Fisher and Mullican, 1997). If the relationship between these two ionic balance values is linear along a slope of -1 away from the origin then Fisher and Mullican (1997) note that cation exchange (e.g. Na^+ for Ca^{2+} and Mg^{2+}) may be important in controlling groundwater mineral composition. Such a plot, displayed as Figure 5.6e, shows the Hermannsburg Sandstone data to plot mainly about the origin (X, Y = 0, 0) though points are generally distributed along the line of -1 slope annotated on this diagram. Although some cation exchange may be indicated from this figure, the spread of data points does not extend very far from the origin like the data shown (origin to -40 along a -1 slope) by Fisher and Mullican (1997) as evidence for ion exchange in the USA. Thus while this plot shows some ion exchange may occur in the Hermannsburg Sandstone the cluster of data points about the origin suggests this process may not be significant. On examination of Figure 5.6d, it can be seen that ion exchange processes are also indicated to be unlikely to be a significant influence on cation chemistry as a rough 1:1 relationship is observed.

These plots (Fig. 5.6 e and d) do not imply, however, that other chemical processes such as dissolution of chlorite etc discussed above are not also influencing groundwater chemistry. Indeed the fairly consistent SiO_2 levels (see Table 5.2 & Fig. 5.5f) and saturation with respect to quartz (Table 5.3) is a possible indication that dissolution of silicates (e.g. Eq. 5.4) does occur in the Hermannsburg Sandstone.

5.4.3.7. Discussion

No matter what processes cause these observed variations in cation ratios, it is apparent that distinct water type groups can be recognised on the basis of cation signature. Examination of the analyses from the whole Hermannsburg Sandstone data set (Table 5.2) shows that all six types of water based on relative dominance of major cations exist. Frequency of these "cation" water type ranges are roughly 37% Mg-Ca-Na, 18% Mg-Na-Ca, 16% Ca-Mg-Na, 14% Na-Mg-Ca, 8% Na-Ca-Mg and 6% Ca-Na-Mg. Based on these cation ratio groups, five "cation" water types are represented in the data set used for study at Palm Valley (Table 5.1). From this grouping it is clear that the majority (55%) of sites sampled in the Hermannsburg Sandstone have Mg^{2+} as the dominant cation with Na⁺ and Ca²⁺ being equally distributed, each being the dominant cation at around 22% of sites.

In the absence of evidence to the contrary, it is possible that these observed differences in cation chemistry are the result of variation in chemical conditions experienced along differing groundwater flow paths through the Hermannsburg Sandstone. Evidence of differing conditions along various flow paths may be helpful in understanding the isotope chemistry of sites sampled in the Palm Valley region discussed below (Chapter 6 and 7).

It should be noted that the use of cation ratios to identify different flow paths is not without precedent. For instance, Edmunds et al. (2003) distinguish convergence of different flow lines with separate geochemical evolution in the Chotts area of Tunisia on the basis of observed differences in cation ratios between some bores sampled in the region.

5.5. Conclusion

Groundwater at Palm Valley and in the surrounding Hermannsburg Sandstone is generally of low salinity (<1000 mg/L) and bicarbonate rich. Comparisons with a regional data set shows groundwater at Palm Valley to be typical of the Hermannsburg Sandstone from the folded rock province (Lloyd and Jacobson, 1987) of the Amadeus Basin. A lack of field measured chemical parameters limits the chemical interpretations possible from the available chemical data. Nevertheless, it is apparent that apart from a few exceptions groundwater is saturated with respect to calcite and dolomite. Magnesium is usually the dominant cation, which apart from a credible carbonate dissolution origin, may also reflect reactions with magnesium rich accessory minerals such as chlorite and micas. Cation ratios are, however, variable throughout this aquifer. Possible processes that explain these differences can be readily identified. The actual processes that control cation chemistry variation are yet be determined. The variations in cation chemistry recorded are taken as evidence that different flow paths with separate geochemical histories have been sampled. This compilation of chemical data provides the basis for studying the stable isotope and radiocarbon hydrology of the Hermannsburg Sandstone at Palm Valley discussed in the following Chapter.

6. Stable Isotope and Radiocarbon Hydrology

6.1. Oxygen-18 and deuterium

6.1.1. Introduction

The isotope ratios of ¹⁸O/¹⁶O and ²H/H in a groundwater sample are measured as the relative difference in ratios to a reference sample by mass spectrometry. The ratios measured are expressed as delta (δ), which is the difference from an arbitrary standard known as the Vienna standard mean ocean water (VSMOW) in units of parts per thousand (‰) (Clark and Fritz, 1997).

Delta units (δ) for the isotope ratio (R) are expressed by the equation

$$\delta (\%) = \left(\frac{\text{R sample - R standard}}{\text{R standard}}\right) \times 1000$$
(6.1)

The δ^{18} O and δ^{2} H values for VSMOW are by definition equal to 0‰.

Changes in water isotope concentration (fractionation) occur when water undergoes phase changes, due to the differing behaviour of different isotopes of water. Thus when water moves through the hydrologic cycle from the sea, changes in δ^{18} O and δ^{2} H values occur. For instance when clouds are formed by evaporation of sea water the more common lighter molecules of water (e.g. H₂¹⁶O) are slightly easier to evaporate, thus clouds are depleted in ¹⁸O and ²H when compared to VSMOW (Gat, 1980). This fractionation affect is also influenced by temperature, thus rainwater (evaporated sea water) δ^{18} O and δ^{2} H composition also varies with latitude, (cold) high latitude rainfall is noted to be more depleted than (warm) tropical rainfall when compared to VSMOW (Gat, 1980). As movement of water vapour over continental land masses can experience cooler temperatures than at the coast (e.g cooler land surface or elevation change over mountains) inland rainfall δ^{18} O and δ^{2} H composition can be more depleted than coastal rainfall (Gat, 1980). However, as first noted by studying fresh water compositions by

Craig (1961), and later slightly modified after world wide analyses of rain water composition (Rozanski et al., 1993) most rainwaters in the world are observed to have δ^{18} O and δ^{2} H compositions that plot approximately along a straight-line relationship characterised by the equation:

$$\delta^2 H = 8.13 \ \delta^{18} O + 10.8 \tag{6.2}$$

which, when plotted on charts such as Figure 6.1, is known as the world meteoric water line (WMWL). Clark and Fritz (1979) note that while the meteoric relationships between ¹⁸O and ²H such as the meteoric water lines plotted on Figure 6.1 are due to fractionation due to condensation, it is a Rayleigh distillation process that causes the difference in warm and cold region rainfall isotopic composition. While the WMWL gives an approximation to rainfall in most regions, detailed studies can benefit from the construction of a local meteoric water line (LMWL), if data are available.

The comparison of groundwater δ^{18} O and δ^{2} H values to these meteoric water lines has various applications for the interpretation of groundwater (Allison, 1981). For example as fractionation of ¹⁸O and ²H is not known to occur within an aquifer under normal processes such as flow and plant root uptake, deviation from a meteoric line may indicate that some form of fractionation process such as evaporation or isotope exchange with mineral phases has occurred. The inset chart shown in Figure 6.1 outlines some of the processes that may cause deviation away from a meteoric water line (Fontes, 1980).



Figure 6-1 Stable isotope data – groundwater samples, Palm Valley. Relationship between $\delta^2 H$ and $\delta^{18} O$ for samples from the Hermannsburg Sandstone in the Palm Valley area. The local meteoric water line (LMWL) is obtained from the Alice Springs station (IAEA/WMO, 2001). While evaporation affects seem to influence isotope ratios at West and Little Palm Spring, the reason for Deuterium enrichment at Oasis Spring is not known (The likely influence of various processes shown as inset after Fontes, 1980).

6.1.2. Results

The δ^{18} O and δ^{2} H values presented for this study were determined by standard mass spectrometry at the CSIRO Isotope Laboratory in Adelaide, South Australia. These results, a LMWL for Alice Springs and the WMWL are shown drawn on Figure 6.1. The LMWL;

$$\delta^2 H = 6.9 \delta^{18} O + 4.5 \tag{6.3}$$

is constructed from the Alice Springs (150 km east of Palm Valley, Fig. 1.1) station of the global network of isotopes in precipitation station (IAEA/WMO 2001). Most of the groundwater samples plot on or near the LMWL. In the main cluster of data points (around $\delta^{18}O = -7\%$) one sample plots slightly to the right of the LMWL. This point, seismic shot hole 1/608, which has an electrical conductivity 1470 µS/cm, may have been more affected by evaporation during recharge than other sites and is thus slightly more enriched in ¹⁸O and ²H.

Oasis Spring plots above the LMWL. The reason for this is not known, though localised condensation affects are attributed to cause the shift of ¹⁸O and ²H above a LMWL on some water condensed on vegetation in Kenya (Ingraham and Matthews, 1988) and in some water condensed in Arctic caves (Lauriol and Clark, 1993). Clark and Fritz (1997) note that in low humidity regions re-evaporation of precipitation from local surface waters creates vapour masses with isotopic signatures that plot above local meteoric water lines. If such vapour recondenses before mixing with other atmospherically derived isotope reservoirs, the resulting water will also plot above the LMWL. It is feasible that such a situation may have occurred at this spring prior to sampling. The different anion ratio noted previously for this site also indicates evaporation processes have probably affected the nature of water at this spring when sampled.

Two data points, Little Palm Spring and West Spring, plot out of the main cluster and are significantly enriched in ¹⁸O and ²H when compared to the other samples. Given these samples are from free standing pools of water, enrichment due to evaporation can be expected.

Allison (1982) showed that when a free body of water is evaporated, subsequent enrichment of stable isotopes results in the evaporated water samples plotting to the right of the starting water stable isotope composition, when displayed on a δ^{18} O against δ^2 H diagram similar to Figure 6.1. The slope of the trend away from the initial isotope composition can be a function of humidity, salinity and temperature (Gat, 1981). Allison (1982) also showed that the maximum slope away from the meteoric water line that could occur for free standing evaporated water samples was along a slope of 5 (i.e. δ^2 H = 5 δ^{18} O) from the starting isotope composition. Thus, if a slope of five was projected through the West and Little Palm Spring plot points to intersect the LMWL, an estimate of the starting composition of the water prior to evaporation could be made. When a trend line between these two springs and the main cluster of data points is constructed, a line with a slope of around 5 (i.e. slope 4.7, Fig. 6.1) connects the two sets of data points.

As this line has a slope no greater than five it can be taken as evidence that spring and groundwater may share a common origin, that is, Little Palm and West Springs at the time of sampling contained evaporated Hermannsburg Sandstone groundwater and were not ponds of recent rainfall. The sample from Road Spring, which plots near the LMWL, may also indicate a groundwater origin of water at this spring. Thus, apart from Oasis Spring, it is concluded that stable isotope data confirms that the waters sampled in the Palm Valley area share a similar origin.

The fact that groundwater plots on or near the LMWL suggests little evaporation during the recharge process, which in turn suggests a rapid indirect recharge mechanism. Jacobson et al. (1989) have previously postulated that rapid recharge through fissures may be the dominant process in the folded rock province sandstones in the Amadeus Basin. If recharge were rapid into the Hermannsburg Sandstone, this would probably influence groundwater chemistry in that dissolution and weathering rather than precipitation would be expected in the vadose zone.

Figure 6.2 compares some oxygen/deuterium data that has been previously sampled from within the regional data sets of bores. These data, presented in Table 5.3, were sourced from the NTG database and originally published by Jacobson et al. (1989).



Figure 6-2 Relationship between $\delta^2 H$ and $\delta^{18}O$ for samples from the regional Hermannsburg Sandstone data set (Table 5.3) and the Palm Valley area (Fig. 6.1). The LMWL is obtained from the Alice Springs station (IAEA/WMO, 2001). $\delta^2 H$ and $\delta^{18}O$ values (apart from some springs) from the Palm Valley area are seen to be similar to other Hermannsburg Sandstone sites.

This plot shows that the Palm Valley study samples plot in similar position to other bores sampled from the regional Hermannsburg Sandstone data set and that given little evidence of evaporation (deviation from LMWL) the postulated rapid recharge mechanism of Jacobson et al. (1989) seems likely for Hermannsburg Sandstone at Palm Valley and elsewhere in the region.

Figures 6.3a and 6.3b show the relation between δ^{18} O and δ^{2} H against chloride respectively. From these figures it can be seen that stable isotopes do not vary with chloride content which also suggests that evaporation is not responsible for the stable isotope composition except at some springs sampled at Palm Valley. These Figures also suggest that chloride content variations within the Hermannsburg Sandstone may result from differing amounts of surface or vadose zone chloride entering the aquifer at the time of recharge at different locations and/or times, rather than differing amounts evaporation. Alternatively it may be that evaporation concentration of chloride is an active process at recharge but its affect is not sufficient to noticeably influence stable isotope composition.

6.1.3. Stable isotopes: a proxy for groundwater age?

It should also be noted that the stable isotope data from the Hermannsburg Sandstone give no hint as to time of recharge as ${}^{2}\text{H} / {}^{18}\text{O}$ depletion is similar in magnitude to that found in aquifers known to contain modern water elsewhere in central Australia (e.g. Wischusen, 1994a). Elsewhere in the world stable isotopes can be used to draw inferences on groundwater age. For instance, Guendouz et al. (2003) note that groundwater believed to have been derived from Pleistocene rainfall in northern Africa is often distinguished from that derived from modern and Holocene rainfall as stable isotopes were more depleted during the Pleistocene. Zuppi and Sacchi (2004) note this phenomenon is widespread in the large sedimentary basins of Europe (e.g. Edmunds and Smedley, 2000) as well as northern Africa (e.g. Guendouz et al., 2003; Rozanski et al., 1993). This depletion of ${}^{18}\text{O}$ compared to modern rainfall is attributed to lower temperatures during the recharge processes during humid periods of the Quaternary.



Figure 6-3 Relationship between stable water isotopes and chloride. Plot (A); δ^{18} O versus chloride, (B); δ^{2} H versus chloride.

They then note that stable isotopes may be used as a proxy to identify palaeo waters over the last 70 ka in this location.

In contrast, Plummer (1993), while noting the widespread correlation of cooler temperatures and depletion of ¹⁸O and ²H values for Pleistocene recharged groundwater, presents data from southern U.S.A. where this depletion phenomenon maybe masked by associated changes in the season of recharge during the Pleistocene.

While temperature dependant amino-acid racemization reactions have been studied in fossil emu egg shells at Lake Eyre to interpret temperatures were some 9 degrees colder during the last glacial maximum in the arid interior of Australia (Miller et al. 1997), relations of ¹⁸O depletion to temperature and hence age of groundwater are not usually drawn in central Australia. This is because precipitation is so variable in the region that the amount effect (Dansgaard, 1964) is thought to dominate stable isotope composition. Rozanski et al. (1993), show that amount of precipitation and not temperature variation is the dominant influence on stable isotope composition at various locations (e.g. Guam and India) affected by monsoon rain events. Given the monsoon (Section 3.2) also affects rainfall in central Australia it is not surprising an amount effect is thought to dominate in central Australia. This is certainly the case at Alice Springs where the ¹⁸O depletion of rainfall is seen to vary from a mean of -2.5% for light rainfall months (< 50mm) to > -13.0% for heavy rainfall (200mm) months (calculated from International Atomic Energy Agency/World Meteorological Organization data by Harrington et al. 2002.) This correlation of δ^{18} O to monthly rainfall amount has been used to study the character of modern recharge rain in several previous studies in central Australia (e.g. Wischusen, 1994a, Calf et al. 1991).

While the amount effect limits a ready characterization of age of groundwater based on stable isotopes in the Hermannsburg Sandstone, it is not evidence that this groundwater is derived from modern rainfall. The fact that groundwater plots on or near the LMWL suggests little evaporation during the recharge process, which in turn suggests a rapid indirect recharge mechanism. Jacobson et al. (1989) have previously postulated that rapid recharge through fissures may be the dominant process in the folded rock province sandstones in the Amadeus Basin and speculate that this may have occurred during some previous climatic regime. If recharge were rapid into the Hermannsburg

Sandstone, this would probably influence groundwater chemistry in that dissolution, rather than precipitation, would be expected in the vadose zone.

6.2. Radiocarbon (¹⁴C)

6.2.1. Background

Radioactive carbon-14 (¹⁴C) is formed naturally in the earth's atmosphere by the bombardment of nitrogen-14 by cosmic rays. Carbon-14 has a half-life of 5730 years. Its activity decays exponentially, the law of radioactive decay describes the rate at which the activity of radiocarbon decreases with time. This is expressed by the decay equation:

$$A_t = A_0 e^{-\lambda t} ag{6.4}$$

where: A_t = Activity at time *t*, A_0 =Activity at time 0 (recharge), λ = decay constant for radio isotope (${}^{14}C = 1.2097 \times 10^{-04} \text{ yr}^{-1}$).

Under natural conditions new carbon-14 forms as older (around 40 ka) ¹⁴C atoms lose their radioactivity (become radioactively dead). Therefore, given a sufficient sample of carbon atoms, the mix of carbon-14 atoms in various states of decay should remain constant, thus atmospheric carbon has a constant carbon-14 activity. This atmospheric (modern) carbon activity is used as the benchmark for radiocarbon analyses with measured activity expressed as percent modern carbon (pmc).

Due to the changes in the atmosphere since the industrial revolution (i.e. release of old dead carbon with use of fossil fuels and increase in natural levels of ¹⁴C following surface nuclear testing in the 1950's and 1960's) the benchmark modern carbon activity is taken to be that of atmospheric carbon prior to the Industrial Revolution. Current atmospheric levels are now well above 100 pmc (Kalin, 2000).

Radiocarbon dating of groundwater operates on the assumption that once water enters an aquifer having picked up atmospheric carbon from soil CO_2 etc, no further addition of radiocarbon from the atmosphere is expected, thus carbon-14 activity of a groundwater sample should be related to the amount of time elapsed since recharge occurred

6.2.2. Carbon 13

Carbon 13 isotope content of groundwater used to be expressed as parts per thousand from the Peedee Belemnite (PDB) standard of $\delta^{13}C = 0^{\circ}/_{oo}$, (measured from CO₂ prepared from Belemnites collected from the upper Cretaceous Peedee Formation of South Carolina, USA; Bates and Jackson. (1987). Due to a decline in the reserves of this standard Belemnite the International Atomic Energy Agency in Viena have set a hypothetical standard (VPDB, virtually identical to PDB) beside which $\delta^{13}C$ determinations are reported (Clark and Fritx, 1997). One of the common uses of carbon-13 data is to give an indication of the behaviour of ¹⁴C in groundwater. For example as ¹⁴C and ¹³C behave very similarly, if a groundwater sample has a lower than expected $\delta^{13}C$ content, then some dilution of radiocarbon activity can be suspected.

The δ shift of ¹³C from the standard can, like ¹⁸O and ²H, be caused by fractionation processes. The change of gaseous CO₂ to aqueous carbonate species when rainwater passes through the soil to an aquifer can theoretically cause around + 8 ⁰/₀₀ fractionation of the soil δ^{13} C value (e.g. Hendy, 1971; Carothers and Kharaka, 1980; Mook et al., 1974; Salomons and Mook, 1986). Reaction of soil CO₂ with aquifer carbonates (e.g. Eqs. 5.1 & 5.3) also enriches the δ^{13} C content of dissolved inorganic carbon (DIC) in groundwater as non marine carbonates are expected to have δ^{13} C values near 0‰ (Herczeg et al., 1991). The extent of this enrichment depends on the availability of mineral carbonate, the partial pressure of soil CO₂ and whether the system is open to the atmosphere (Clark and Fritz, 1979). Thus the δ^{13} C of DIC in groundwater is influenced by several factors including the extent of aquifer carbonate dissolution which influences radiocarbon interpretations (see below). In studies of carbonate behaviour in groundwater it is common to plot the relationship between δ^{13} C and DIC (e.g. Schofield and Jankowski, 2004). For instance if δ^{13} C values become more enriched as DIC content increases this may be evidence of an increased component of rock carbonate derived carbon entering the groundwater system.

Figure 6.4a shows a plot of δ^{13} C against DIC for the available analyses from the Hermannsburg Sandstone data set (Table 5.3) where it can be seen that no strong correlation exists. Thus these data allow no obvious conclusion as to the effect of rock or matrix carbonate dissolution on the stable carbon isotope content and hence also no readily apparent potential dilution of radiocarbon activity.

In some other Australian studies (e.g. Calf, 1978 and1979) the method of plotting δ^{13} C against 1/total carbonate outlined by Pearson and Swarzenki (1974) has been used to estimate soil and mineral δ^{13} C content by extrapolating trends to intersect the δ^{13} C axis at zero reciprocal carbonate and also to derive estimates of soil gas δ^{13} C from the slope of the line of best fit. When δ^{13} C is plotted against 1/DIC as shown in Figure 6.4b it can be seen that the spread of plot points (as noted above in Fig. 6.4a) does not allow a distinct trend to be drawn, hence the use of the Pearson and Swarzenki (1974) technique is not readily applied here.

Furthermore, Allison (1981) gives some examples on the use of δ^{13} C data to correct age determination with ¹⁴C activity data, and also notes that the amount of fractionation of ¹³C can vary in response to water pH and temperature changes, therefore the accuracy of corrections based on ¹³C may be limited.

6.2.3. Corrections

The major complication in determining radiocarbon age is adequately allowing for dissolved groundwater carbon reactions with soil and aquifer carbon reservoirs, such as aquifer and soil carbonates and soil CO₂ gas. Interaction between water and subsurface carbon reservoirs has the potential to significantly lower the initial ¹⁴C activity (A₀). For example, recharging water reacting with CO₂ gas and CaCO₃ in the subsurface as shown in Eq. 5.3 would produce groundwater bicarbonate with both solid carbonate and CO₂ derived carbon atoms. As subsurface carbonate minerals usually contain old "dead" carbon (¹⁴C activity = 0 pmc), then, depending on the amount of carbonate dissolution,

groundwater ¹⁴C activity can be expected to be less than atmospheric carbon ¹⁴C activity, even at the time of recharge.

Various methods exist that attempt to account for carbon exchange processes so as to estimate the initial groundwater activity A_0 at the time of recharge. Some of the commonly used methods to estimate A_0 for this study have been applied; the empirical approach of Vogel (1967), Tamers' chemical mass balance model (Tamers, 1967), Pearson's isotope mass balance model (Ingerson and Pearson, 1964) and the Fontes and Garnier (F-G) chemical mass- balance with isotope exchange model (Fontes and Garnier, 1979).

Due to a lack of data, several commonly used assumptions are necessary to apply these models; the atmospheric and mineral carbon activities were taken to be 100% and 0% (pmc) respectively, δ^{13} C composition of soil gas CO₂ and matrix carbon was taken to be -20%, and -1.5% PDB respectively. The matrix carbon δ^{13} C composition is taken from measurements made of calcite veins within the Hermannsburg Sandstone by Jacobson et al. (1989). No soil gas CO₂ measurements are available in this region so a value representing a possible mix of carbon fixing mechanisms expected for arid regions (Clark and Fritz, 1997) was adopted. This value is considered a reasonable estimate as a soil gas CO₂ value of around -20% was also assumed by Love at al. (2000) in the arid north of South Australia.

Reported measured soil gas CO_2 values from other arid areas (e.g. Murray Basin, Australia, -18‰, Leaney and Allison 1986; Karmel mountains, Israel, -22‰ Bajjali and Abu-Jaber, 2001; Continental Terminal aquifer, Niger, -14 to 21‰, Le Gal La Salle et al. 1995; Continental Terminal aquifer, Niger, -20‰ in soil under native vegetation, Le Gal La Salle et al., 2001; West Texas, -17‰ and Russia, -18‰, cited by Calf, 1979) also indicate -20‰ is a reasonable assumption.

Furthermore, as enrichment of soil CO₂ gas δ^{13} C values occur during hydration of CO₂ and also with increasing carbonate dissolution towards calcite saturation (Clark and Fritz, 1997), it seems reasonable to assume a soil CO₂ gas δ^{13} C value significantly depleted compared to measured groundwater values (Table 6.1) here.



Figure 6-4 Relationship between $\delta^{13}C$ and dissolved inorganic carbon DIC. Plot (A); $\delta^{13}C$ versus DIC, (B); $\delta^{13}C$ versus 1/DIC.

The speciation capabilities of the PHREEQC (Parkhurst and Appelo, 1999) model were used to compute the dissolved inorganic carbon (DIC) carbon content of samples.

6.2.4. Results - Radiocarbon age

The results of carbon–14 and carbon–13 analyses from water samples taken from the Hermannsburg Sandstone at Palm Valley are shown in Table 6.1. The age of groundwater, as calculated using the different models discussed above, is also presented for comparison in Table 6.1. As assumptions have been used to calculate these results, the reliability of absolute ages determined is questionable. Nevertheless, comparison of relative ages is interesting in that a significant reduction in initial ¹⁴C activity is indicated, particularly by the F-G model. The chemical mass balance component of the F-G model uses cation concentrations to determine the amount of matrix (¹⁴C dead) derived carbonate in solution. The greater the proportion of matrix derived carbonate, the lower A_0 . Given, however, that Mg^{2+} (Eq. 5.2) and Ca^{2+} (Eq. 5.5) maybe derived from sources other than carbonate dissolution (Section 5.4), the portion of matrix-derived carbonate may be overestimated, and hence the F-G model may underestimate groundwater ages.

Furthermore, Harrington (1999) and Harrington and Herczeg, (2000, 2003) note in a study of groundwater in the Cainozoic Ti Tree sedimentary Basin (250 km north East of Palm Valley) that existing ¹⁴C correction schemes may not be applicable in some arid settings. This was considered because comparison of strontium isotope (87 Sr/ 86 Sr) ratios in groundwater with those of local carbonate and silicate minerals indicated that a significant proportion of groundwater HCO₃⁻ may be derived from silicate weathering (e.g. Eqs. 5.4 and 5.5). As such a source of dead carbon is not adequately accounted for by the standard correction models, groundwater ¹⁴C ages may be younger than calculated using the standard correction methods if silicate hydrolysis is significant. Given that dissolution of silicates has been identified as a possible process (Section 5.3.2.2) it may be that corrected ¹⁴C ages are similarly affected in the Hermannsburg Sandstone.

SampleID	¹⁴ C (pmc)	δ ¹³ C‰ (PDB)	DIC* (mmol/L)	Uncorrected age (years)	Tamers (1967) age	lngerson & Pearson (1964) age	Vogel (1967) age	Fontes & Garnier (1979) age
RN004508	3.1	-10.8	7.65	28716	24008	23009	27373	22819
RN010774			6.57					
RN012024	22.5	-9.4	5.84	12331	7080	5267	10987	5656
RN014165			6.73					
RN014616	4.2	6-	5.22	26206	20597	18709	24862	16966
RN014952			3.61					
PV04/200			4.01					
ROAD Sp	42.4	-11.8	4.10	7093	1667	2234	5749	modern
L PALM Sp	87.2	-9.2	2.69	1132	modern	modern	modern	modern
OASIS			3.76					
WEST Sp	79	-11	3.47	1949	modern	modern	605	modern

* DIC Dissolved inorganic Carborn calculated using PHREEQC (Pankhurst and Appelo, 1999)

Another potential complication to the results shown in Table 6.1 is the fact three of the samples are from surface water pools at springs, this may mean gaseous exchange between DIC and atmospheric CO_2 has occurred. This could then elevate the ¹⁴C activity of any "old" groundwater discharging at these springs. The fact that ¹⁴C activities at these springs are well below the current atmospheric ¹⁴C activity of > 100 pmc (Kalin, 2000) suggests at least that these springs are groundwater derived rather than ponds of recent rainfall.

It should also be noted that the two bores RN 4508 and RN14616 are at or near background ¹⁴C activities, for instance Love et al. (2000) note that < 4% (pmc) is considered the limit of detection for the direct adsorption method. Given that significant drawdown can be expected from low k aquifers (e.g. Theis, 1940), it is possible that aeration during pumping (production pumps used for sampling) has resulted in slight gaseous exchange with atmospheric CO₂. Furthermore, Walker and Cook (1991) have also noted that vertical diffusion of atmospheric ¹⁴C may increase radiocarbon activity in low recharge arid zone aquifers. Geyh (2000) also notes that diffusion of modern carbon is significant when old groundwater rises towards the surface, so much so that Pleistocene groundwater may erroneously carry a Holocene ¹⁴C signature in some instances. Thus, it may well be that groundwater at these bores is in fact older than the detectable limit of radiocarbon (>30 ka).

Given these potential shortfalls in estimating groundwater ¹⁴C ages and lack of knowledge of the exact chemical composition and process occurring, it may be better to think of the radiocarbon technique as a only a semi quantitative indicator of groundwater age in the Hermannsburg Sandstone.

Apart from these considerations of absolute radiocarbon age determination, results in Table 6.1 show groundwater age may vary across, or even exceed, the whole radiocarbon detection range. Consequently, it was decided to further examine the age of groundwater using a radionuclide with a longer half-life than 14 C.

7. Chlorine-36

7.1. Chlorine – 36 in the environment

7.1.1. Introduction

Chlorine-36 is an unstable isotope, production in the atmosphere is by cosmic-ray interaction, mainly with argon in the stratosphere and fallout shows a strong latitudinal dependence. The greatest atmospheric fallout occurs at mid-latitudes. The atmosphere also contains stable chloride (35 Cl and 37 Cl) derived from sea-spray, and from remobilised terrestrial salts. Chlorine-36 mixes with stable chloride in the atmosphere and this mixture of isotopes then travels the hydrologic cycle as the chloride ion. Chlorine-36 is also produced by cosmic ray interaction with near-surface rocks (epigene production) and hence may be released by weathering. Chlorine-36 is also produced in the deep subsurface (hypogene production) through neutron-capture on 35 Cl (the neutron flux arising from the decay of natural radioactivity), producing a low and roughly constant level of background chlorine-36 to groundwater systems (Bentley, et al., 1986a). Figure 7.1 shows a schematic of 36 Cl and Cl⁻ in the hydrological cycle which illustrates some of these processes.

The ratio of chlorine-36 to stable chloride (³⁶Cl/Cl) can be measured readily using accelerator mass spectrometry on a few milligrams of chloride precipitated from groundwater as silver chloride (Elmore et al., 1979) and varies in natural systems from around a thousand parts in 10¹⁵ of total chloride, down to a background of a few parts in 10¹⁵. The fact that ³⁶Cl is radioactive and has a half life of around 300 ka means that this radio isotope offers the potential to date groundwater to over a million years (Bentley et al., 1986a). The conservative nature of the chloride ion chemistry, hence also ³⁶Cl, in groundwater suggests use of this isotope for dating old groundwater should be widely applicable (Fontes and Andrews, 1994).

Age dating groundwater with radioisotopes is based on the principle that once water reaches the saturated zone it is closed off from the atmosphere. Thus any measured radioisotope activity in groundwater that is less than atmospheric activity at the time of recharge, can be taken to indicate radioactive decay, and can then be used to estimate groundwater age or residence time. The law of radioactive decay (as previously described in Eq. 6.4) describes the rate at which the activity of a radioisotope decreases with time. For radio chlorine this is expressed by the decay equation:

$$A_t = A_0 e^{-\lambda t} \tag{7.1}$$

where: A_t = Activity at time *t*, A_0 =Activity at time 0 (recharge), λ = decay constant for ${}^{36}Cl = 2.3028 \times 10^{-06}$.

Thus, if activity at the time of recharge (A_0) is known, measured groundwater activity can be used to compute age by rearranging the standard decay equation as;

$$t = \frac{-1}{\lambda} \left(\ln \frac{A_t}{A_0} \right) \tag{7.2}$$

where A_t is the activity for the groundwater sample, A_0 is the radioisotope activity at time of recharge, t is the period of radioactive decay in years and $1/\lambda$ is the mean life of the radioisotope (${}^{36}Cl = 434,251$ yrs).

The direct application of equating groundwater age with observed radioactivity is not, however, necessarily straightforward. Some of the considerations and pitfalls of applying radioactive tracer techniques are discussed in detail in specialist texts such as Fritz and Fontes, (1986), Clark and Fritz, (1997), Mazor, (1997) and Cook and Herczeg, (2000).

For groundwater age studies using chloride-36, the amount of ³⁶Cl in the atmosphere in relation to stable chloride needs to be established at the time of recharge. Commonly the ratio of ³⁶Cl to stable chloride in rainfall (³⁶Cl/Cl) is assumed to represent the initial or input radioactivity of chloride-36. The determination of the natural (input) rainfall ³⁶Cl/Cl ratios is a major undertaking in its own right and is not yet clearly defined in all parts of the world (Phillips, 2000).



patterns. The evolution of ³⁶Cl is illustrated in 3 cases where R₀ is the input and R_e is the secular equilibrium ratio of ³⁶Cl/Cl. Case 1 shows in situ activation of ³⁵Cl in U rich environments from neutron flux. Case 2 shows decay of the input ratio along flow paths without CI addition in low U sediments. Case 3 shows the affect Figure 7-1 A schematic of ³⁶Cl in the hydrological cycle. Marine Cl is depleted in ³⁶Cl as the residence time of Cl in the oceans greatly exceeds the 301 ka half life of ³⁶Cl. Dilution of cosmogenic ³⁶Cl by ³⁵Cl and ³⁷Cl derived from marine aerosols deceases with distance from the coast and also varies with weather hemisphere. Reprinted from Nuclear Instruments and Methods in Physics Research B, vol 92, Fontes and Andrews 1994 "Accelerator mass spectrometry in of subsurface addition of Cl⁻ on ³⁶Cl/Cl ratio. Note a) similarity of response in case 2 and 3 and b) Cosmogenic ³⁶Cl fallout varies with latitude and possibly hydrology" p 367-375 © 1994 with permission from Elsevier.

Three further complexities also exist: this input ratio may have varied in response to changes in the earths magnetic field in the past (Plummer et al., 1997). Atmospheric testing of nuclear weapons in the 1950 and 1960 produced a short lived "bomb" peak of ³⁶Cl many orders of magnitude above natural ³⁶Cl levels (Suter et al., 1987). Chlorine-36 can also be produced by subsurface nuclear reactions associated with neutron production from uranium-thorium decay; when this production is significant compared to decay concentrations of atmospheric derived Cl, interpretation of groundwater age is difficult.

Many previous ³⁶Cl groundwater studies have found evidence that chloride additions post recharge have complicated interpretation of measured ³⁶Cl/Cl ratios. For example dissolution of old chloride with no ³⁶Cl can lower the ³⁶Cl/Cl ratio of groundwater, and indicate an age greater than that derived from radioactive decay of ³⁶Cl. Given the potential complexities in calculating which mechanism is responsible for measured ³⁶Cl/Cl ratios, it is apparent that all sources and sinks of chloride to an aquifer should be determined and allowed for when estimating groundwater age using the ³⁶Cl decay method.

A background to the processes affecting ³⁶Cl distribution in the environment is presented below.

7.1.2. Measurement

The abundance of ³⁶Cl in the environment is very rare with activity in natural samples ranging from 1 by 10⁻¹⁶ to 1 by 10⁻¹⁰ ³⁶Cl/Cl. Most samples, however, record ³⁶Cl/Cl ratios near the lower end of this range (Bentley et al., 1986a). Indeed chemical data reference texts reflect this rarity as they only list the abundances of ³⁵Cl and ³⁷Cl (at 75.5% and 24.5% respectively). The measurement of the ratio ³⁶Cl/Cl only became readily feasible with the advent of Accelerator Mass Spectrometry (AMS) techniques in the late 1970s (Elmore et al., 1979). The ³⁶Cl/Cl ratio is commonly measured in natural groundwater systems from a background level of 1 to over 1000 parts in 10¹⁵ of total chloride. However, fallout from atmospheric nuclear testing may result in ³⁶Cl/Cl ratios of over 5000 parts in 10⁻¹⁵ being measured in some systems (e.g. Davis et al., 2001).

Most ³⁶Cl analyses conducted for studies in Australia are run at the Australian National University (ANU) in Canberra. This is one of the four centres in the world which routinely determine ³⁶Cl on water samples, (Phillips, 2000). Fifield et al. (1987, 1990, 1994) describe the 14 million-volt 14UD accelerator AMS system used at the ANU, Canberra, Australia. The technique involves converting a small sample (of only a few milligrams) into ions, and then accelerating these ions in the 14UD particle accelerator. As the ions emerge from the accelerator, they are separated by magnetic and electrical fields according to their mass, and then counted by various detectors (Fifield et al., 1987). Analysis of ³⁶Cl from water samples requires chloride to be precipitated out as a pure silver chloride, low in sulphur. This solid is then subjected to a beam of Cs ions which then provides the source of Cl⁻ ions for acceleration through the AMS, and subsequent analysis of ³⁶Cl isotope content. The ratio of chlorine-36 to total chloride is derived from the observed currents of stable Cl ion beams and the counting rate of ³⁶Cl ions measured at a detector (Fifield et al., 1987).

By convention, this ratio is reported as ³⁶Cl/Cl by 10⁻¹⁵ which give values from 0 to several thousand for the amount of ³⁶Cl typically found in the environment. Fifield et al. (1994) note that repeat measurements on reference standards show system variations contribute at most a 2% standard deviation to ³⁶Cl/Cl ratios determined at the Australian National University AMS. When total chloride content of the water sample is determined separately by standard analytical techniques such as titration by silver nitrate with potassium chromate indicator (APHA, 1998) or ion chromatography, the number of ³⁶Cl atoms per litre can also be calculated:

$$\frac{{}^{36}Cl}{Cl} \times mCl \times A^{\circ} = {}^{36}Cl \text{ atoms per litre}$$
(7.3)

where; ${}^{36}Cl/Cl$ is the ratio determined by AMS, *mCl*⁻ is the number of moles of Cl⁻ determined by standard analytical techniques, A^{o} is Avogadro's number i.e. around 6.022 by 10²³. This concentration of ${}^{36}Cl$ is often multiplied by a factor of 10⁷ to give values between 0 and 100 (Clark and Fritz, 1997).

It is common practice to report results as both the ³⁶Cl/Cl ratio measured by AMS and as the number of ³⁶Cl atoms per litre calculated. This is because different processes can

be examined by each result. For example when comparing waters that have undergone differing amounts of evaporation the 36 Cl/Cl ratio does not change during concentration by evaporation whereas the concentration of 36 Cl atoms will (Clark and Fritz, 1997).

The relationship between ³⁶Cl atoms, ³⁶Cl/Cl and total Cl varies for different processes that may affect groundwater. The effect of various processes on ³⁶Cl and Cl levels is shown for various combinations of these parameters on the charts presented in Figure 7.2. Many ³⁶Cl decay dating studies report results in a similar format to one or more of these charts.

7.1.3. ³⁶Cl production

7.1.3.1. Atmospheric fallout

Chlorine-36 is formed in the atmosphere by spallation reactions of atmospheric gases with cosmic radiation. The spallation reaction of 40 Ar is the most important reaction;

$${}^{40}\text{Ar} + p \rightarrow {}^{36}\text{Cl} + n + \alpha \tag{7.4}$$

Where: p = proton, n = neutron and $\alpha = alpha particle$.

The neutron activation reaction of ³⁶Ar to produce ³⁶Cl was once thought to contribute around 33% of ³⁶Cl production in the atmosphere (Bentley et al., 1986a). However, subsequent studies cited by Andrews and Fontes (1992) have since shown negligible ³⁶Cl is produced by this reaction;

$${}^{36}\mathrm{Ar} + \mathrm{n} \rightarrow {}^{36}\mathrm{Cl} + \mathrm{p} \tag{7.5}$$

Where n = neutron; $\rho = proton$.

Most (>60%) of the atmospheric 36 Cl is produced by spallation in the stratosphere (i.e. the layer of atmosphere above about 12 km from the surface of the Earth at mid latitudes) with the rest produced by spallation in the underlying troposphere layer.



CI (mg/L)



Figure 7-2 The affect of various processes on ³⁶Cl and Cl⁻ levels illustrated by a) ³⁶Cl/Cl versus Cl⁻ and b) ³⁶Cl/Cl versus ³⁶Cl atoms. After Bird et al., 1991.

Variability in the Earths magnetic field, hence also variability in cosmic radiation shielding efficiency, means spallation production of ³⁶Cl varies in relation to geomagnetic latitude within the stratosphere (Fontes and Andrews, 1994).

Due to various atmospheric circulation processes and seasonal adjustments of troposphere thickness, much of the air mass in the stratosphere transfers to the troposphere at mid (35-40⁰) latitudes (Keywood, 1995; Hainsworth et al., 1994). As a result, the ³⁶Cl formed in the stratosphere mixes with aerosols comprised of marine (and terrestrial) derived Cl along with ³⁶Cl produced in the troposphere. These aerosols only survive in the atmosphere for about a week (Andrews and Fontes, 1992) and then fall out as either dry or wet (rain) deposition to the surface of the Earth.

Consideration of these factors of spallation rate, latitude dependence and likely atmospheric circulation, has allowed charts showing expected ³⁶Cl fallout in atoms m⁻² s⁻¹ to be plotted against latitude. The initial latitude dependence of ³⁶Cl fallout was determined by Lal and Peters (1967). This estimate was then altered (Bentley et al., 1986a) to incorporate possible additional ³⁶Cl production in the atmosphere. Andrews and Fontes (1992), on consideration of subsequent research showing a lack of evidence for ³⁶Ar neutron activation reactions (Eq. 7.5) in the atmosphere, reinstated the Lal and Peters model.

Andrews and Fontes (1992) show ³⁶Cl fallout latitude dependency as symmetrical for both hemispheres – ranging from a maximum of 20 atoms $m^{-2} s^{-1}$ at mid latitudes to less than 5 atoms $m^{-2} s^{-1}$ at the poles and equator. The mean ³⁶Cl fallout is computed as 11 atoms $m^{-2} s^{-1}$.

As marine Cl has negligible ³⁶Cl activity (Fontes and Andrews, 1994) and terrestrially derived Cl usually has low ³⁶Cl activity, the ratio of ³⁶Cl ions to stable Cl ions in precipitation is largely determined by the ratio of cosmogenic ³⁶Cl to marine and terrestrial derived Cl (Davie et al., 1989). Near the coast, precipitation contains a large amount of marine Cl; however, because of fallout (dry and wet) with some modification due to prevailing wind and orographic affects (Bentley et al., 1986a), the amount of chloride in rain decreases markedly with distance inland (Hutton, 1976). The fallout of ³⁶Cl is superimposed on this pattern (see Fig. 7.1) with only a latitude affect (Bentley et
al., 1986a). Consequently, at a given latitude, a lower rain ³⁶Cl/Cl is expected at continental margins than in continental interiors. By combining an estimate of latitude dependant ³⁶Cl fallout with records of stable chloride content of rainfall, Bentley et al. (1986a) predicted the long-term average ³⁶Cl/Cl in atmospheric fallout across the USA.

Later studies of ³⁶Cl in rainfall in the USA and Australia (Hainsworth et al, 1994; Keywood, 1995; Keywood et al., 1998) found that the predictions of Lal and Peters (1967) underestimated recorded ³⁶Cl fallout to the Earth. Consequently, Hainsworth et al. (1994) note the initial prediction of ³⁶Cl/Cl fallout variation across the USA by Bentley et al. (1986a) based on Lal and Peters (1967) work is also likely an underestimate of ³⁶Cl fallout in some regions. Keywood (1995) proposed that the estimates of Lal and Peters should be multiplied by a factor 1.47 in the southern hemisphere. Phillips (2000) synthesised data from the USA and Australia (Keywood's 1998 data) to produce a new estimate of ³⁶Cl latitude dependent fallout. In contrast to previous studies, Phillips (2000) estimates ³⁶Cl fallout latitude dependency to range from a maximum of about 50 atoms $m^{-2} s^{-1}$ at mid latitudes down to around 10 atoms m^{-1} 2 s⁻¹ at the poles. Phillips (2000) and Keywood (1995) both estimate the mean fallout to the surface of the earth to be around 30 atoms $m^{-2} s^{-1}$. The reasons for ³⁶Cl fallout being higher than the earlier estimates is unknown, though various explanations such as the existence of unidentified additional ³⁶Cl atmospheric production reactions are speculated upon by Phillips (2000) and Keywood, (1995).

Keywood et al. (1998), on comparing the smaller fallout rates recorded in Australia than those in published USA data sets, note that fallout of ³⁶Cl may well be asymmetric about the two hemispheres, as was the case for ⁹⁰Sr fallout from surface weapons testing. Unfortunately, direct comparison between hemispheres is not possible, as very little overlap of sample collection latitude exists. The possibility that, if bomb derived ³⁶Cl is still retained in the biosphere (e.g. Milton et al., 1997; Cornett et al., 1997), it is more likely to occur in the northern hemisphere, is another feasible reason proposed by Keywood et al. (1998) to explain the observed skewness of ³⁶Cl fallout between hemispheres. Moreover, analyses of rain and dry Cl fallout from composite annual samples collected from 28 meteorological stations around the world (Blinov et al (2000); Scheffel et al., 1999) have also shown ³⁶Cl fallout to be higher than predicted.

The possibility that bomb derived ³⁶Cl is still circulating in the atmosphere is the explanation of elevated ³⁶Cl fallout preferred by these authors.

While these calculations of ³⁶Cl fallout infer a constant rate at a given latitude, actual results of methodical rainfall monitoring undertaken in Australia (Keywood, 1995, 1998) and USA (Kniess et al., 1994; Hainsworth et al., 1994) have shown significant seasonal variations in fallout at any one field site. Furthermore, Kniess et al. (1994) record ³⁶Cl concentrations to vary even during the one rainfall event. Keywood et al. (1998) noted only a weak correlation between ³⁶Cl fallout and rainfall amount in Australia in contrast to stronger correlation noted by Kniess et al (1994) in the wetter less variable rainfall climate of Indiana, USA.

The fallout rate is an important parameter to measure for hydrological studies. For instance, in the absence of nearby rainfall monitoring of ³⁶Cl content (as is usually the case) estimates of ³⁶Cl fallout are combined with best knowledge of total stable chloride fallout (estimate of dry fallout and Cl content of rainfall) to calculate the likely input ³⁶Cl/Cl ratio of any groundwater recharge. Dry fallout, which in general terms can be expected to mix with subsequent rain and contribute to recharge, is estimated to represent < 50% of Cl fallout by Keywood (1995), and < 30% by Hainsworth et al. (1994).

Other local studies of ³⁶Cl in precipitation have been conducted in some other parts of the world. Herut et al. (1992) report that ³⁶Cl determinations from high Cl rainfall events in Israel indicate a significant terrestrial derived Cl source for these events. Magaritz et al. (1990) use a ³⁶Cl/Cl determination from snow on a mountain in Jordan to determine likely precipitation ³⁶Cl content in this region. Finkel et al. (1980) report on ³⁶Cl determinations from Antarctic surface ice.

Phillips (2000), notwithstanding the documented variations in ³⁶Cl fallout observed between hemispheres, seasons and rain events, has used selected northern hemisphere data sets to derive a precipitation ³⁶Cl deposition correlation, normalised for latitude and a mean latitude precipitation rate. This correlation is used to update the work of Bentley et al. (1986a) and construct a new map of ³⁶Cl/Cl ratio of meteoric deposition across the USA. Interestingly, when data typical of Alice Springs, central Australia (chloride concentration of rain 0.6 μ eq/l, mean rain of 250 mm yr⁻¹, latitude of 22⁰ S; Keywood et al., 1997, 1998) are used in Phillips (2000) correlation equation, the meteoric deposition ³⁶Cl/Cl ratio calculated is about 300 by 10⁻¹⁵. This is comparable to the mean ³⁶Cl/Cl ratios calculated from two years of rain and dry fallout observations in this region by Keywood (1995).

The use of annual average assumptions commonly applied for hydrologic assessments in temperate zones should only be employed with caution in arid areas, given the spatial and temporal variation of rainfall in arid areas (e.g. Lerner et al., 1990). This implies a long term assessment (decades) of atmospheric fallout may be necessary to delineate the possible range of ³⁶Cl/Cl ratios in rainfall. For instance Ambroggi (1977) postulates unusually heavy rain events may only be expected once every fifteen years in arid regions. Such events may be crucial sources of replenishment to arid zone aquifers. Thus, unless a considerable period of monitoring is undertaken, the ³⁶Cl/Cl ratios of some significant hydrologic events may go unrecorded. This dilemma is demonstrated by Keywood's (1995) work where the period of sample collection in central Australia (1992 – 1994) was unremarkable in that annual rainfall was within 150 mm of the mean. By contrast during the "big" wet years of 1974 and 2000 annual rainfall was more than five times the mean (www.bom.gov.au). Due to the likelihood of significant recharge for some arid area aquifers during these unusually wet years, it is probable that a more accurate determination of likely groundwater input ³⁶Cl/Cl ratios would have been obtained if assessment of atmospheric ³⁶Cl fallout had occurred during "big" wet years. Such considerations are common to all aspects of hydrological assessment in arid areas (Lerner et al., 1990), hence the need to heed the caution of using average annual assumptions (unfortunately often the only available data) when assessing hydrological data in arid zones.

Natural fallout of ³⁶Cl is still being defined. Recent data collected from both hemispheres seems to indicate that the rate of ³⁶Cl fallout to Earth is higher than predicted by Lal and Peters (1967). The reasons for this disparity between theoretical and recorded fallout rates are yet to be determined.

7.1.3.2. Changes in fallout rate in the past

Bird et al. (1991) note that because of the long half life, changes in the past affecting ³⁶Cl concentrations such as unknown variation in cosmic ray flux or climate could affect the accuracy of any radio chlorine determined age. Earlier ³⁶Cl groundwater studies (e.g. Bentley et al., 1986b) did not consider past variation in ³⁶Cl atmospheric production. More recently Plummer et al. (1997) calibrated ³⁶Cl data from fossil pack rat middens with ¹⁴C data and concluded that atmospheric ³⁶Cl/Cl ratios dropped markedly at the start of the Holocene in semi arid inland U.S.A. This variation is thought to arise from recorded changes in the earth's magnetic field strength. When the earth's magnetic field is stronger, more cosmogenic energy is deflected and ³⁶Cl production is less (i.e. atmospheric ³⁶Cl/Cl ratios are likely to be less during periods of higher magnetic field strength). At the around start of the Holocene, field strength rose to around current levels after a long period (>10 ka) of lower mean field strength (Tric et al., 1991; Guyodo and Valet, 1999). Thus, pre Holocene atmospheric ³⁶Cl/Cl ratios are expected to have been up to 50% higher than present levels at mid latitudes (Plummer et al., 1997).

While Plummer et al. (1997) also postulate that one possible cause of this past atmospheric ³⁶Cl/Cl variation is climate change affecting atmospheric circulation and hence latitude dependence of ³⁶Cl fallout rate (Section 7.1.3.1), both Davis et al. (1998a) and Phillips (2000) consider that past variation in the earth's magnetic field to be the most significant influence on cosmogenic ³⁶Cl production. This conclusion is partly based on the fact that detailed independently dated ¹⁴C measurements show radiocarbon activity (which is not latitude dependent) in the atmosphere 25 ka was almost double that of modern times and that the pattern of ¹⁴C variation with time seems to correlate with reconstructions of past magnetic field variation (Phillips, 2000). However, a detailed study of ¹⁴C concentration calibrated against a ²³⁰Th chronology in a stalagmite by Beck et al. (2001) concluded that the variation in the ¹⁴C record cannot be accounted for by modulations of the earth's magnetic field alone, and suggest that substantial fluctuations in the carbon cycle must also have occurred over the last 45 ka.

Given this evidence for a climate change effect, in addition to the geomagnetic influence on cosmogenic nucleide production, exact reconstruction of past fluctuations

of ³⁶Cl production in the atmosphere may be difficult. Nevertheless Davis et al. (1998a) use the geomagnetic palaeo-intensity data of Tric et al. (1991) to construct a diagram showing relative production rate of atmospheric ³⁶Cl over the last 80 ka. This diagram is reproduced in Figure 7.3 where it can be seen that the major past deviations from current conditions have been periods of increased atmospheric ³⁶Cl production rates. In keeping with this pattern Guyodo and Valet (1999), in a study of the relative intensity of the earths magnetic field, note that many intensity lows have occurred over the last 800 ka. Modern magnetic intensity is then apparently at around the maximum field intensity recorded for this period, as seen in Figure 7.4 which shows Guyodo and Valet's (1999) work. Given the above data, and relying on earth's magnetic field intensity as the major control on ³⁶Cl production, it would seem unlikely that atmospheric ³⁶Cl levels were ever significantly lower than current levels during the Pleistocene. Thus, ³⁶Cl decay dating of groundwater is only likely to underestimate groundwater age if past fluctuations in atmospheric ³⁶Cl/Cl ratios at the time of recharge are unknown. Given these past fluctuations, however, there is a potential discrepancy where two groundwater samples, recharged 10 ka apart, may have markedly different ³⁶Cl/Cl ratios. In recognition of this and other potential complications such as bomb pulse ³⁶Cl contamination Davis et al. (1998a) recommend a suite of complementary analyses when assessing shallow groundwater ³⁶Cl values. These and other considerations are discussed in Section 7.1.4 below.

7.1.3.3. Epigene production

Production of ³⁶Cl by cosmogenic reactions also occurs in the upper few metres of the earths crust and oceans. As argon is not abundant here, the main production reactions (Phillips et al., 1986a) are from spallation of K and Ca by energetic secondary neutrons and thermal neutron activation of ³⁵Cl:

$${}^{35}\text{Cl} + n \rightarrow {}^{36}\text{Cl} + \gamma$$
(7.6)

39
K + n \rightarrow 36 Cl + n + α (7.7)

$${}^{40}Ca + n \rightarrow {}^{36}Cl + \rho + \alpha \tag{7.8}$$

Where n = neutron; ρ = proton; α = alpha particle; γ = gamma radiation, (Clark and Fritz, 1997).



Figure 7-3 Relative production rate of atmospheric ³⁶Cl over the last 80 ka. Reproduced from Davis et al., 1998a with kind permission of Springer Science and Business Media. Original data from Tric et al., 1991.



Figure 7-4 Variation of Earths Magnetic field strength over last 800 ka. Note magnetic field (VADM) has never been higher than at present – hence atmospheric fallout ³⁶Cl is unlikely to have been lower than at present during the last 800 ka. Reprinted by permission from Macmillan Publishers Ltd: Nature, Guyodo and Valet, 1999. © 1999.

Andrews and Fontes (1992) calculated the equilibrium 36 Cl/Cl ratio in the topmost metres of the ocean from epigene production would be 1300 x 10⁻¹⁵, however, with the oceanic circulation of 36 Cl dead chloride from deep old waters continuously mixing with near surface chloride, the 36 Cl/Cl ratio of the oceans is actually < 4.2 x 10⁻¹⁵. Fallout of atmospherically produced 36 Cl to the oceans is similarly diluted.

On the land, ³⁶Cl from cosmic ray processes builds up as soon as a rock is exposed at the surface. Eventually the ³⁶Cl/Cl ratio will attain a secular equilibrium with the production rate and the ³⁶Cl half-life (i.e. the exposed rock accumulates ³⁶Cl until the rate of ³⁶Cl decay, which increases with the number of ³⁶Cl atoms, equals the rate of production, Phillips et al., 1986a). If the chemical composition of the rock and its altitude are known, it is possible to calculate the secular equilibrium ³⁶Cl/Cl ratio (Bentley et al., 1986a). If a rock were suddenly exposed at the surface (eg. basalt) secular equilibrium is not reached until around 1 Ma, thus ³⁶Cl accretion information can be useful as geological dating tool in the 1 to 500 ka range (Bentley et al., 1986a). As most rocks are exposed at the surface by erosion and epigene ³⁶Cl production is rapidly attenuated with depth, most rocks do not reach a secular equilibrium ³⁶Cl/Cl ratio. If the erosion rate is fairly constant, it is possible calculate erosion rates from the amount of cosmic-ray produced ³⁶Cl build-up (Bentley et al., 1986a). Gosse and Phillips (2001) give a detailed account of surface exposure dating methods using multi terrestrial in situ cosmogenic nucleides and Phillips et al. (2001) present improved methods for calculating neutron flux in the epigene.

The potential effect of ³⁶Cl epigene production on groundwater ³⁶Cl/Cl ratios has been considered by a number of workers. Andrews et al. (1991) note the theoretical epigene secular equilibrium ³⁶Cl/Cl ratio at 1m depth for the sandstone at Milk River, Canada to be in the order of 16,000 x 10⁻¹⁵, and speculate that epigene produced ³⁶Cl may explain high ³⁶Cl/Cl found in this aquifer. Andrews and Fontes (1992) note that in arid zones much chloride may exist at shallow depths due to evapotranspiration processes and thus a potential source of mobile epigene ³⁶Cl may complicate hydrological applications of ³⁶Cl in these regions. Fontes and Andrews (1993) calculated a 0.3 m deep brine with Cl = 60,000 mg/l and stream water with 7 mg/l of Cl at 0.3 m to have theoretical epigene secular equilibrium ³⁶Cl/Cl ratios of 105 and 7800 x 10⁻¹⁵ respectively. Interestingly the rates of increase in ³⁶Cl/Cl ratios as a result of epigene production were calculated to be

modest at only 0.24 ka for the brine and 18 ka for the stream water. As epigene ³⁶Cl production takes many thousands of years to build up, and water flux through the upper few metres of subsurface is comparatively usually much more rapid, it seems unlikely that epigene ³⁶Cl production would affect many shallow groundwater systems. Furthermore, bomb peak ³⁶Cl horizons are found in the unsaturated zone in many localities (Section 7.1.3.5) including arid zones (e.g. Liu et al., 1995). This suggests that epigene production, which theoretically produces more ³⁶Cl than bomb peak fallout given sufficient time (Andrews et al., 1991), is not readily discernible at many locations.

Phillips et al. (1986a) note that as epigene ³⁶Cl is created by cosmic ray reactions largely within mineral grains that must be released by weathering before entering groundwater, epigene production is rarely likely to affect groundwater ³⁶Cl/Cl ratios. Furthermore, Phillips (2000) calculates the annual release of in situ cosmogenic produced ³⁶Cl from most rock types, under high and moderate weathering rates, to be extremely small. Carbonate rocks, due to greater susceptibility to weathering, are likely to release the most ³⁶Cl. As cosmogenic ³⁶Cl production increases with altitude, all rock types other than sandstone were shown to begin to have appreciable release when above 2 km in altitude. Based on these calculations Phillips (2000) concludes that except at high mountain elevations ³⁶Cl released by weathering from most rock types will be insignificant in comparison to meteoric input. Davis et al. (1998a) also doubt that epigene production would normally affect to any significant degree the input ratio of ³⁶Cl/Cl in groundwater systems.

Given these considerations, it seems that while epigene production can theoretically produce high ³⁶Cl/Cl ratios in shallow subsurface rocks and water, the rate of build-up and release is so slow when compared to the flux of water that the effects of epigene ³⁶Cl production can usually be disregarded for groundwater studies.

7.1.3.4. Hypogene production

Chlorine-36 is also produced in the deep subsurface through neutron-capture on ³⁵Cl, the neutron flux arising from the decay of natural radioactivity producing a low, and roughly constant level of background chlorine-36 to groundwater systems (Bentley, et al., 1986a).

Fontes and Andrews, (1994) note that α particle emissions by U and Th and their decay products in rocks produces ³⁶Cl by way of the reactions:

$${}^{35}\text{Cl} + n \rightarrow {}^{36}\text{Cl} + \gamma \tag{7.9}$$

$${}^{39}K + n \rightarrow {}^{36}Cl + \alpha \tag{7.10}$$

Of these two reactions the thermal neutron activation of ${}^{35}Cl$ is by far the most significant. Consequently, subsurface production rate of ${}^{36}Cl$ is usually proportional to Cl⁻ concentration. Therefore, hypogene ${}^{36}Cl$ production is usually expressed as the ${}^{36}Cl/Cl$ ratio at secular equilibrium (R_{se}) with subsurface neutron flux, and not as the number of ${}^{36}Cl$ atoms produced per unit volume per unit time (Phillips, 2000).

Hypogene production reaches 97% of equilibrium after 5 half-lives of ³⁶Cl, about 1.5 Ma (Fontes and Andrews, 1994). The equilibrium rate is usually much less than atmospheric rates unless a very high source of radioactivity, such as a uranium ore body, is present.

Generally, at depths below the influence of cosmogenic radiation, the maximum R_{se} expected is about 50 x 10⁻¹⁵ (Phillips, 2000). Bentley et al. (1986a) outline a method of calculation for R_{se} and note that apart from needing to know the rock density, porosity and chloride concentration of groundwater, the calculation of the neutron flux derived from natural radioactivity is difficult. The method presented by Feige et al. (1968) is commonly used to estimate neutron flux (e.g. Phillips, 2000, Andrews and Fontes, 1992, Bentley et al., 1986a). Bentley et al., (1986b) note that hypogene production is a factor to consider if determining absolute groundwater age by way of the equation:

$$t = \frac{-1}{\lambda_{36}} \ln \frac{R - R_{se}}{R_0 - R_{se}}$$
(7.11)

where *t* is time, λ_{36} is the decay constant of ³⁶Cl, *R* is the measured ³⁶Cl/Cl ratio, *R_o* the initial ³⁶Cl/Cl ratio, and *R_{se}* the secular equilibrium due to hypogene production.

From this equation, hypogene production can be seen to be an important factor for determining groundwater ages, particularly in very old groundwater where hypogene production of ³⁶Cl may approach the level of ³⁶Cl activity left after decay of initial atmospheric ³⁶Cl.

Fontes and Andrews (1994) tabulated the hypogene production R_{se} for various rock types, as calculated from published average US rock analyses. R_{se} for common rock types; basalt, granite, sandstone, limestone and clay/shale are calculated as 6.9, 28.3, 2.1, 8.8 and 2.1 by 10⁻¹⁵ respectively. Bentley et al., (1986a) give guideline averages of 30 x 10⁻¹⁵ for granites, 10 x 10⁻¹⁵ for shales and limestones and 5 x 10⁻¹⁵ for sandstones and basalts. These values give an indication of the groundwater ³⁶Cl/Cl ratio in various systems that can be approached before ³⁶Cl groundwater dating is implausible. Addition of chloride along flow lines in an aquifer by leakage from aquitards, subsurface brines or mixing with another groundwater of unknown Cl⁻ concentration can complicate interpretation of hypogene content and hence also groundwater dating. Such situations seem common as outlined in Chapter 8 below.

Conversely, in high U and Th host rocks such as uranium ores or some granites the R_{se} can be very high. For example Fabryka-Martin et al. (1987) report very high ³⁶Cl/Cl ratios in some groundwaters (up to > 1700 by 10⁻¹⁵) around the Alligator River Uranium Ore Province in the Northern Territory (NT) of Australia. Songsheng et al. (1994) also report on high ³⁶Cl/Cl ratios in groundwater (up to 1200 by 10⁻¹⁵) around a uranium ore body in China.

High levels of ³⁶Cl production associated with granites are reported from Sweden (Andrews et al., 1986; Mahara et al., 2001) and Canada (Gascoyne et al., 1994). Lehman and Loosli (1990) report on hypogene production from various lithology types in northern Switzerland. In these instances hypogene production is greater than atmospheric sources of ³⁶Cl and groundwater dating by way of ³⁶Cl decay is not possible. Fontes and Andrews (1994) and Andrews and Fontes (1992) note, however, that dating may sometimes be possible by determining the amount of ³⁶Cl ingrowth before secular equilibrium is obtained. This situation, with no Cl⁻ addition, is illustrated in Figure 7.1.

The use of discordance between calculated subsurface equilibrium ratios and measured values in samples to calculate water residence time (ingrowth method of Fontes and Andrews, 1994) is the usual method of dating geothermal systems (Phillips, 2000). For example, Aquilina et al. (2002), in a study on the origin of thermal springs in southern France, make a rough (order of magnitude) estimate of geothermal groundwater age (200 ka) by assuming that the ingrowth of hypogene ³⁶Cl (assumed secular equilibrium of 11 x 10⁻¹⁵) has raised the deep aquifer ³⁶Cl/Cl ratio above the input sea water ratio (i.e. $\approx 0 \times 10^{-15}$). Fehn et al. (1994) make preliminary age estimates of very deep (> 1.8 km) groundwater in Germany and California, USA by considering the amount of likely subsurface production of ³⁶Cl in each region.

Lippmann et al. (2003) consider the high ³⁶Cl concentrations of groundwater sampled in ultra deep mines (>0.7 km) in South Africa to indicate subsurface ³⁶Cl production is at secular equilibrium and thus groundwater is assumed to be at least 1.5 Ma (5 times λ ³⁶Cl). This conclusion is confirmed by complementary noble gas data that show groundwater to be very old (range 1 – 100 Ma).

Phillips (2000) notes ³⁶Cl is also useful to distinguish the source of the Cl in geothermal systems. For example Balderer and Synal (1996), on the basis of ³⁶Cl results and complementary geological data, speculate that at the Tulza geothermal area of Turkey tectonic processes are squeezing brine from Agean Basin sediments through crystalline basement faults, where the brines are heated before rising and mixing with low salinity shallow groundwater in overlying volcanics. A similarly complex source of chlorides is derived from ³⁶Cl data for the hydrothermal fluids in the Valles caldera in New Mexico, USA by Rao et al. (1996). They propose that brines are derived from meteoric waters that travelled through Precambrian basement and Palaeozoic strata leaching chloride before reaching the intra-caldera volcanic suite via convective up-welling. Some low chloride, high ³⁶Cl/Cl groundwaters of recent meteoric origin are also distinguishable at this caldera.

Recent work by Chimiel et al. (2003) has shown rocks such as carbonaceous shales, coals, phosphorites and granites/pegmatites can concentrate high levels of uranium and may consequently have significant in situ ³⁶Cl production when compared to atmospheric sources of ³⁶Cl.

Consequently, on the basis of the above discussion, it appears that unless measured 36 Cl/Cl approaches expected R_{se} values and/or exceptionally high rock U and Th values are suspected, 36 Cl groundwater dating should be possible in systems with consistent chloride concentrations. If R_{se} is unknown then an approximation of age may be made, assuming R_{se} to be small or negligible.

7.1.3.5. Anthropogenic sources

7.1.3.5.1. Bomb Peak

The advent of nuclear technology has resulted in an anthropogenic component of radioactive nuclides now being present in the environment. Surface testing of thermonuclear weapons in the 1950s and 60s released many radioactive nuclides into the environment (Bentley et al., 1982). The production of ³⁶Cl occurred during these events because ³⁵Cl has a large thermal neutron capture cross section and free neutrons are easily absorbed (see Eq. 7.6). Consequently, the nuclear testing on barges in the sea (i.e. irradiation of seawater ³⁵Cl) near Bikini and Eniwetok atolls in the Pacific between 1952 and 1958 resulted in a large amount of ³⁶Cl being released into the atmosphere (Phillips, 2000).

The actual amounts of ³⁶Cl released from these tests have been calculated by Synal et al. (1990). Fallout of ³⁶Cl from these tests was recorded in precipitation by Schaeffer et al. (1960). Measurement of ³⁶Cl in shallow ice cores from Dye-3 in Greenland has provided a record of this "bomb peak" or pulse (Elmore et al., 1982; Suter et al., 1987; Synal et al., 1990). The results from three separate cores near Dye-3 show a peak in the late 1950s of 500 to 1000 times the expected cosmogenic ³⁶Cl production. (Synal et al., 1990). These data are interpreted to show that ³⁶Cl fallout had returned to natural levels by around 1985. In keeping with this interpretation, Vogt et al. (1994) concluded no bomb ³⁶Cl was present in precipitation samples they analysed from Indiana in the USA.

The distribution of the bomb peak ³⁶Cl fallout from these tests near the equator was assumed by Bentley et al. (1986a) to be equally dispersed about the hemispheres. This

interpretation was backed up by cited data (Lockhart, 1959), which shows ¹⁸⁵W fallout from some of these tests to be equally distributed in northern and southern hemispheres.

Conversely, Keywood et al. (1998) cite 90 Sr fallout data (Machta and List, 1959) that shows fallout from these tests was four times greater in the northern hemisphere than the southern. This asymmetry is also noted to apply to aerosols and ozone and is attributed to the different topography of the northern hemisphere land masses causing a much more turbulent troposphere, which then results in greater troposphere – stratosphere transfer in the northern hemisphere (Keywood et al., 1998).

The worldwide peak of ³⁶Cl injected into the atmosphere from the thermonuclear weapons testing programs of the 1950s and 1960s is of potential benefit to hydrological studies as it provides a readily detectable tracer (Bentley et al., 1986a). This is particularly so as bomb peak tritium (half life of only 12.4 years) has largely decayed to modern rain levels in the subsurface, which has virtually rendered bomb tritium an ineffective tracer in the southern and most of the northern hemisphere (Hendrickx and Walker, 1997). Phillips (2000) notes that the advent of chlorofluorocarbon (CFC) and tritium/helium 3 techniques to study modern water has probably resulted in bomb ³⁶Cl being under utilised as a tracer. Nevertheless, Bomb Peak ³⁶Cl has been used as a tracer for various hydrological applications in many parts of the world.

Inferences on recharge have been attempted, using bomb peak ³⁶Cl. For example; Liu et al., (1995) in shallow vadose zone in the desert of Arizona, USA; Cook et al. (1994) in the vadose zone of the Murray Basin, South Australia; Scanlon et al. (1990) in Texas, USA; and Norris et al. (1987) in Nevada, USA. Phillips et al. (2000) note that the use of bomb ³⁶Cl in the vadose zone to study recharge has been largely inconclusive in many arid zones as slow recharge rates have resulted in the bomb ³⁶Cl being concentrated near the root zone.

Another application of bomb ³⁶Cl is as an indicator for the presence of a component of modern (post nuclear testing) recharge in groundwater. Samples that show ³⁶Cl/Cl ratios above expected atmospheric levels can indicate the presence of bomb³⁶Cl. For example Campbell et al. (2003) present some data from the Yucca Mountain proposed nuclear waste repository in Nevada. High ³⁶Cl levels at 200m below ground surface indicate

water containing bomb ³⁶Cl has travelled via preferred pathways to this depth in the vadose zone. Murad and Krishnamurthy (2004) interpret the ³⁶Cl data from eastern United Arab Emirates aquifers to indicate a component of bomb ³⁶Cl, and hence post 1950 recharge, has reached these aquifers. Although no comparison to expected atmospheric ³⁶Cl/Cl ratios is made, the evidence of anthropogenic contamination from other chemical data confirm the presence of bomb peak ³⁶Cl. A study in Zimbabwe by Larsen et al. (2002) attempted to look at recharge rates by tracing bomb peak ³⁶Cl in the Karoo Formation but preliminary results proved inconclusive.

Davis et al. (1998a) discuss criteria to identify groundwater that may have a component of bomb ³⁶Cl so that such samples can be disregarded in ³⁶Cl decay age dating studies.

7.1.3.5.2. Recirculation of bomb ³⁶Cl

While surface nuclear testing has ceased and atmospheric ³⁶Cl levels are now back to pre-testing levels (Suter et al., 1987) some recent studies (Blinov et al., 2000; Corcho Alvarado et al., 2005; Cornett et al., 1997; Herczeg et al., 1997; Milton et al., 2003; Scheffel et al., 1999) note that some bomb ³⁶Cl may still be circulating in the environment with the potential to complicate hydrological studies using ³⁶Cl. In North America due the fact that chloride is known to be an essential micronutrient for higher plants (e.g. White and Broadley, 2001), this effect is thought to be related to the entrapment, and subsequent slow release, of bomb ³⁶Cl in the vegetation rich biomass associated with conifer forests (Milton et al., 2003; Cornett et al., 1997.) Production by plants is now recognised as a significant source of methyl chloride in the atmosphere (Rhew et al., 2000, 2003) so, depending on the halide retention time within plants and availability of bomb ³⁶Cl, some bomb ³⁶Cl may well be currently released to the atmosphere. The magnitude of this effect on atmospheric ³⁶Cl/Cl ratios is difficult to estimate. The evidence from the Dye-3 ice core analyses discussed above suggests this effect may be only subtle on a global scale. This point is reinforced by Yokouchi et al. (2000) who suggest only 12% of estimated global CH₃Cl flux cannot be accounted for by oceanic fluxes.

In south-eastern Australia Herczeg et al. (1997) noted that groundwater ³⁶Cl/Cl values are greater than predicted atmospheric ³⁶Cl/Cl ratios and speculate that bomb ³⁶Cl may be recirculating back to the environment in the vicinity of sinkholes and thereby affecting groundwater ³⁶Cl/Cl ratios. However, as the predicted atmospheric ratios ³⁶Cl/Cl ratios are based on the Lal and Peters (1967) ³⁶Cl fallout rate which has since been found to underestimate natural fallout (Keywood, 1998; Phillips, 2000), it may be that no evidence of bomb ³⁶Cl exists. Indeed Herczeg et al. (1997) note an alternative explanation is that a component of continental-derived Cl in rainfall from dust containing some bomb ³⁶Cl, as discussed by Simpson and Herczeg (1994), results in groundwater ³⁶Cl/Cl ratios being above expected atmospheric levels. In the Yass Valley of New South Wales Turner et al. (1992) also explain recorded ³⁶Cl/Cl ratios from shallow groundwater slightly above the expected Lal and Peters (1967) predictions as being the result of bomb ³⁶Cl. Again it is possible that a reappraisal of atmospheric ³⁶Cl/Cl ratios in this region may diminish the perceived effect of bomb ³⁶Cl.

Scheffel et al. (1999) and Blinov et al. (2000) suggest recycling of bomb ³⁶Cl may be occurring. This is on the basis of ³⁶Cl analyses of composite annual rainfall from various sites around the world showing elevated ³⁶Cl/Cl compared with predictions of an atmospheric transport model they had developed. However, it should be noted that samples from various years between 1989 and 1996 were used and only two sites from the southern hemisphere in New Zealand were sampled. Thus, other factors such as rainfall intensity and alternative atmospheric circulation models as proposed for the southern hemisphere by Keywood et al. (1998) may also influence these results. By contrast Vogt et al. (1994) from results at Indiana, USA concluded no bomb ³⁶Cl was circulating in the atmosphere. A recent study of groundwater ³⁶Cl/Cl ratios from 38 sites across the United States and previously published data by Moysey et al. (2003) attributes observed differences in ³⁶Cl/Cl to natural variation expected in the atmosphere (e.g. variation in Cl fallout in rainfall) across a continental land mass. The possibility of bomb ³⁶Cl still circulating in the environment was not invoked to explain ³⁶Cl/Cl variation across the continent. Thus it may be that recycling of bomb ³⁶Cl in the environment is only a localised phenomenon.

Hence, given there is still some uncertainty that recirculating bomb ³⁶Cl is ubiquitous in the environment, it may be that further studies of atmospheric ³⁶Cl fallout need to be undertaken to clarify this matter.

7.1.3.5.3. Localised anthropogenic ³⁶Cl sources

Apart from the worldwide distribution of bomb ³⁶Cl resulting from nuclear testing in the Pacific, other anthropogenic activities produce ³⁶Cl in the environment which may be used as a tracer at a local scale. For example, Milton et al. (1994) and Beasley et al. (1993) trace reactor-produced ³⁶Cl in the vicinity of nuclear power plants at Ontario, Canada and Idaho, USA respectively.

Local nuclear testing may also provide useful radioisotope tracers. Cresswell et al. (2002) speculate that British onshore nuclear weapons testing at Maralinga in central South Australia between 1952 and 1963 has resulted in a ³⁶Cl bomb peak in some shallow aquifers further north in the Anangu Pitjantjatjarra Lands of South Australia. British weapons' testing is considered to be the source because this bomb peak is recorded only in a narrow band to the north east of the test site. Bird et al. (1991) note ³⁶Cl/Cl ratios at up to 10⁻¹¹ at the surface near Maralinga, which is the same as recorded (non-hazardous) contamination around the Lucas Heights research reactor in southern Sydney. Although onshore tests are not thought to produce significant ³⁶Cl (Phillips, 2000) due to a lack of nearby ³⁵Cl, this pattern might indeed represent the remnants of localised windblown radioactive contamination following the weapons testing.

7.1.4. Inputs

To measure the ³⁶Cl decay in groundwater it is necessary to have an idea of the initial or input ³⁶Cl/Cl ratio to use the decay equation (Eq. 7.2). When local ³⁶Cl fallout data are not available, as is often the case, other strategies are needed to estimate input ³⁶Cl/Cl ratio. Davis et al. (1998a) give a review of the considerations necessary to best predict this ratio. One method is to use the theoretical fallout of ³⁶Cl and the expected amount of chloride in precipitation for a given area to calculate atmospheric ³⁶Cl/Cl ratios. Another more empirical approach is to measure ³⁶Cl/Cl ratios in groundwater at

locations perceived to lie in recharge areas and assume these ratios represent preanthropogenic atmospheric input ratios. Consideration of changes in past fallout ratios due to variations (discussed Section 7.1.3.2) in the earth's magnetic field are also recommended by Davis et al. (1998a).

Given the problems in accurately predicting atmospheric ³⁶Cl fallout discussed above (Section 7.1.3.1) and inherent uncertainties associated with identifying recharge zones Davis et al. (1998a) note these methods may have serious weaknesses.

Previous Australian studies or discussions have relied on measured ³⁶Cl/Cl ratios of shallow groundwater in assumed recharge zones to provide an estimate of input ratio (e.g. Kellett et al., 1990; Torgersen et al., 1991; Mazor, 1992; Love et al., 2000; Cresswell et al., 1999a). While such an approach has, on first impressions, the potential to provide a true input ratio as natural variations in atmospheric ³⁶Cl/Cl ratios may all be integrated into a representative sample, care must be taken because many factors may potentially affect the ³⁶Cl/Cl of shallow groundwater.

Davis (1998a) propose that when no actual atmospheric ³⁶Cl/Cl measurements are available, shallow groundwaters need to be screened to assess that no bomb ³⁶Cl or dead chloride additions have affected the ³⁶Cl/Cl ratio. Screening of samples usually involves measuring for other radio isotopes (tritium or ¹⁴C) to ensure that no bomb ³⁶Cl contamination is indicated and that the sample is in fact young enough to provide an estimate of modern atmospheric ³⁶Cl input.

While unravelling the provenance of all chloride content in groundwater to check for addition of dead chloride or other potential complications is acknowledged to be difficult, much weight is given to the insights offered from studying the Cl/Br ratios of groundwater. Like chloride, bromide is also considered to act conservatively in groundwater (Hem, 1985) and therefore the Cl/Br ratio of recharge water should remain constant through an aquifer unless addition or a loss of halides occurs. In simple terms, the use of the Cl/Br ratio is based on the premise that halides in rainfall or recharge water are of oceanic origin. Given the high solubility of halides, losses of chloride or bromide via precipitation of salts from solution are not expected in any but hyper-saline groundwaters. Hence, as the Cl/Br mass ratio of seawater is around 290 (Hem, 1985),

any measured groundwater Cl/Br ratio in excess of 290 can indicate chloride content may have a component derived from other sources. If other sources of chloride are indicated use of the decay equations (Eq.7.2) for ³⁶Cl decay dating becomes problematical. Given this utility, the use of Cl/Br ratios has been recommended as part of routine assessment of ³⁶Cl groundwater data sets by Mazor (1992) and Fontes and Andrews (1994).

There is an exception to this general trend that should be considered for studies of hyper saline groundwater. For instance Penny et al. (2003) note groundwaters derived from remnant evaporated brines beyond halite saturation can exhibit Cl/Br ratios less than seawater. Davis et al. (1998b) cite several studies of hyper saline (TDS > 100,000 mg/L) groundwaters which have very low Cl/Br ratios apparently as a result of this phenomenon. Davis et al. (1998b) give a review of the use of Cl/Br mass ratios in groundwater studies and recommend bromide concentrations to be determined as a matter of course in groundwater investigations.

The use of Cl/Br (or Br/Cl) ratios as a tool to trace subsurface chloride origins has been widely reported in many studies. For example: Edmunds (1996) uses Br/Cl ratios to examine origins of salinity in several British aquifers; Fabryka- Martin et al. (1991) use halide ratios to study the origins of subsurface chlorides in the Milk River aquifer in Canada; Herczeg et al. (1991) use Cl/Br ratios to study the behaviour of chloride in the Great Artesian Basin; Andrews et al. (1994) and Edmunds and Smedley (2000) use Br/Cl ratios to examine the Midlands aquifer of England; Cartwright et al. (2004) make much use of Cl/Br ratios when tracing the origins of salinity in the Murray Basin of south eastern Australia.

In recognition of the usefulness of shallow groundwater analyses to determine ³⁶Cl/Cl input ratios, a large project to assess ³⁶Cl distribution across the USA has focussed on shallow groundwater ³⁶Cl/Cl ratios. Results from this project are reported in a series of publications (Davis et al., 1998a, 2000, 2001, 2003; Moysey et al., 2003). At the culmination of this study Davis et al. (2003) set criteria to evaluate groundwater samples for anthropogenic affects or chloride contamination (e.g addition of subsurface brine) to ensure useful ³⁶Cl/Cl data are determined. These criteria are: Chloride concentration < 10 mg/L, Tritium < 2 units, ¹⁴C < 86% (pmc), ³⁶Cl/Cl < 2000 x 10⁻¹⁵,

Cl/Br mass ratio < 260, NO₃ < 5 mg/L. In other parts of the world, some of these criteria may not be suitable as indicators of anthropogenic contamination. For example in central Australia, where virtually no agriculture or pollution is expected in this sparsely populated region, high NO₃ levels are endemic with natural levels often in excess of 50 mg/L (Murray and Seibert, 1962). Various natural biological factors are considered responsible for generating these high nitrates in central Australia (Barnes et al., 1992). Groundwater salinity, hence also chlorinity, is also high in central Australia. Tickell (1997) maps groundwater salinity to be at least above 500 mg/L TDS in this region. Another study in this region, which included data from over 800 bores, recorded only one analysis where chloride was less than 10 mg/L (Wischusen, 1998).

Consequently, it seems that radioactive evidence of anthropogenic contamination (Tritium, ¹⁴C, high ³⁶Cl/Cl ratio) and deviation from expected atmospheric Cl/Br ratios to highlight subsurface brine contamination etc. (Davis et al. (1998b, 2001), are the only criteria that may be applicable to determining integrity of shallow groundwater ³⁶Cl data in central Australia (and most likely other sparsely populated arid regions).

7.1.5. Conclusions

Natural fallout of ³⁶Cl to the earth is yet to be categorically determined. The theoretical fallout calculated by Lal and Peters (1967) seems to underestimate fallout when compared to the available, but limited, empirical data sets. The amount of ³⁶Cl fallout in precipitation is shown to vary with seasons at any particular field site, so an average annual assumption is used to estimate ³⁶Cl fallout. More routine collection of ³⁶Cl in rainfall across continents is needed to better estimate the spatial distribution of atmospheric ³⁶Cl/Cl ratios.

Studies of the palaeo intensity of the earth's magnetic field over the last 800 ka indicate intensity has never been significantly higher than at present. Consequently, atmospheric ³⁶Cl/Cl ratios are unlikely to have ever been lower over the same period.

The effects of epigene ³⁶Cl production can usually be ignored on groundwater studies as the rate of epigene ³⁶Cl production is very slow compared to the flux of water in the subsurface.

Hypogene production is usually only small (<10 by 10^{-15}) when compared to atmospheric ³⁶Cl/Cl ratios. Only when exceptionally high rock U and Th contents are suspected, or measured groundwater ratios approach hypogene production levels, is it an important consideration in ³⁶Cl decay dating. Geothermal systems can sometimes be dated by comparing the amount of ³⁶Cl ingrowth with a known secular equilibrium ratio. More often age is difficult to quantify. Use of ³⁶Cl has proved useful as a tracer to help identify origins of chloride in some geothermal systems.

Bomb peak ³⁶Cl is a useful hydrological tracer. There is some evidence that bomb produced ³⁶Cl is still being released into the environment via methyl chloride production from plants. Based on known global methyl halide budgets the significance of this effect seems slight. It may be that in some heavily vegetated areas the effect is more pronounced.

Determination of an initial input ³⁶Cl/Cl ratio when atmospheric data are absent is necessary to determine ³⁶Cl decay age. Evidence of anthropogenic radioactive contamination and deviation from expected atmospheric Cl/Br ratios are the most appropriate criteria against which to test shallow water samples for ³⁶Cl integrity in sparsely populated arid regions.

7.2. Results

7.2.1. Introduction

Chlorine-36 (³⁶Cl) is a radioactive isotope with a half-life of 301 000 years. As chloride is hydrophilic, study of this isotope has the potential to be used as a groundwater dating tool for waters some millions of years old. Thus, this isotope also has the potential to determine whether measured ¹⁴C activity is truly representative of groundwater residence time or merely the signature of the mixing of older water with modern carbon-

14 bearing water. Bethke and Johnson (2002) illustrate how only slight contamination of a very old water by a modern water can result in an apparent isotope age very much younger than the actual mix ratio of vintages would suggest.

Once all the factors that can affect ³⁶Cl decay dating are accounted for (see review of ³⁶Cl in the environment in Section 7.1 above), ³⁶Cl/Cl ratios can be used to estimate age of groundwater if four assumptions are made (e.g. Andrews and Fontes, 1992):

- the only sink for ³⁶Cl in the aquifer is radioactive decay;
- the only source for additional ³⁶Cl is normal deep subsurface production;
- additional ³⁶Cl sources can be identified and quantified;
- production rate for 36 Cl is the same now as at the time of recharge.

The groundwater may then be dated using the standard decay equation (Eq. 7.2) where A_t is the ³⁶Cl/Cl ratio for the groundwater sample, A_o is the ³⁶Cl/Cl ratio for the recharge (or input function), and 1/ λ is the mean life of ³⁶Cl (434,251 years).

7.2.2. ³⁶Cl analyses

The results of ³⁶Cl analyses of samples taken from Palm Valley and at several other sites along strike in the vast Hermannsburg Sandstone outcrop are shown in Table 7.1.

Plots of these data using one of the standard graphical presentations outlined by Bird et al. (1991) are shown in Figures 7.5 and 7.6. When the data are plotted as shown on Figure 7.5 it can be seen most data points fall along the ³⁶Cl decay line centred on a chloride content of around 100 mg/L. One point (shot hole 1/608) plots along the evaporation line, another, RN 14952, seems to plot on a dilution line, when these results are compared to the processes chart included in Figure 7.5. Figure 7.6 shows a distinct decay trend when data are compared to the included processes chart. It seems that evaporation effects could account for all deviations off the decay line seen in Figure 7.6.

Valley study
Palm
data,
Isotope
7.1
Table 7

SITE	LOCATION	δ ¹⁸ O ‰ SMOW	δ ² H ‰ SMOW	¹⁴ C % MC	δ ¹³ C ‰ PDB	CI mg/L	³⁶ Cl atoms/L by 10 ⁶	³⁶ CI/CI by 10 ⁻¹⁵	ERROR ³⁶ CI/CI analysis	³⁶ Cl age, years	Depth of sample (m)	Water level (m)
RN 4508	Palm Valley	-7.51	-49	3.1	-10.8	118	290	130	ω	397900	-70	-1.5
RN 10074	Areyonga					76	220	175	10	268818	-100	-3.0
RN 12024	Palm Valley	-7.7	-51.3	22.5	-9.4	132	545	245	10	122705	-18	-8.4
RN 14165	Hermannsburg					114	350	190	10	233106	-161	-0.5
RN 14616	Palm Valley	-7.16	-45.8	4.2	<u>6</u> -	81	310	220	15	169444	-95	-0.8
RN 14952	Intjarrtnama					83	390	290	15	49480	-180	-41.0
PV04/200	Palm Valley	-7.25	-45.8			100	220	160	10	307733	ထု	-6.0
PV01/608	Palm Valley	-6.72	-46.7			212	520	205	10	200109	പ്	-3.0
Road Sp.	Palm Valley	-6.89	-44	42.4	-11.8	96	260	190	15	233106	-0.5	surface
Lt. Palm Sp.	Palm Valley	-3.75	-28.9	87.2	-9.2						-0.5	surface
Oasis Sp.	Palm Valley	-3.59	0.6									surface
West Sp.	Palm Valley	-4.29	-36.2	79	-11	114	360	265	15	88629	<u>-</u>	surface



Figure 7-5 Plot of ³⁶Cl/Cl versus chloride for Hermannsburg Sandstone process trends are shown by arrows. If the concentration of chloride in rainfall by evaporation is assumed to account for groundwater chloride concentration of around 100 mg/L, then the range of data points is most (from the process chart) readily explained as radioactive decay of Chlorine – 36. The outliers to this trend 14952 and 1/608 can be explained by differing amounts of evaporation affect. The Data for the Ti Tree rainfall is after Keywood, (1995).



Figure 7-6 Plot of ³⁶Cl/Cl versus ³⁶Cl atoms for Hermannsburg Sandstone sites sampled for this study, process trends are shown by arrows. This diagram also shows a ³⁶Cl decay trend like Fig. 7.3. Data points not on the decay trend shown by the main cluster of data points can again be explained by different amounts of evaporation affect. These data support the conclusion that radioactive decay is the main reason for ³⁶Cl/Cl variations within the Hermannsburg Sandstone and that consequently groundwater age varies by hundreds of thousands of years between some sample points.

It should be noted that while the magnitude of likely error for these ³⁶Cl/Cl ratios has been computed by the ANU and included in Table 7.1, these errors are too small to be included as error bars on these plots.

It should also be noted that no aquifer ever has exactly consistent chloride content when sampled along flow paths. NT Government records show that even samples taken from the same bore at different intervals will have slightly differing chloride contents. As discussed above (Section 7.1) this variation can arise from slight differences in the amount of evaporation during recharge or even addition of subsurface chlorides. It is because of this that ³⁶Cl results are initially evaluated by way of process charts, as shown in Figure 7.2, so that the likely cause of variation of chloride concentration between samples can be seen. Given this expected variation, statistical appraisal of ³⁶Cl results by way of r^2 values of lines of best fit etc. on say ${}^{36}Cl/Cl$ v Cl plots is never applied. Indeed there is no instance of statistical analyses of trend lines for ³⁶Cl/Cl or ³⁶Cl atoms versus chloride plots reported in any of the literature cited in this thesis. Thus, while the result plots shown in Figure 7.5 and 7.6 are useful to look for relevant trends, like any graphical technique, some caution should be exercised when interpreting results. For instance manipulation of axis scales can change the apparent significance of any process trend observed. For example, if larger scale axes had been used in Figure 7.5 the decay trend observed might have been more pronounced. Conversely, if smaller scale axes had been used no significant trend may have been discernable as data points would plot as a scatter filling the graph. To illustrate that the decay trend seen here is real and not an artefact of manipulation of plot axes, the ³⁶Cl decay results are compared graphically to many other ³⁶Cl groundwater results from other studies in Chapter 8 below.

This clear decay trend, an unusual luxury in ³⁶Cl groundwater studies (see Chapter 8), allows these data to be interpreted without contemplating un-measurable complex sources and sinks of chloride within the aquifer. For example the limited range of chloride in the samples here (44 - 212 mg/L, Table 7.1) is reasonably explained as variation in evaporation during recharge and discharge or in the dissolving of some residual sub surface chloride left over from evapotranspiration processes en route to the aquifer, rather than an intra aquifer source or sink of chloride effect.

The variation between the highest and lowest ³⁶Cl/Cl ratio is about one half life for all samples from the Hermannsburg Sandstone. At Palm Valley a significant variation between sampled ³⁶Cl/Cl ratios is also apparent when bore RN 4508 is compared to either RN 12024 or West Spring (Table 7.1). Thus, if a shallow groundwater with some radiocarbon activity (e.g. West Spring) was taken as a likely input ratio, (a strategy adopted for some Australian ³⁶Cl studies e.g. Kellett et al., 1990; Torgersen et al., 1991; Mazor, 1992), groundwater age would vary by about one ³⁶Cl half life (300 ka) between West Spring and RN 4508.

7.2.3. Factors affecting the use of 36 Cl as a chronometer.

7.2.3.1. Introduction

In Section 7.1 a review of factors that may affect the use of ³⁶Cl decay to date groundwater were presented. These factors are considered below in relation to the results from Palm Valley.

7.2.3.2. Input ratio

In Section 7.1.31 a review of studies examining fallout of ³⁶Cl to the Earth's surface was presented. The validity of the observed ³⁶Cl data at Palm Valley is complemented by recent work elsewhere in central Australia. Keywood (1995) measured atmospheric ³⁶Cl/Cl ratio in precipitation at Ti Tree (250 km north east of Palm Valley) at 325 by 10⁻¹⁵. This value compares to the least decayed data point shown on Figure 7.6 (290 x 10⁻¹⁵ at bore RN14952 Injarrtnarma). Furthermore, this value is approximately the same as that calculated using the formula of Phillips (2000) discussed in Section 7.1.3.1.

The variation between the highest and lowest ³⁶Cl/Cl ratio is about one half life for all samples from the Hermannsburg Sandstone. At Palm Valley a significant variation between sampled ³⁶Cl/Cl ratios is also apparent when bore RN 4508 is compared to either RN 12024 or West Spring (Table 7.1). Thus if a shallow groundwater with some radiocarbon activity (e.g. West Spring) was taken as a likely input ratio, (a strategy

adopted for some Australian ³⁶Cl studies e.g. Kellett et al., 1990; Torgersen et al., 1991; Mazor, 1992), groundwater age would vary by about one ³⁶Cl half life (300 ka) between West Spring and RN 4508.

Therefore assigning a 36 Cl/Cl input ratio of around 300 by 10^{-15} for groundwater studies in this region has some credence. If the assumptions pertaining to equation (2.2) are assumed valid, and an input ratio of 325 by 10^{-15} is used, groundwater age is calculated to vary by over 300 000 years in the Hermannsburg Sandstone (Table 7.1).

Given that ³⁶Cl/Cl ratios are noted to vary with season (Keywood, 1995) and rainfall amount (Kniess et al., 1994) an exact input ratio may be impossible to estimate. Thus while an indication of relative ages remains feasible, predicting actual age of groundwater using ³⁶Cl decay maybe somewhat problematical. An error function of say at least around +/- 20 % may be the best way to view estimates of absolute groundwater age at Palm Valley. This implies correlation of recharge to past Quaternary pluvial events may also only be an approximation here.

7.2.3.3. Past changes of ³⁶Cl fallout rate

The potential for ³⁶Cl fallout to have been different in the past has been discussed in Section 7.1.3.2. While potential complications are recognised from this phenomenon it is highly unlikely that ³⁶Cl fallout has ever been less than at present. Thus, the relative difference in ³⁶Cl/Cl ratios, if taken to be a result of radioactive decay, indicate around 300 ka distinction between some groundwater sampled at Palm Valley.

7.2.3.4. Epigene Production

Epigene ³⁶Cl production in the upper few metres of the Earths crust is reviewed in Section 7.1.3.3. This review concluded that due to the relatively fast movement of groundwater compared to the rate of epigene ³⁶Cl production that in most instances epigene ³⁶Cl production can be ignored (e.g. Phillips, 2000). There is no evidence that suggests this general conclusion should not also be valid at Palm Valley.

7.2.3.5. Hypogene production.

Hypogene production of ³⁶Cl is reviewed in Section 7.1.3.4. In general, hypogene ³⁶Cl production rates are low. For instance, the maximum ³⁶Cl/Cl secular equilibrium for sedimentary rocks is around 10 x 10^{-15} . Furthermore, it is only when groundwater ³⁶Cl/Cl ratios approach hypogene production rates that its affect on ³⁶Cl decay age determination is significant.

As the lowest ³⁶Cl/Cl ratio at Palm Valley is 130 by 10^{-15} it is unlikely that, if typical hypogene production rates apply (i.e. sandstone R_{se} max 10 x 10^{-15}), hypogene production is a significant factor here. The only way hypogene production is likely to affect results here is if the Hermannsburg Sandstone has an unusually high uranium content.

To investigate this possibility samples from bores at Palm Valley were examined for trace element content by Inductively Coupled Plasma Atomic Emission Spectra (ICPAES) techniques. The results of these analyses are presented in Table 7.2 where it can be seen that as U content does not exceed 9 μ g/L, which is within the normal range expected for groundwater (Hem, 1985).

No whole rock chemistry analyses of the Hermannsburg Sandstone were conducted for this study. Simpson et al. (1991) however, reports XRF analyses from 20 sites in the Hermannsburg Sandstone from near the top of the Palm Valley anticline only 3 km north of the main stand of Palms. While Simpson et al. (1991) do not report a uranium measurement, thorium levels are reported. The Th concentrations reported are never more than 14 parts per million (ppm) and usually well less than 10 ppm.

ELEMENT	4508	12024 14616		
		All results in µg/L		
Ag	0.01	<0.006	0.01	
A	273.39	4.71	45.41	
As	2.70	2.85	0.35	
Ba	87.30	77.15	45.47	
Be	0.10	0.01	0.01	
Bi	<0.02	<0.02	<0.02	
Cd	8.48	2.69	2.73	
Ce	0.48	0.01	0.23	
Co	0.27	0.11	0.16	
Cr	6.36	0.73	4.62	
Cs	0.39	0.09	0.09	
Cu	3.77	2.06	32.43	
Dv	0.08	0.01	0.03	
Fr	0.04	0.01	0.01	
Eu	0.04	0.02	0.03	
Ga	1 99	1 61	1 02	
Gd	0.11	0.00	0.04	
Ge	0.17	0.00	0.04	
Ho	0.17	0.04	0.03	
	0.02	0.00	0.01	
	22.91		12 10	
	0.01	0.00	0.00	
Mn	25.44		46.09	
Mo	25.44	<0.07	40.90	
Nb	0.43	0.00	0.10	
	<0.03	<0.03	<0.03	
	0.34	0.02	0.21	
	23.64	2.47	2.70	
PD Dr	26.03	0.64	10.90	
Pr Dt	0.07	0.00	0.05	
	<0.009	<0.009	<0.009	
RD	20.30	9.64	5.21	
	0.05	0.03	<0.02	
	<0.11	0.07	<0.11	
Se Sm	3.32	0.30	3.08	
	80.0	<0.009	0.04	
Sn Cr	<0.07		<0.07	
	285.22	3/2.35	204.96	
	0.03	0.03	0.02	
	0.01	0.00	0.01	
	0.13	0.01	0.01	
	1.12	8.79	2.37	
	0.92	19.43	0.30	
W	0.22	0.12	0.08	
	0.45	0.06	0.11	
Yb	0.04	0.00	0.01	
Zn	318.20	55.60	151.33	
Zr	0.13	0.11	0.08	

Table 7.2 Trace element groundwater chemistry from 3 Palm Valley sites

BORE ID

The less than (<) values are detection limits for the method

As Th (a daughter product of U decay) is nearly always more abundant than uranium in rocks (Hem, 1985) these data are taken to indicate that U levels are not anomalously high in the Hermannsburg Sandstone. Examination of Gamma Ray geophysical logs of bores drilled in the Hermannsburg Sandstone held in the NTG database show a typical sandstone low count response, this is indicative of lower U and Th levels than those typically found in clays or shales.

Consequently, it is concluded that hypogene ³⁶Cl production is not a significant consideration when assessing ³⁶Cl results from Palm Valley.

7.2.3.6. Anthropogenic Sources

Anthropogenic ³⁶Cl production (mainly from surface testing of nuclear weapons) is reviewed in section 7.1.3.1. Elevated ³⁶Cl/Cl ratios have been attributed to weapons testing elsewhere (e.g. Davis et. al., 2000). As ³⁶Cl/Cl ratios are consistently at or below known nearby atmospheric ratios (Keywood, 1995) and carbon-14 values are less than 100 %MC, there is no prima facie evidence to suggest ³⁶Cl results at Palm Valley are complicated by bomb pulse affects.

7.2.3.7. Addition of "dead" chloride.

In Chapter 8 it is noted that many previous attempts to use ³⁶Cl decay to date groundwater have suffered because subsurface addition of chloride is indicated and consequently ³⁶Cl decay results were liable to ambiguous interpretation. At Palm Valley the plots of ³⁶Cl and Cl data (Figs. 7.5 & 7.6 discussed above) do not indicate that subsurface addition of Cl has occurred.

7.2.3.8. Cl/Br ratios

Many workers (e.g. Andrews and Fontes, 1993; Mazor, 1992; Davis et al., 1998a, 2000; see Section 7.1.4) recommend comparing groundwater Cl/Br ratios, with reference to those of seawater, as a check for addition of extra aquifer derived chloride. The results of four Cl/Br analyses from water samples collected in 2002 are shown in Table 7.1. As these ratios are less than the Cl/Br (mass) ratio of seawater (i.e. 290,

Hem, 1985), which is also expected in inland Australian rainwater due to loss of chloride from sea salt aerosol by volatilisation with H_2SO_4 and HNO_3 (Keywood et al., 1997), an atmospheric source of chloride can be assumed here. It should be noted that inland rainfall with Cl/Br ratios less than seawater have also been described in Nigeria by Goni et al., 2001.

Given the significance of subsurface chloride addition to ³⁶Cl decay dating studies it is worth looking at Cl/Br ratios in more detail. As part of her investigation into rainfall across a north to south transect of continental Australia, and a west to east transect across Western Australia, Keywood (1995) also recorded Br levels in precipitation. Bromide and chloride concentrations from precipitation were determined by Ion chromatography. Given the low concentrations recorded, accuracy of Br analyses are reported by Keywood (1995) to be less (9.6%) than those of Cl (0.31%). Even with these accuracy constraints compilation of these data allow a check on atmospheric Cl/Br ratios in the central Australian region.

Table 7.3 shows a compilation of Cl/Br ratio values calculated from Keywood's (1995) data obtained from her central Australian sample stations (i.e. stations that are a similar distance from the coast as Palm Valley.). Figure 7.7 shows the location of these stations and a graphical display of the data presented in Table 7.3. These data confirm that Cl/Br levels are indeed below the seawater ratio of 290 in inland Australian precipitation. Seasonal and perhaps amount related variation in precipitation composition along with the limited accuracy of the Br measuring technique probably preclude quantitative use of the recorded atmospheric Cl/Br ratios to determine actual amounts, if any, of subsurface Cl addition. Nevertheless, these values indicate groundwater Cl/Br ratios in central Australian aquifers should be below the seawater ratio when there is no contamination by subsurface derived chloride.

Thus, comparison of groundwater Cl/Br data to the seawater ratio is seen to be an appropriate technique to investigate the sources of chloride in central Australian aquifers. Furthermore, contamination of groundwater by windblown halite (i.e. a possible source of dead chloride that could lower ³⁶Cl/Cl ratios in groundwater) derived from distant central Australian playas (i.e. halite Cl/Br = 1000+, Davis et al., 1988b) is not indicated here.

STATION	SEASON	RAIN	CI	Br	Cl/Br
		(mm)	ა g/L	ს g/L	(mass)
Carnegie	A91	16	170.15	0.80	213
Carnegie	W91	153	181.50	3.19	57
Carnegie	SP91	1	187.88	0.80	235
Carnegie	S91	55	372.21	2.40	155
Carnegie	A92	194	195.32	7.19	27
Carnegie	W92	31	167.67	1.60	105
Carnegie	SP92	20	203.12	3.19	64
Carnegie	S92	33	38.99	1.60	24
Gunbarrel	A91	36	256.65	1.60	161
Gunbarrel	W91	118	120.52	2.40	50
Gunbarrel	SP91	14	197.45	1.60	124
Gunbarrel	S91				
Gunbarrel	A92	265	141.08	4.79	29
Gunbarrel	W92	31	221.55	1.60	139
Gunbarrel	SP92		0.00	0.00	
Gunbarrel	S92	79	75.86	2.40	32
Everard Jn	A91	11	147.11	0.00	
Everard Jn	W91	118	97.13	3.19	30
Everard Jn	5P91	50	218.21	1.60	174
Everard Jn	591	101	0.00	0.00	20
Everard Jn	A9Z	191	163.06	0.39	26
Everard Jn	VV92	22	153.65	1.60	90
Everard In	5792				
Everaru Jri	392	27	206.25	2 10	02
Wintinna	SD02	12/	290.33	J.19 7 10	30
Wintinna	SC 92	21	233.90	1.19	202
Wintinna	A03	31	152 /3	3.00	202
Wintinna	M/93	49	375 75	5.59	67
Wintinna	SD03		272.05	1 70	57
Wintinna	593	74	272.00	3 99	56
Wintinna	Δ <u>9</u> 4	43	198 51	3 99	50
Alice Sp	W/92	10	81 53	0.80	102
Alice Sp	SP92	69	163.06	3.99	41
Alice Sp	S92			0.00	
Alice Sp	A93	149	77.99	18.37	4
Alice Sp	W93	14	189.29	1.60	118
Alice Sp	SP93	62	400.57	6.39	63
Alice Sp	S93	85	114.50	3.19	36
Alice Sp	A94	2	54.95	0.00	
Tennant Ck	W92	0	5.67	0.00	
Tennant Ck	SP92	57	236.44	3.99	59
Tennant Ck	S92				
Tennant Ck	A93				
Tennant Ck	W93	5	25.17		
Tennant Ck	SP93	59	404.11	7.19	56
Tennant Ck	S93	56	804.68	11.18	72
Tennant Ck	A94	0	14.18	0.00	
	AT OTATIONO				
MEAN CI/Br	AISTATIONS			ſ	VIEAN CI/Br
Carnegie					79
Gunbarrel					70
Everad Jn					70
Wintinna					65
Alice Springs					32
Tennant Ck.					62

Table 7.3 Keywood's (1995) central Australian atmospheric Cl/Br (mass) ratios

NB. A = Autumn, W = Winter, SP = Spring, S = Summer.

Note also different period measured for WA sites. Data derived from Keywood, 1995.



Figure 7-7 Inland Australian atmospheric Cl/Br ratios as recorded at several stations over several seasons by Keywood, 1995. The location of these stations in relation to Palm Valley is shown. The graphical display of the Cl/Br data for each season sampled at each station is based on data presented in Table 7.3. Actual seasons sampled vary slightly between WA and the SA and NT stations (see Table 7.3). These data confirm that atmospheric Cl/Br levels are below the seawater ratio of 290 in inland Australian precipitation.

7.2.3.9. Complementary data

Edmunds (1995), Edmunds and Smedley, (2000), Edmunds et al. (2003) and Guendouz et al. (2003) note that other chemical factors, given the right conditions, can sometimes be used as a proxy to indicate groundwater age. For instance in confined aquifer systems decrease in nitrate concentration can be taken as an indication of an increasingly reducing environment and hence increase in groundwater age. Similarly, some trace elements such as Lithium behave conservatively and are not typically subject to solubility constraints. Therefore, such elements can be used as a proxy indicator of relative groundwater age as concentration is expected to increase along a flow path as dissolution associated with groundwater evolution continues. The Hermannsburg Sandstone is unconfined and apart from the data shown Table 7.2 no other trace element data are available. Thus, no other complementary chemical indicators of groundwater age are available for this study.

7.3. Conclusion

The plots of ³⁶Cl/Cl ratio against ³⁶Cl atoms (Fig. 7.6) and chloride (Fig. 7.5) show no apparent loss or gain of chloride within the aquifer. These data and other considerations outlined above all support radioactive decay as being the cause of variation of ³⁶Cl/Cl ratios within the Hermannsburg Sandstone at Palm Valley. Hence groundwater age varies from recent to around 300 ka at Palm Valley.

8. Relevance of this study

8.1. Background

The use of ³⁶Cl decay dating to provide a window beyond the radiocarbon dating range for groundwater resource assessment was championed with the advent of suitable AMS techniques in the late 1970s. The review of ³⁶Cl decay dating studies outlined below, however, shows that the early promise has since been rendered a controversial technique. It seems that depending on opinion, results to date have been difficult or impossible to interpret. Many of the data suites from previous studies display these ambiguous trends with no clear decay component when plotted on a ³⁶Cl/Cl versus Cl chart (see Section 8.2.4).

The controversial status of ³⁶Cl decay dating of groundwater is mirrored in textbooks on isotope hydrology. Clark and Fritz (1997) express doubt on the validity of the technique when subsurface chloride addition is indicated. However Phillips (2000) thinks ³⁶Cl decay dating is still feasible when well over 50% of aquifer chloride is derived from subsurface sources.

8.2. Review of groundwater ³⁶Cl decay studies

8.2.1. Introduction

Studies of groundwater using ³⁶Cl as a tracer have been undertaken in many aquifers around the world. While there have been many applications of ³⁶Cl in groundwater studies it is commonly acknowledged (eg. Davis 1998a: Andrews and Fontes, 1992; Phillips, 2000) that interpretation of results has often been difficult. This is usually because of doubt about sources and sinks of ³⁶Cl in the subsurface and/or the magnitude of atmospheric ³⁶Cl/Cl ratio at time of recharge.

This is particularly so for groundwater dating where, as previously discussed in Section 7.1, the ³⁶Cl/Cl ratio may be used to estimate ages of groundwaters only if certain assumptions are made (Andrews and Fontes, 1992). Davis et al. (2000) note in most instances that the main problem with ³⁶Cl dating is an inverse relationship between chloride concentration and ³⁶Cl/Cl ratio being observed in the aquifer system. Indeed Andrews and Fontes (1993) make the general observation that dating of groundwater using ³⁶Cl cannot be undertaken when chloride is added along flowpaths. However, in many ³⁶Cl groundwater studies (discussed below) addition of chloride in the subsurface has been accounted for and age estimates made. Phillips (2000) states ³⁶Cl decay dating should be feasible even when up to 90% of groundwater Cl is aquifer derived in most instances. The necessity of estimating/assuming the amount and rate of chloride addition over long time periods must, however, diminish the reliability and accuracy of the ³⁶Cl method in these circumstances.

8.2.2. Renowned ³⁶Cl decay groundwater studies

8.2.2.1. Great Artesian Basin

An example which deals with the issue of chloride addition along flow paths is the pioneering ³⁶Cl decay dating Australian studies in the Great Artesian Basin (GAB). The ³⁶Cl decay dating of the GAB groundwater by Bentley et al. (1986a) and Torgersen et al. (1991) have been reinterpreted by Mazor (1992), Mazor and Nativ (1994) and Andrews and Fontes (1992, 1993). These reinterpretations question the groundwater age determinations presented as it is considered that sources and sinks of chloride along basin flow paths were inadequately accounted for.

The original authors argue that they have recognised and allowed for the addition of Cl⁻ along flow lines (down dip) in these basins. Data plots show general trends of ³⁶Cl/Cl and ³⁶Cl to decrease along perceived slow flow paths, ³⁶Cl decay commensurate with expected range of hydrogeological parameters and likely atmospheric ³⁶Cl/Cl inputs is indicated, thus some addition of Cl⁻ at a slow rate is invoked to explain increase of chlorinity along flow path. Andrew and Fontes (1992, 1993) argue that mixing of two

³⁶Cl/Cl end members in the aquifer, one at atmospheric level, the other at aquifer secular equilibrium, can equally well explain these ³⁶Cl and Cl⁻ data. This point is then illustrated with theoretical mixing curves plotted on ³⁶Cl/Cl against Cl⁻ concentration diagrams. Andrews and Fontes (1993) show that variation of the mixing rate (from 100+ millions of years to instantaneous) of subsurface derived Cl⁻ with input atmospheric derived Cl⁻ to attain observed aquifer Cl⁻ levels has no discernible affect on mixing trend lines. Given this potential ambiguity, Andrews and Fontes (1993) state ³⁶Cl decay dating of old groundwater should be limited to cases where chlorinity is constant. In reply, Torgersen and Phillips (1993) note among other points that the extreme example of instantaneous mixing is not evident as ³⁶Cl decay in the GAB waters and therefore ³⁶Cl decay dating of groundwater in the GAB is sound.

Mazor (1992) Mazor and Nativ (1994) and Mazor (1995) introduce a concept of stagnant groundwater that may affect some parts of the GAB, and postulate measured ³⁶Cl/Cl ratios only reflect aquifer secular equilibrium values and not ³⁶Cl decay and thus no dating is possible. This concept, which is in contrast to standard theories of groundwater flow (eg Toth 1995), has been refuted by various other workers (Kellett et al. 1993; Phillips, 1993). Fontes and Andrews (1993b) also question much of the interpretation made by Mazor (1992) of previous ³⁶Cl groundwater studies. Mazor (1993a,b,c) reiterates his arguments in a series of replies to this criticism but there is no apparent resolution of these differences of opinion.

The debate on the applicability of ³⁶Cl decay dating in the GAB and elsewhere is not yet resolved. Clark and Fritz (1997) in their text on isotope hydrogeology discuss the ³⁶Cl data for the GAB and reiterate the arguments raised by Andrews and Fontes (1992). They plot GAB groundwater samples on a ³⁶Cl/Cl versus Cl⁻ diagram and illustrate how a mixing process between a subsurface high Cl⁻ low ³⁶Cl brine and a low Cl⁻ higher ³⁶Cl recharge water also explains the GAB results. Clark and Fritz (1997) do, however, concede that the difference in ³⁶Cl activities between recharge and distal groundwaters in the GAB indicate water is at least 300 ka (one ³⁶Cl half life) old. This discussion which incorporates aspects of several opinions on ³⁶Cl decay dating in the GAB illustrates that the use ³⁶Cl as a dating tool is still somewhat controversial.
Subsequent ³⁶Cl studies have been undertaken in the GAB. Radke et al. (2000) in their compilation of all GAB hydrochemical data note that ³⁶Cl data only provide limited information on groundwater flow rates as in much of the GAB ³⁶Cl/Cl ratios are commensurate with expected in situ ³⁶Cl production rates. In other areas, interpretation of ³⁶Cl data requires an estimate of the magnitude and rate of Cl addition to the aquifer to be made. Radke at al. (2000) note an inverse relationship between ³⁶Cl/Cl ratio and alkalinity and speculate that further work on this association may allow ³⁶Cl decay age determinations to be extrapolated further into low ³⁶Cl/Cl regions of the GAB.

Love et al. (2000) sampled bores in the western GAB in South Australia and note that unmeasurable rates of diffusion of Cl from aquitards and potential past secular variation in atmospheric ³⁶Cl fallout preclude determination of absolute groundwater age in the western GAB. They prefer to use ³⁶Cl/Cl data to compute relative age differences (hence groundwater flow velocity) between sample points. Love et al. (2000) show Cl concentration decreases along flow path in some areas and propose upward leakage with increasing head may remove solutes from the aquifer. The possibility that decrease in chloride concentration downdip may also occur as the result of less evaporation due a different climatic regime at the time of recharge is also noted. This situation, which is contrary to that found in the eastern GAB (Bentley et al., 1986b), allays some of the concerns raised by Andrews and Fontes (1992) for ³⁶Cl studies in the GAB discussed above.

In a follow up study, Lehmann et al. (2003) calibrate the western GAB ³⁶Cl data presented by Love et al. (2000) with Krypton-81 measurements and calculate some bores to have around 80% subsurface derived Cl, which they consider unsuitable for ³⁶Cl decay dating. The technique of age dating groundwater with ⁸¹Kr is only suitable for confined aquifers and has the logistical inconvenience of requiring large volumes of sample for analyses and is a major technical undertaking (Collon et al., 2000) which explains in part why this isotope has not used in conjunction with ³⁶Cl more often. Sturchio et al. (2004) have now developed a ⁸¹Kr measurement technique that requires less sample volume (2 tonne), if such improvements in sampling logistics continue this technique may be more readily applied in future groundwater studies.

The hydrological characteristics of the GAB have also been used as a template for a hypothetical reactive transport modelling exercise of ³⁶Cl decay and aquitard diffusion of ³⁶Cl by Bethke et al. (2000). This study concludes groundwater age in the GAB is probably older than that indicated from Darcy flow calculations, however many of the assumptions needed for previous studies in the GAB still need to be invoked for this work. Park et al. (2002) developed this modelling exercise further noting the potential limitations of ³⁶Cl decay dating given conditions similar to those found/perceived in the GAB. Among these potential limitations, it is noted ³⁶Cl decay age cannot be interpreted when groundwater is only moderately saline (Cl = 75 – 150 mg/l). Park et al. (2002) assume an input Cl concentration to their model of 23 mg/l, as used in previous GAB studies, thus the threshold percentage of subsurface derived Cl that precludes ³⁶Cl decay dating based on this work is around 65 – 85 %.

8.2.2.2. Milk River

The International Atomic Energy Agency (IAEA) sponsored study of the Milk River Aquifer, southern Alberta, Canada by Phillips et al. (1986b) encountered similar problems to the GAB, in that chloride concentration increases down flow path in this aquifer also. The results were difficult to interpret here as this aquifer is considered much more hydrochemically and hydraulically complex than the GAB. Phillips et al. (1986b) reconcile ³⁶Cl decay age and hydraulic data by proposing the additional chloride is derived from an ion filtration process and discount the possibility that additional Cl⁻ is sourced from the flushing of connate brines. In a subsequent appraisal of this aquifer Hendry and Schwartz, 1988 prefer a diffusion model to account for additional Cl⁻, and deduce a younger groundwater age in the aquifer than Phillips et al. (1986b). Nolte et al. (1991) obtained further ³⁶Cl data and further developed a Cl⁻ diffusion flux model for the aquifer and estimated the maximum groundwater age to lie between the respective age estimates of 2 Ma and 0.5 Ma by Phillips et al. (1986b) and Hendry and Schwartz (1988). Fabryka-Martin et al. (1991) resolved the conflict on theories of the source of additional Cl⁻ to this aquifer by proposing an intra aquifer origin as the dominant source. Other processes such as ion filtration and diffusion from aquitards are still considered potentially valid origins of extra Cl, but in a minor capacity only. Fabryka-Martin et al. (1991) propose local diffusion of saline water primarily from low permeability units within the Milk River Formation as the simplest explanation of source of additional Cl⁻ within the Milk River Aquifer. A common subsurface input source for all halides in the aquifer is indicated as halide ratios approach constant values along flow paths. Even though up to 90 % of the chloride at the distal ends of this aquifer may be derived from subsurface origins Fabryka-Martin et al. (1991) still calculate groundwater age from ³⁶Cl decay data at around 1 Ma.

Mazor (1992), in a similar vein to his criticisms of the GAB ³⁶Cl studies mentioned above, questions whether the Milk River Aquifer is hydraulically continuous. Phillips (1993) again refutes Mazor's interpretation and cites many studies that confirm the hydraulics of the Milk River Aquifer.

Andrews and Fontes (1992) also discuss the Milk River Aquifer ³⁶Cl results and note high ³⁶Cl inputs compared to predicted local atmospheric ³⁶Cl/Cl ratios may reflect a component of ³⁶Cl produced in the epigene zone or barring this it may mean that bomb ³⁶Cl is leaking through confining beds which may confuse interpretation. As mentioned above (section 7.1.3.3) epigene production is now considered an unlikely complication in groundwater studies. A recent empirical study of shallow groundwater ³⁶Cl/Cl ratios in the USA (Davis et al., 2003) now show higher ratios than previously predicted for this region of North America. Thus this indication of leakage of bomb ³⁶Cl, noted by Andrews and Fontes (1992) may now not be significant.

Frohlich et al. (1991) give an overview of Milk River Aquifer studies and support the work of Fabryka-Martin (1991) to explain the various Cl⁻ addition mechanisms proposed. Frohlich et al. (1991) also notes that ³⁶Cl decay age estimates, with in situ production accounted for, were only around 30% older than those calculated using zero in situ production. However, it is also noted that when groundwater chloride reaches approximately 500 mg/l in the Milk River Aquifer, in situ production dominates the ³⁶Cl signature. Hence no reliable ³⁶Cl decay ages are considered likely when chloride content is high. The high atmospheric ³⁶Cl/Cl (input) ratios compared to predicted subsurface in situ ³⁶Cl production rates help the application of ³⁶Cl decay dating at Milk River.

Clark and Fritz (1997) however, in keeping with points raised by Andrews and Fontes (1992) make the comment that at distal ends of this aquifer the ³⁶Cl brought in with the additional subsurface Cl⁻ reflects the residence time of the brine and that no ³⁶Cl decay

dating is possible. This interpretation, which is in contrast to conclusions drawn by Phillips (2000) in another isotope hydrology text book again illustrate the differing opinions on the applicability of ³⁶Cl decay dating when addition of subsurface Cl⁻ is apparent. Phillips (2000) presents plots that show ³⁶Cl/Cl ratio and ³⁶Cl concentration to decrease with distance along flow path to back his argument. Process charts, like those shown in Figure 7.2, which are used by Andrews and Fontes (1992) to illustrate the ambiguity of ³⁶Cl decay dating when subsurface additions of Cl⁻ occur in the GAB are not illustrated. Instead Phillips (2000) presents a diagram that, given certain assumptions (i.e. atmospheric ³⁶Cl/Cl 30 times greater than in situ secular equilibrium) shows subsurface production of ³⁶Cl from additional subsurface Cl would not be expected to dominate atmospheric ³⁶Cl in low salinity groundwater until 500 ka of residence and when over 90% of Cl⁻ was derived from subsurface sources. Given this debate, it is readily appreciated that at the Milk River Aquifer the validity of ³⁶Cl decay dating is controversial. This is largely due to the uncertainty regarding the source and rate of subsurface additions of chloride that occur within the aquifer post recharge.

8.2.2.3. Magadi-Natron Basin

Another ³⁶Cl study that has sparked debate in the literature is the examination of Magadi-Natron Basin in the East African Rift, near the Kenyan-Tanzanian border, by Kaufman et al. (1990). They date brine in the basin at around 760 ka by comparing ³⁶Cl/Cl ratios in dilute modern waters with those found in the brine. To do this the assumption is made that the basin is a closed system with Cl derived from accumulated precipitation since the basin was formed. Concentration by evaporation and continual mixing of a component of dilute recharge with recirculating brine is also assumed.

Fontes and Andrews (1993a) comment on this work in the Magadi-Natron Basin and dismiss the attempt to use ³⁶Cl decay to date brines as they consider that there are to many uncertainties in the estimated input (atmospheric) ³⁶Cl/Cl ratio used and epigene and hypogene ³⁶Cl production are not accounted for adequately. Furthermore, they note the influence of bomb peak ³⁶Cl fallout in the closed system has not been calculated and that mass balance considerations of retaining all Cl fallout within the basin in the period proposed could not be accommodated unless considerable halite was deposited.

Kaufman et al. (1993) respond by noting that the inherent uncertainties in estimating atmospheric fallout of ³⁶Cl, bomb peak fallout, epigene and hypogene production and past atmospheric ³⁶Cl/Cl ratios can equally be invoked to refute the arguments of Fontes and Andrews (1993a).

In his text book chapter on ³⁶Cl Phillips (2000) cites the Kaufman et al. (1990) work in a discussion on ³⁶Cl as a tracer for salinity studies without mention of the subsequent debate outlined above. It is not known if this is a tacit rebuttal of Fontes and Andrews (1993a) concerns. This point notwithstanding, it is apparent from the Magadi-Natron study that when uncertainty of sources and sinks of Cl in the subsurface exist, doubts about the viability of ³⁶Cl decay dating method will also exist.

8.2.2.4. Mazowsze Basin

Dowgiallo et al. (1990) sampled the Oligocene aquifers of the Mazowsze Basin in Poland for ³⁶Cl analyses. The results are taken to indicate ³⁶Cl decay has occurred and that some groundwater is over 200 ka old, however, the likelihood of addition of subsurface derived chlorides is noted.

Zuber et al. (2000) criticise this interpretation as it does not match their understanding of the flow dynamics where maximum residence time is expected to be only 35 ka and that no trend of 36 Cl/Cl or 36 Cl decreasing along flow path in the aquifer is discernible. Furthermore, it is noted that noble gas data indicates these groundwaters are connected to the atmosphere. It is argued that the differences in 36 Cl/Cl ratios observed in this basin (488 to < 20 by 10⁻¹⁵), taken to indicate 36 Cl decay by Dowgiallo et al. (1990), may be explained by variation in the amount of epigene 36 Cl/Cl production at different recharge sites. This potential for complications of 36 Cl/Cl interpretation due to recharge waters picking up epigene produced 36 Cl and thereby causing differences in 36 Cl/Cl inputs throughout the aquifer was originally noted by Andrews and Fontes (1992) when discussing the Milk River Aquifer. As mentioned above, it is now generally thought unlikely that epigene production is a significant factor to be considered in 36 Cl groundwater studies, particularly at altitudes less than 2000 m. Thus it may be that unknown complications of the flow system and subsurface chloride need to be resolved before assessing 36 Cl/Cl data for this basin.

Further potential for complication is noted by Zuber et al. (2000) in that hydrochemistry of the Oligocene aquifer is described as a mosaic throughout the basin and factors such as the influence of salt diapirs and discharge from underlying Mesozoic units are called upon to explain variations in chloride concentrations. This observation suggests that not all sources and sinks of chloride to the aquifer are known and therefore reliability of the ³⁶Cl decay dating by Dowgiallo et al. (1990) is problematic. These uncertainties may equally render some of Zuber et al. (2000) conclusions on groundwater age uncertain.

It is apparent that, as noted by both Dowgiallo et al. (1990) and Zuber et al. (2000), more information is needed to properly assess groundwater age in the Mazowsze Basin. The process chart of ³⁶Cl/Cl versus Cl presented by Dowgiallo et al. (1990) illustrates the complexities in the chloride cycle of this basin, as no clear trends are discernible.

8.2.2.5. Murray Basin

The studies of ³⁶Cl in the Murray Basin in south-eastern Australia by Kellett et al. (1990) and Davie et al. (1989) have also resulted in some debate. In this basin, decreasing ³⁶Cl/Cl with increase in chloride concentration is taken to show addition of dead Cl to the confined aquifer and not ³⁶Cl decay. The ³⁶Cl data are considered roughly consistent with expected atmospheric ³⁶Cl/Cl ratios in the vicinity and are used to elucidate recharge processes. Mazor (1992) by way of a graphical technique to determine ³⁶Cl/Cl input ratio considers evidence of ³⁶Cl decay exists and that some of the groundwater in the Murray Basin maybe very (100s ka) old. Kellett et al. (1993) and Fontes and Andrews (1993b) dismiss most of the points raised and propose that Mazor (1992) invokes assumptions that do not match the known hydrological environment to support his controversial arguments. Some of this debate also considered likely atmospheric ³⁶Cl/Cl inputs in the Murray Basin region. Simpson and Herczeg (1994) as outlined previously, also discuss aspects of the atmospheric ³⁶Cl/Cl ratio in this region.

8.2.3. Other Studies that attempt ³⁶Cl decay dating

8.2.3.1. Introduction

Studies that attempt ³⁶Cl decay dating of groundwater are still reported in the literature. These studies have seemingly not attracted the same level of debate on ³⁶Cl decay dating as that outlined above for the GAB and Milk River Aquifer systems. As most of these studies also show an inverse Cl to ³⁶Cl/Cl relationship, the literature debate outlined above is probably equally applicable to some of these studies. The reason for a lack of debate may reflect a general acceptance of the limitations of ³⁶Cl decay dating and/or the untimely deaths of both John Andrews and Jean-Charles Fontes who led much of the discussion on the application of ³⁶Cl decay dating in aquifers without constant chloride content. Some geothermal systems, as discussed above in section 7.1.3.4, have also been dated using the ingrowth of ³⁶Cl compared to estimates of host rock ³⁶Cl secular equilibrium rates.

8.2.3.2. International studies

Penny et al. (2003) present some ³⁶Cl/Cl ratio data from two transects in a coastal plain aquifer in Alabama, USA as a small part of an integrated groundwater study. If natural ³⁶Cl decay along flow paths is assumed groundwater ages >100 ka are calculated. Such ages are commensurate with the known hydraulic characteristics of this aquifer. Caution is recommended when viewing these data as progressive mixing from low ³⁶Cl high Cl residual evaporate brines is known to occur in this region. This uncertainty is reflected by the fact that the ³⁶Cl/Cl versus chloride concentration chart presented shows no clear decay trend. The low chloride and decreasing ³⁶Cl/Cl ratio along flow path data presented for the eastern transect look, on face value, promising for the application of ³⁶Cl decay dating. Perhaps further investigation in this area will better define ³⁶Cl decay ages along this transect.

Sturchio et al. (2004) correlate six ³⁶Cl/Cl ratio results from the Nubian aquifer in Egypt against ⁸¹Kr decay age results and fit a decay curve to estimate ³⁶Cl decay ages. Discordance of ⁸¹Kr - ³⁶Cl ages of some samples are attributed to subsurface addition of Cl or hypogene ³⁶Cl. Different atmospheric ³⁶Cl/Cl input ratios between some samples

are also noted to be a possible explanation of this discordance. Sturchio et al. (2004) conclude these results demonstrate the vulnerability of ³⁶Cl decay dating integrity to subsurface Cl additions, whereas the new ⁸¹Kr decay dating method they present is considered much less vulnerable to interpretation problems associated with subsurface sources or sinks of Kr. When these ³⁶Cl/Cl ratios are compared to Cl concentration, an inverse relationship between Cl and ³⁶Cl/Cl ratio is apparent rather than a decay trend. Thus uncertainty about the amount and rate of subsurface Cl addition render the ³⁶Cl decay aspects of this study liable to ambiguous interpretation like the GAB discussed above.

Cresswell et al. (2000) present ³⁶Cl data from three deep wells and a spring from the Kathmandu Basin in Nepal and make some preliminary estimates of groundwater age. of between 200 and 400 ka. The ages are considered feasible within the hydrogeological context. A slight increase in Cl⁻ concentration with decrease in ³⁶Cl/Cl is, however, evident from this limited data set. Given this flag for potential difficulties of interpretation of ³⁶Cl results (e.g. Andrews and Fontes, 1992) and the lack of chemical or other isotope data to corroborate this story, the estimate of groundwater age is rightly considered provisional.

Zhou et al. 2001 report on ³⁶Cl results measured in the Hebei Plain in eastern China. Differences in ³⁶Cl/Cl ratio measured at surface (62×10^{-15}) and deep borehole (23×10^{-15}) were use to calculate that some groundwater in Quaternary sediments is over 250 ka old. This interpretation relies upon the assumption that ion filtration is a valid process (see Fig 7.2) that can reconcile differences in chloride concentration between some samples. This assumption and the fact that only four analyses are available make this calculation of groundwater age potentially dubious as other processes such as addition of subsurface chloride may equally well explain these limited results. It may be that ongoing work on these heavily stressed aquifers in the North China Plain (e.g. Foster et al., 2004) will refine the use ³⁶Cl decay dating in this region.

8.2.3.3. Australian Studies

8.2.3.3.1. Attempted ³⁶Cl decay dating

Several other Australian studies apart from the GAB and Murray Basin discussed above have attempted to date groundwater using ³⁶Cl decay as a chronometer.

In the Northern Territory of Australia ³⁶Cl results from Amadeus Basin aquifers have been reported by Jacobson et al. (1994) and Cresswell et al. (1999b). Age dating of groundwater is basically attempted by comparing the ³⁶Cl/Cl ratios from shallow bores with ³⁶Cl/Cl results obtained from deep petroleum exploration wells that intercept the same aquifer at depth. While ages of several hundreds of thousands of years are determined, which is in keeping with the known flow dynamics of this basin, there is a readily observable inverse relationship between ³⁶Cl/Cl ratio and Cl concentration which needs to be accounted for. Cresswell et al. (1999b) deal with this potential interpretation problem by estimating a rate of addition of subsurface chloride so ³⁶Cl decay dating can be attempted. A mixing rate of 0.2 mg/L of Cl⁻ per 1000 years is assumed and plotted as a best fit to link data plot points from shallow and deep samples on a ³⁶Cl/Cl versus Cl content diagram. No heed is given to any of the concerns raised by some workers (as discussed above for the GAB) about the validity of ³⁶Cl decay dating when subsurface additions of chloride are present. While the air drilled petroleum exploration holes in the Amadeus Basin gave a rare opportunity to sample deep aquifers, the significant but difficult to quantify addition of surface make - up water and compressed air during drilling has the potential to complicate the interpretation of ³⁶Cl results. This is evidenced by the results of some the samples taken while drilling the Palm Valley 8 exploratory gas well. At 990 m depth a concentration of ¹⁴C equal to 29 pmc (i.e. less than 30 ka old) was recorded (NT Government bore file RN 16018), however, the 897 m sample above is recorded to have a ³⁶Cl/Cl of 99 x 10⁻ ¹⁵ which is taken to indicate an age of several 100 ka using regional shallow bore 36 Cl/Cl ratios of around 176 x 10⁻¹⁵ as an input ratio, as did Creswell et al. (1999b). The 990 m sample was the lowest conductivity water produced from the hole and was thought at the time of sampling to be the least contaminated by drilling fluids when drilling was monitored in 1994 (NTG, Water Resource Branch file No. 60.16). The

sample at 928m recorded a ¹⁴C concentration of < 2% pmc (i.e. beyond detection range of ¹⁴C). The difficulty in deciphering these results, where apparent age is not correlated to depth, highlight the difficulties in obtaining uncontaminated groundwater samples during potentially favourable air drilling of petroleum wells. While air drilling of petroleum wells in the consolidated sediments of the Amadeus Basin gives a rare opportunity to access deep aquifers, the attempts to sample groundwater as a piggy back operation to petroleum drilling has resulted in some doubts of groundwater sample integrity. Incorporation of groundwater sampling into the well prognosis (e.g. packer test of aquifer) from the outset is probably the only, but expensive, way to ensure groundwater sample integrity from deep aquifers intersected by hydrocarbon drilling operations.

Cresswell et al. (1999a) report on ³⁶Cl/Cl results from bores spread over a wide area of arid country west of Alice Springs. Samples for ³⁶Cl analyses were opportunistically obtained during a survey of drinking water quality from bores in the region. Consequently no specific hydrogeological system was investigated by such methods as sampling sequential bores down dip in the same aquifer. Nevertheless, some preliminary conclusions on groundwater age based on apparent ³⁶Cl decay were attempted. ³⁶Cl/Cl ratios from 18 bores are observed to roughly fall into two distinct populations with 36 Cl/Cl ratios of 170 and 205 x 10 $^{-15}$. The higher value was taken as an indication of input ³⁶Cl/Cl ratio in the area and the lower group to represent groundwater 80 ka old, which may match an interglacial pluvial event. While no inverse trend of ³⁶Cl/Cl ratio and Cl⁻ is apparent from these results, chlorinity varies markedly and different degrees of evaporation for each sample needs to be assumed (rather than subsurface Cl⁻ addition etc) to assess groundwater age from these results. These assumptions, discordance with ${}^{14}C$ data (all samples have measurable ${}^{14}C$) and measured atmospheric ³⁶Cl/Cl ratio in the region (325 x 10⁻¹⁵, Keywood, 1995), mean reliability of this interpretation is uncertain. Other interpretations such as all water is less than 40 ka old as indicated by ¹⁴C results and ³⁶Cl/Cl variation is due to differing amounts of contamination by subsurface Cl or ³⁶Cl addition remain valid alternatives. Interpretation of this data set may benefit from consideration of additional chemical data and better integrating physical hydrogeological factors.

Jacobson et al. (2000) present some ³⁶Cl and other isotope data from Cainozoic and underlying Palaeozoic aquifers near Ayers Rock (Uluru), Northern Territory. Considerable discordance of ¹⁴C and ³⁶Cl indicated ages from most samples were taken to indicate subsurface mixing of different 'vintages' and possibly some diffusion of modern ¹⁴C below the water table. A few samples with ¹⁴C concentration below detection limits gave the slightly lowest ³⁶Cl/Cl ratios and these data allowed an age of over 200 ka to be calculated for some samples. The plots of ³⁶Cl data from this region on ³⁶Cl atoms against Cl (mg/L) and ³⁶Cl/Cl against Cl (mg/L) charts did not, however, show any clear ³⁶Cl decay trend. On comparison with the processes charts shown in Figure 7.2, data points from this study are seen to plot along a dilution – evaporation process line. This lack of clear ³⁶Cl decay trend line and discordant isotope ages make interpretation of these results complicated. While it may be that these data are a consequence of complex hydrogeology in the region, a better understanding of chloride flux in the subsurface would aid interpretation here.

Bureau of Rural Sciences (2001), as part of an assessment of low permeability strata to house nuclear waste in the centre of South Australia, measured ³⁶Cl in nine bore holes. Due to a large variation in chlorinity between samples a ³⁶Cl atoms against Cl (mg/L) plot showed these samples to be distributed along a dilution – evaporation line. Although such a trend usually precludes ³⁶Cl decay dating, the authors note that if two distinct evaporation trends are assumed, then a difference of 100 ka is indicated between trend lines. As no consideration of possible subsurface addition of chlorides or how these samples maybe hydraulically related is undertaken, the reliability of such an interpretation is uncertain. Further work is needed to confirm these estimates of groundwater ages.

8.2.3.3.2. Unsuitable for ³⁶Cl decay dating

Several previous Australian studies have considered uncertainties on sources and sinks of subsurface chloride and/or lack of great hydraulic age to preclude the use ³⁶Cl as a groundwater dating tool.

Commander et al. (1994) and Turner et al. (1995) present some ³⁶Cl results from Palaeochannel aquifers in the Goldfields region of Western Australia but found probable mixing of remobilised playa salts in these hyper-saline aquifers hindered any quantitative interpretation of ³⁶Cl results.

Hostetler (2000, 2002) presents some ³⁶Cl data from shallow bores in the Alice Springs region of the Amadeus Basin. Because there is no observed relationship between ¹⁴C and ³⁶Cl/Cl he considers groundwater too young to be dated by the ³⁶Cl decay method. No decay trend is evident from ³⁶Cl/Cl v Cl plots of these analyses, though as different unconnected aquifers have been sampled in part, this is does not necessarily preclude ³⁶Cl decay.

Turner et al. 1992 found ³⁶Cl/Cl ratios in shallow groundwater with variable chlorinity in some inland catchments in NSW, Australia to all be of a similar magnitude and hence not be suitable for ³⁶Cl decay dating. The magnitude of ³⁶Cl/Cl ratios was noted to be higher than the expected atmospheric ratios for this latitude; however, subsequent upwards revision of expected atmospheric ³⁶Cl/Cl ratios in Australia (Keywood, 1995, 1998) may now mean most of these data are closer to expected atmospheric ratios. Nevertheless ³⁶Cl decay dating of groundwater in these small catchments is unlikely to be feasible as groundwater is thought move very rapidly (Shekarforoush, 2000).

Thorpe and Davidson (1990) measured ³⁶Cl/Cl ratios from confined aquifers in the Perth Basin, Western Australia and found that the ³⁶Cl/Cl ratios were very low (< 10 by 10^{-15}). As these values could approximate both the input ³⁶Cl/Cl atmospheric ratio at this coastal setting and expected in situ ³⁶Cl production from sandstones, no ³⁶Cl decay dating of groundwater was possible.

Jacobson et al. (1991) ran a few samples for ³⁶Cl determination from various depths in a couple of exploratory bores drilled at Lake George near Canberra, Australia. They concluded no ³⁶Cl decay ages were determinable as mixing and dilution rather than ³⁶Cl decay were the dominant processes evident from ³⁶Cl/Cl against Cl⁻ and ³⁶Cl against Cl plots. Interpretation was further complicated as the sample with the lowest ³⁶Cl/Cl ratio, at 100m in bore 353, had a significant tritium value compared to other samples.

While some of these studies may benefit from additional data and further examination of local hydrology, they illustrate that the ³⁶Cl method is certainly not suitable in all locations.

8.2.4. Discussion

Table 8.1 shows a summary of the groundwater ³⁶Cl studies discussed in the preceding sections. There are, comparatively, a large proportion of Australian studies listed in this table. This is probably both a reflection of ready access to some unpublished data sets from within Australia and the keen interest in potentially old groundwater resources in this arid continent. Apart from the Australian studies the other studies cited in this table are readily obtained from publications. Some of these international studies are classified as being occurrences where the application of ³⁶Cl decay dating was considered unsuitable for a variety of reasons. These studies have not been discussed in the text due to brevity considerations but are shown in Table 8.1 to complete the review of ³⁶Cl groundwater literature.

Figure 8.1 shows a plot of ³⁶Cl/Cl versus chloride content for the studies that have been identified as attempting ³⁶Cl decay dating of groundwater. On comparison of this Figure with the process chart shown in Figure 7.2 it can be seen that apart from the data of Penny et al. (2003) no data set shows a clear ³⁶Cl decay trend. When these data are viewed with Cl concentration plotted on a log axis, as shown in Figure 8.2, it can be seen that the low chloride aquifer from Alabama (Penny et al., 2003) also does not show a ³⁶Cl decay trend.

For the purpose of comparison and to give a regional perspective for this study Figure 8.3 shows the results of all the Australian ³⁶Cl groundwater studies cited above. On comparing this figure with the process chart shown in Figure 7.2 it can be seen that no Australian study cited shows a clear ³⁶Cl decay trend. Many factors such as the addition of chloride, dissolution of vadose chlorides and variations in Cl concentration due to different evapotranspiration effects have been invoked to estimate a ³⁶Cl decay age in some of these instances.

Reference	Location	Study	Lithology	Clear ³⁶ Cl	comments
				decay trend ?	
³⁶ Cl decay age estim	ated				
Cresswell et al., 1999a	Australia	Westen Water	seds/ volcanic	ou	regional study, samples not necessarilly hydraulically connected.
Cresswell et al., 1999b	Australia	Deep Amadeus	seds	ou	deep aquifers sampled via gas well drilling, sample integrity not ideal?
Mazor, 1992	Australia	Murray Basin	seds	ou	Re-Interpretation of Davie et al., 1989 data
Love et al., 2000	Australia	GAB western	seds	ou	Part of Nth transect shows decay trend, see also ^{al} Kr by Lehmann et al., 2003.
Bentley et al., 1986.	Australia	GAB	Seds	ou	hydraulic indication of old groundwater, ³⁶ CI/CI approaches secular equilibrium
Torgersen et al., 1991	Australia	GAB	seds	ou	as above
Radke et al., 2002	Australia	GAB	seds	ou	Compilation of all chemical and isopic data from GAB
BRS, 2002	Australia	Radwaste, S.A.	seds	ou	mixed lithology, clays sands
Sturchio et al., 2004	Egypt	Nubian Aquifer	seds	ou	Deep confined aquifer, old water ^{sit} kr used to help calibrate age
Cresswell et al., 2001	Nepal	Kathmandu Basin	seds	ou	Sand and clay interbeds with some lignite, very low CI in part
Dowgaiallo et al., 1990.	Poland	Mazowsze Basin	seds	ou	multi layered lithology, CI origins potentially complex
Phillips et al., 1986b	Canada	Milk River	seds	ou	Large basin, hydraulic data indicate old water
Nolte et al., 1991	Canada	Milk River	seds	ou	as above
Zhou et al., 2001	China	Hebei Plain	seds	ou	Limited sample size, preliminary results?
Penny et al., 2003	NSA	Alabama	seds	ou	Mixed lithology Basin, subsurface halite known in region, low CI content
Kaufman et al., 1990	Africa	Magadi- Natron	seds/volcanic	ou	Infilled basaltic graben, large variation in CI content
no ³⁶ Cl decay age co	nsidered feas	ible			
Davie et al., 1989	Australia	Murray Basin	seds	ou	subsurface CI input identified, ³⁶ CI data elucidates recharge, contrast Mazor, 1992
Kellett et al., 1990	Australia	Murray Basin	seds	ou	as above
Commander et al., 1994	Australia	Goldfields WA	seds	ou	Mixed sed lithology of paleo channels, salt lakes hydraulically connectec
Turner et al., 1992	Australia	NSW Murray Basin	seds	ou	Little variation in ³⁶ Cl ratio
Turner et al., 1995	Australia	Goldfields WA	seds	ou	as above
Hostetler, 2002	Australia	Amadues	seds	ou	regional study, samples not necessarilly hydraulically connected.
Jacobson et al., 1991a	Australia	Lake George	seds	ou	Mixed lithology, CI decrease with depth, playa brine mixing
Andrews et al., 1994	England	Midlands Triassic	seds	ou	Subsurface mixing of CI identified
Purdy et al., 1996	USA	Aquila	seds	ou	36 Cl increases along flow line (Bomb 36 Cl ?)
Moran and Rose, 2003	NSA	Nevada	seds	ou	Subsurface input of CI identified
Le Gal la Salle et al., 1995	Niger	Illumenden Basin	seds	ou	Data plots indicate near secular equilibrium, subsurface CI input also identified

Table 8.1 $\,^{36}$ Cl groundwater studies discussed in the text

Page 1 of 2

Reference	Location	Study	Lithology	Clear ³⁶ Cl	comments
				decay trend ?	
Fontes et al., 1984 Yecielli et al., 1996 Balderer et al., 1997	Algeria Israel Hungary	Sahara Dead Sea great hungarian plain	seds seds seds	on on on	likely hypogene production and subsurface CI input identified Mixing of rain water and ancient playa brines indicated as source of CI in Dead Sea Input of subsurface CI identified
Examples of epigene p	production (effect			
Andrews et al., 1986	Sweden	Stripa	hard		Determined nuetron flux of granite, ³⁶ Cl data used to trace Cl origins
Aquilina et al., 2002	France	Thau lagoon	hard/ carbonate	ou	Ingrowth of hypogene ³⁶ Cl used to estimate age
Balderer and Synal, 1996	Turkey	Antolia	seds & hard	ou	hypogene ³⁶ Cl used as tracer to study origin and mixing of groundwater
Gascoyne et al., 1994	Canada	Manitoba	hard	ou	Hypogene production precludes decay dating, ³⁶ Cl useful as tracer
Thorpe and Davidson, 1990	Australia	Perth Basin	seds	ou	³⁶ CI/CI ratios close to both atmospheric and secular equilbrium
Mahara et al., 2001	Sweden	Aspo	hard	ou	no 36 Cl decay age possible due to hypogene affect
Lehmann et al., 1990	Switzerland	Nagra	hard	ou	no 36 Cl decay age possible due to hypogene affect
Rao et al., 1996	NSA	Vallero? Caldera	hard/ carbonate	ou	Indication of age from hypogene production, ³⁶ Cl useful tracer to study Cl origins
Fehn et al., 1994	USA & German	ly Rochester program	seds & hard	no	Ingrowth of hypogene ³⁶ Cl used to estimate age
Lippmann et al., 2003	South Africa	Witwatersand	seds	DO	Ingrowth of hypogene ³⁶ Cl used to estimate age
Examples of Bomb ³⁶ Cl	as tracer				
Herczeg et al., 1997 Cook et al., 1994	Australia Australia	Murray Basin/ Gambier L/s Murray Basin/	carbonate seds	n/a n/a	Recharge study, high ³⁶ CI/CI taken to indicate bomb ³⁶ CI affect recharge study
Murad and Krishnamurthy, 2004	UAE	North Oman Moutains	hard & seds	n/a	Bomb ³⁶ cl used as tracer, mixing of groundwater identified
Cambell et al., 2003	USA	Yucca Mountain	volcanics	n/a	Tracer study, deep penetration (>200m) of modern rain indicated in vadose zone
Liu et al., 1995	USA	Arizona	soil	n/a	recharge study
Scanlon et al., 1990	USA	Техаѕ	seds	n/a	recharge study
Norris et al., 1987	USA	Yucca mountains	soil	n/a	recharge study
Larsen et al., 2003	Zimbabwe	Zambesi	seds	n/a	recharge study, inconclusive

Table 8.1 $\,^{36}$ Cl groundwater studies discussed in the text







been plotted, and b) the Bureau of Rural Science (2002) data has been estimated from graphical plots only and may therefore be slightly Figure 8-2 The same data set as Fig. 8.1 with ³⁶CI/CI versus the log of CI concentration. On comparison with Figure 7.2a it can be seen that no clear decay trends are apparent. Note a) for the sake of clarity only the NT portion of the GAB (Radke et al., 2000) data set has inaccurate c) The process trends will be slightly different because of the logarithmic CI⁻ concentration scale.





However, from the data shown in Figure 8.3 and discussion presented above regarding interpretation of ³⁶Cl decay data (e.g. considerations of Andrew and Fontes, 1992) the lack of a clear decay trend in these studies renders any such age estimate ambiguous.

8.2.5. Conclusion

Previous attempts at ³⁶Cl decay dating of groundwater have been controversial. Uncertainties on the provenance of subsurface chlorides often manifested as an observed inverse relationship between ³⁶Cl/Cl ratio and chloride concentration along flow path has been the chief concern. There seems to be two schools of thought – those that think ³⁶Cl decay dating is still possible even when well over 50% of Cl is derived from subsurface sources post recharge and those that think indications of subsurface chloride additions preclude the use of the ³⁶Cl decay dating method.

In nearly every instance where ³⁶Cl decay dating of groundwater has been attempted chloride to ³⁶Cl/Cl relationships are such that no clear decay trend is evident. In some instances supporting hydrologic evidence such as decrease in ³⁶Cl content along flow paths perceived to have very slow flow rates exists, however, there is a school of thought that doubt that a ³⁶Cl decay age has been proved. In short, where ³⁶Cl decay dating has been attempted to date, results are ambiguous.

8.3. Comparison to other studies

The work at Palm Valley, central Australia, outlined above and summarised by Wischusen et al. (2004), presents a suite of ³⁶Cl data showing a distinct decay trend (Figs 7.5 & 7.6). No subsurface additions of ³⁶Cl or Cl are apparent and results are consistent with a viable flow model (Fig 9.2). When these results are compared to other ³⁶Cl decay studies, as shown on Figure 8.4, it can be seen that the Palm Valley results stand alone in displaying a clear decay trend. This pattern is also observed when results are compared to data from all other previous Australian studies Figure 8.5.



Figure 8-4 ³⁶Cl/Cl versus the log of Cl concentration for the studies that have been identified as attempting ³⁶Cl decay dating of groundwater (i.e. Fig 8.2) compared to the results of this study at Palm Valley On comparison with Fig. 7.2a it is evident that the results from Palm Valley are alone in displaying a clear (vertical) decay trend. Note a) for the sake of clarity only the NT portion of the GAB (Radke et al., 2000) data set has been plotted, and b) the Bureau of Rural Science (2002) data has been estimated from graphical plots only and may therefore be slightly inaccurate.



compared with the results from Palm Valley. Again on comparison with Figure 7.2a it is evident that the results from Palm Valley are Figure 8-5 ³⁶Cl/Cl versus the log of CI concentration for all of the Australian ³⁶Cl groundwater studies cited in the text (i.e. Fig 8.3) alone in displaying a clear (vertical) decay trend.

While it may be that further work or alternative hypotheses are developed to refute this interpretation, it is worth considering why this study, seemingly rare among ³⁶Cl decay studies, allows an unambiguous interpretation of ³⁶Cl decay.

8.3.1. Discussion

Most ³⁶Cl decay studies (Table 8.1) have, naturally enough, been orientated to aquifers with potential for old groundwater. This has meant that large confined sedimentary aquifers have usually been sampled (e.g. GAB, Milk River). Given bores in such environments are deep and expensive to drill, existing water supply bores comprise most of the sample points. These bores are therefore constructed to maximise yield rather than provide a sample window to discrete aquifer horizons.

Thus in many instances sample integrity of ³⁶Cl samples cannot be assured as doubts exist as to which formation(s) have provided chloride to the sample. For example, bore screens may be set so that several litholgies are intersected and/or in the case of gradational aquifer – aquitard contacts found in many basins, inadvertently set adjacent intra aquitard sand lenses rather than the main aquifer. If only some samples within a suite of analyses contain a mix of water and/or chloride derived from two sources with different ³⁶Cl/Cl ratios due to age of chloride or hypogene production rates, the interpretation of results can be difficult or at least ambiguous. This is because a plot of data points on any ³⁶Cl/Cl against Cl or ³⁶Cl diagram would then be expected to show a scatter rather than any clear trend such as decay or solely addition of dead chloride.

Furthermore, the intercalation of aquitards and aquifer horizons in many sedimentary settings means natural leakage of chlorides from aquitards into aquifers can be ongoing along flow paths, which adds further potential complication to interpreting ³⁶Cl results. Fabryka-Martin et al. (1991) note that even intra aquifer low permeability horizons are probably the source of the increasing halide concentrations observed along flow paths in the Milk River Aquifer that complicate the interpretation of ³⁶Cl results.

Another potential complication is the fact that many basins previously sampled are stressed due to groundwater extraction. This may then result in the equilibrium that existed for many years under natural conditions being upset. Such a change of conditions may alter rates of leakage of chlorides from aquitards at various parts of an aquifer system, which could also complicate interpretation of results.

Another scenario, which may also cause problems, is when chloride concentrations vary markedly due to differing concentration by evapotranspiration effects at the time of recharge. For instance, an older more evaporated parcel (low ³⁶Cl/Cl, high Cl) of water further along the aquifer flow path when plotted with modern less evaporated water parcel nearer the start of the flow path (higher ³⁶Cl/Cl, lower Cl) may give the impression of addition of dead chloride when plotted on a ³⁶Cl/Cl versus Cl process chart. Such considerations are why tracers indicative of chloride provenance such as Cl/Br ratios are recommended (e.g. Davis et al., 1998a) accompaniments to ³⁶Cl decay studies.

Recharge mechanisms may also be a relevant factor. In some instances, such as large arid inland sedimentary basins where direct recharge rates are low, it may be that very old chloride held in a thick vadose zone slowly (and erratically?) mixes along groundwater flow paths mainly derived from indirect recharge along basin margins or streams. The potential for such an effect to complicate the determination of subsurface chloride and stable isotope origin is developed in detail by Wischusen (1998) when discussing groundwater chemistry of the Cainozoic Basins a few hundred kilometres north of Palm Valley. The hazard of vadose zone salts contaminating groundwater systems is demonstrated by a study in the semi arid Murray Basin of south eastern Australia by Leaney et al. (2003). They note that increased mobilization of vadose salts due to enhanced recharge rates following land clearing may result in the salinization of important underlying fresh groundwater resources. Direct recharge rates are thought to be very low in some inland Australian Basins. For example Kruseman (1997), in a discussion on the Cainozoic Western Murray Basin in South Australia, notes the water flux of very saline water (chloride 10000 to 20000 mg/L), below 2m depth under eucalypt vegetation and above the 30m deep watertable, to be between 0.04 and 0.06 mm/yr (about 1m in 20000 years). So in some regions it is feasible that very old (100+ ka) chlorides slowly leak into much younger, saturated groundwater flow paths. Thus old stored vadose zone chlorides derived from a slow direct recharge mechanism may also have the potential to complicate the interpretation of ³⁶Cl decay results in some

instances. While this consideration may also be a factor in considering ³⁶Cl results for the Hermannsburg Sandstone, the stable isotope data (Section 6.1.2) that plot along the LMWL indicate little evidence of an evaporation effect. Hence, assuming some evaporation effect would be discernible from stable isotope data plots if a direct recharge mechanism contributed significantly to groundwater composition, there is no evidence that a direct (vadose zone) recharge mechanism contributes chlorides to the Hermannsburg Sandstone. However, it is noted that apportioning the contribution of various recharge mechanisms to an aquifer is a very difficult task (e.g. Lerner et al., 1990). Hence measuring the effect of differing recharge mechanisms on ³⁶Cl/Cl ratios in an aquifer is also likely to be a difficult task.

All samples at Palm Valley are taken from the Hermannsburg Sandstone, an extensive, thick (> 900m) outcrop of Devonian litharenite. This unit is described as a monotonous sequence of fine to very coarse grained sandstones with only very rare sandy siltstone interbeds (Jones, 1972). The high energy deposition environment (in response to nearby orogenic event) of this sandstone means apart from occasional mud clasts that virtually no aquitard material is associated with this unit. As all samples at Palm Valley are taken from areas of outcrop, the nearest significant aquitard occurs in the underlying Parke Siltsone formation, many hundreds (>500) of metres below sampling points. Thus, in effect, samples from this study are sourced completely from only one lithological unit, mixing of waters or chlorides derived from different lithologies are unlikely. Consequently, integrity of ³⁶Cl samples is assured in this instance.

The demonstration of a distinct suite of ³⁶Cl decay data from such a setting suggests that to be given a chance to assess ³⁶Cl decay, future studies should strive to sample water derived from only one common lithological unit. In many previous study sites however, ensuring such integrity of samples may prove very expensive and logistically challenging (e.g. electric logging, setting packers etc).

8.3.2. Conclusion

Determination of ³⁶Cl/Cl ratios has been likened to identifying one particular sugar grain from a room full of sugar. Consequently, it stands to reason that ³⁶Cl decay dating of groundwater is likely to be unusually sensitive to contamination by outside aquifer sources of chloride. Consequently, all efforts to minimise potential ambiguity on the

provenance of chlorides including maintaining sample integrity whenever possible seem justified. While such measures cannot necessarily counter unfavourable conditions such as distribution of lithologies, disruption of hydraulic equilibrium due to extraction or additional ³⁶Cl production factors, the successful application of the ³⁶Cl decay method at Palm Valley shows that ³⁶Cl decay dating can be a viable tool for studying groundwater systems.

9. Discussion

9.1. Flow

9.1.1. Flow threshold?

The isotope and hydraulic indicators of old groundwater in aquifers raises the question of whether groundwater actually flows or whether groundwater age is a result of zero flow conditions. For instance there is some speculation that there may be a threshold gradient in some media below which groundwater does not flow (e.g. Neuzil, 1986; Bernadiner and Protopapas, 1994). Similarly, it is also noted that groundwater gradients in some instances may be the result of coupled flow (an osmosis and ultrafiltration effect) or even geologic factors such as subsidence. Under such conditions non-Darcian behaviour (i.e. no direct proportionality between flow and groundwater gradient) might be expected (Neuzil, 1986).

While feasible mechanisms to cause such effects can be speculated upon, there is little or no experimental evidence to prove them. This is because the theoretical conditions for such phenomena (coupled flow etc) cannot be readily reproduced or measured in a laboratory due to the very low hydraulic conductivities and gradients involved. Neuzil (1986) reports experiments that have recorded Darcian flow behaviour at hydraulic conductivities lower than 10^{-10} m/day, though this requires high artificially created gradients (around 10^2). In the 1930's a series of experiments conducted by the United States Geological Survey proved Darcian flow valid in porous media at gradients as low as 10^{-5} (Meinzer and Fishel, 1934; Fishel, 1935).

The simultaneous testing of flow at very low gradients and very low hydraulic conductivities remains problematical due to the prohibitive lengths of time involved. However, as the hydraulic conductivity and gradient measured at Palm Valley (i.e. $k > 10^{-5}$ m/day, gradient > 10⁻⁴, Section 4.3 and Table 4.1) are well within the normal range expected for Darcy flow behaviour, it is reasonable to assume groundwater is indeed moving through the Hermannsburg Sandstone.

9.1.2. Diffusion from flow in fractured aquifers

Burton et al. (2002) note that using environmental tracers to date groundwater in fractured rock aquifers may be affected by dispersion and diffusion effects which can result in apparent tracer ages older than actual hydraulic age. In a detailed study using modern multi tracers (e.g. chlorinated fluorocarbons) and nested piezometers they note such effects are difficult to quantify and, in their study of the effects of aquifer anisotropy on groundwater ages in a fractured sedimentary aquifer in Pennsylvania, note while diffusion may effect computed ages by an unknown amount, the pattern of relative ages should not be altered.

Even though the Hermannsburg Sandstone is in part a fractured rock aquifer, effects of diffusion on apparent isotope age are not considered for this study. This is because, as detailed in Chapter 4, this aquifer is considered a granular porous medium of significant porosity and permeability that is only occasionally fractured. While some discrepancy between matrix and fracture groundwater residence times may exist in some instances, the use of the continuum approach for studying regional groundwater flow caters for this by conceptualising fractures as repeating large pores within the system (Fetter, 1993). Thus, radiocarbon data are not considered to have been affected by this phenomenon. Furthermore, any diffusion of Cl⁻ from fractures to the matrix is only likely to affect Cl⁻ concentration but not alter groundwater ³⁶Cl/Cl ratios. Hence indicated ³⁶Cl decay ages are unlikely to be affected by diffusion from flow through fractured media.

9.1.3. Flow model

9.1.3.1. Introduction

The data presented above shows the groundwater isotope data at Palm Valley to be discordant. That is, if groundwater simply moves from west to east down gradient, as shown in Figures 4.7 and 4.9, then an increase in groundwater age to the east should be

matched by decreasing activity of both ¹⁴C and ³⁶Cl down gradient. For instance, bore RN 12024, which is 5 km away and down gradient from bore RN 4805, is more radioactive than the up gradient bore. Similarly, Road Spring has more ¹⁴C but a lower ³⁶Cl/Cl ratio than RN 12024 further down gradient. It should be noted, that the variation in cation chemistry (Chapter 5) and the fact that many of the sites sampled are discharge zones, discounts the likelihood that these discordant data can be explained by simple mixing of old groundwater with modern rainfall. Given these considerations, a hydraulic model that accounts for the distribution of isotopically discordant groundwater along a seemingly continuous west to east regional flow system (Brown et al. 1990; see also Fig. 4.7) needs to be developed.

In the past such discordant isotope data in the same basin has been taken as evidence for compartmentalisation of groundwater and support for a lack of hydraulic continuity in some basins (Mazor, 1992). Toth (1995), however, has long argued that hydraulic continuity should exist even if low permeability barriers are present. In developing his argument that hydraulic continuity is expected even in large sedimentary basins, Toth (1995) describes groundwater flow within a unit basin model.

9.1.3.2. Toth unit basin model

In a unit basin three distinctly different groundwater flow regimes are noted - recharge, midline and discharge regions characterised by descending, lateral and ascending flow respectively. However, when significant topographical variations exist composite flow patterns are superimposed on the unit basin flow pattern (Toth, 1995).

An example of the composite flow pattern arising from topographical affects on a simple homogenous aquifer as constructed by Toth (1995 - also discussed by Freeze and Cherry, 1979) is shown in Figure 9.1 where the superposition of different scale (order) flow systems can be clearly seen. Given the topographic expression of the outcropping Hermannsburg Sandstone aquifer seen at Palm Valley (see Figs. 3.9 & 9.3) and in fact along the whole region of outcrop between Areyonga and Injarrtnama (Fig. 4.7), it can be reasonably assumed that groundwater flow patterns as least as complicated as that shown in Figure 9.1 exist within the Hermannsburg Sandstone if Toth's model holds true.



Figure 9-1 Composite Flow Patterns, Unit Basin Flow system in areas with significant topography. Schematic of composite flow patterns expected in a Unit Basin flow system in areas with significant local topography. The composite patterns are characterised by laterally alternating recharge-discharge regions where different domain flow systems may discharge in close proximity as shown in the area highlighted. After Toth, (1963, 1990, 1995). As the Hermannsburg Sandstone is probably anisotropic (Brown et al. 1990) with hydraulic conductivity greater parallel to bedding, it is probable that flow patterns are more complex than shown in Figure 9.1. Indeed Toth (1995) and Ophori and Toth (1990) show variable flow length patterns calculated for heterogenous aquifers with varying topography to be similar in style but more complex when compared to the homogenous aquifer model. While this likelihood of more complex flow paths in the Hermannsburg Sandstone is noted, actual distribution of aquifer heterogeneity within the Hermannsburg Sandstone is unknown, thus the simpler homogenous aquifer model is used for illustrating the potential complexity of flow patterns at Palm Valley.

Such an approach is considered valid by Domenico and Schwartz (1998, p83) when discussing regional flow in basins; "Although the ideal homogenous case is contrary to field observations, it is merely a condition in the mathematical development, not an assumption necessary to the general validity of the theory. Further, the steady state solutions do not deny the existence of a fluctuating upper boundary of flow".

While the validity of Toth's flow model is still debated (e.g. Mazor, 1996; Laing, 1996), the flow patterns outlined, even in a homogenous aquifer, appear complicated enough to explain how seemingly discordant isotope ages could be found nearby in the same aquifer. For example the two bores RN 14616 and RN 4805, which have ¹⁴C contents (Table 6.1) approaching levels unsuitable for dating (indicated age > 25 ka, Andrews and Fontes, 1993), have significantly different ³⁶Cl contents (Table 7.1). As these bores have much the same: depth of sample point, stable isotope composition, ¹⁴C content and water type (bicarbonate waters with slightly different $^{36}Cl/Cl$ ratios, e.g. 1 and 1.3 respectively) it is readily conceivable that the different ³⁶Cl/Cl ratios observed at each bore are due to variations in groundwater flow path length.

Thus ³⁶Cl (groundwater age) variance may be explained as the sampling of either different flow cells or different flow paths at Palm Valley.

The regional topography shows the Palm Valley area (see Figs. 3.9 & 4.7 and Plate 3.3) to be part of the Palm Plain, a low-lying flat area surrounded by high hills. On a regional scale this suggests the whole Palm Plain has the potential to have areas where groundwater flow lines of various length may discharge in close proximity, as

illustrated in Toth's homogenous model, Figure 9.1. This scenario and relation of topography to the Palm Valley area is shown in Figure 9.3.

9.1.3.3. Conceptual model

Figure 9.2 shows a conceptual model based on Toth's work on groundwater flow in an isotropic aquifer with topographic effect at Palm Valley. It can be seen that even assuming a simple isotropic aquifer a flow pattern complicated enough to explain the seemingly discordant isotope data at Palm Valley may still result. Data from Palm Valley gas well number 3 shows water level within the Hermannsburg Sandstone at the apex of the Krichauff Range (see Fig. 9.3) at around 627m. This is in keeping with the conceptual model. If analogous flow systems are assumed to be equally likely at the other sites sampled along the large outcrop of Hermannsburg Sandstone (i.e. Hermannsburg, Areyonga, and Injarrtnama, Fig. 4.7), then the similar isotope and chemistry data observed at these sites (Tables 5.1 and 7.1) may be explained in the same manner.

In Section 4.3.5 an analytical model of flow in the Hermannsburg Sandstone is presented. A range of feasible average linear velocities, based on drilling data and pump tests, allow analytical modelling. For instance, if a low k value is adopted as being representative of the Hermannsburg Sandstone, as evidenced by some of the testing near Palm Valley (i.e. k = around 0.0025 m/day, see RN14161 and 4508 Table 4.1) a water age of 300 ka would (using Eq. 4.4 and the hydraulic properties presented in Section 4.3.5) represent travel through 9 km of the Hermannsburg Sandstone occurred before reaching Palm Valley. Alternatively, if an average linear velocity an order of magnitude larger (0.025 m/day i.e. around double the mean k value of 0.06 m/day calculated from all results shown in Table 4.1) is assumed, then a water age of 300 ka would represent travel through 90 km of the Hermannsburg Sandstone. It is entirely feasible given the extensive outcrop of Hermannsburg Sandstone (Fig. 4.7) that water could have recharged at a point 9 to 90 km away and is now discharging (or in a shallow bore) at Palm Valley. Any future numerical model that was developed for this area would also need to be based on these data. As there are no bore hydrographs at Palm Valley future numerical modelling would probably be a steady state, fairly broad

scale approximation and not necessarily much of an improvement on the existing modelling of Brown et al 1991.

The groundwater ages calculated from ³⁶Cl decay data are credible when compared to the hydraulic analytical model. The flow pattern conceptual model is based on the Toth model which was developed purely on mathematically derived hydraulic principles. Consequently it is considered that the hydrogeological conceptual model developed here is adequately tested against both mathematical and hydraulic principles.

9.2. Sustainability

9.2.1. Introduction

Assessment of groundwater resources often involves determinations on how best to manage an extraction program so that perpetuation of a useful resource is maintained with no undue consequences for the environment. Such assessments usually require that a sustainable or safe yield of an aquifer be calculated. The concept of sustainable or safe yield can be difficult to define in many instances and has long been the subject of debate (e.g. Thies, 1940; Schoeller, 1959; Todd, 1959; Bredehoeft, 1997; Alley et al., 1999). At Palm Valley the interest centres on the likelihood that natural groundwater flow in the Hermannsburg Sandstone can sustain the flora refuge.



SCHEMATIC CONCEPTUAL FLOW MODEL FOR THE PALM VALLEY AREA

Figure 9-2 Schematic conceptual flow model for the Palm Valley area. Different domain (order) flow systems are shown to discharge at several locations and to have different orientations depending on topography. Localised flow cells are shown superimposed on the regional scale west to east flow direction shown in Figs 4.7 and 4.9. The potential for differing length flow paths and/or a mixture of different length flow paths discharging at some locations is illustrated.



Figure 9-3 Regional topography of the Palm Valley area with variable flow path lengths. The regional topography of the Palm Valley area and a schematic of how discharging groundwater can be expected to have variable flow path lengths at the topographically lower Palm Valley area. This diagram is in keeping with the work of Toth shown in Fig. 9.1 and assumes the Hermannsburg Sandstone to be a homogenous aquifer for the purpose of this illustration.

In arid areas, groundwater recharge is often irregular, and the natural groundwater flow of an aquifer may then vary with time. This implies that the determination of the likelihood and magnitude of recharge are important factors in assessing arid groundwater systems. Lerner et al. (1990) and Foster (1988) note that the determination of recharge is extremely difficult in arid areas.

There is no evidence of significant recharge (nor depletion) in modern times in the Hermannsburg Sandstone. Hydrograph records are, however, of relatively short duration (≈ 30 years). Thus, it may be that some recharge occurs at some locations under the current climatic regime. For example rare super floods associated with extreme southerly transgressions of the northern monsoon, thought to have occurred in the Holocene (e.g. Pickup et al., 1988; Bourke, 1998), may cause some recharge. The short duration of observed occasional modern day incursions of the northern monsoon (usually days) limits the likely significance of recharge from these events.

9.2.2. Palaeo-climate considerations

Jacobson et al. (1994) discuss the likelihood of Holocene and late Pleistocene recharge in the Amadeus Basin. Kershaw and Nanson (1993), Croke et al. (1996), Magee et al. (1995) and Nanson et al. (1992) present evidence for periods of thousands of years (interglacial pluvial) within the Quaternary glacial cycles when climate in central Australia may have been significantly wetter, particularly around 100 ka. These workers note some evidence for renewed wet conditions around 50 -35 ka may exist, however, this event may not have been as significant in central Australia as in south eastern Australia.

Nanson et al. (1998) note there is also some evidence for a possible wet period just prior to the last glacial maximum (22-26 ka) which may correlate with the evidence of ephemeral fluvial deposition also noted at about this time by Croke et al., (1996). Johnson et al. (1999) however, present evidence that the influence of the Australian monsoon in central Australia (Lake Eyre region) has declined since around 45-65 ka, which may limit the likelihood of significant recharge since this time. Hesse et al. (2004) note that the palaeo-climatic reconstruction of the Australian arid zone is limited by many unconstrained variables that control climate proxies such as runoff, vegetation and aeolian processes. Given this uncertainty in reconstructing and accurately dating past climate events and notwithstanding the pitfalls of dating groundwater mentioned above, it is apparent more data are needed before definite correlation of groundwater ³⁶Cl decay ages and palaeo climate events can be made.

9.2.3. Discussion

If significant recharge to the Hermannsburg Sandstone is mainly restricted to periods of wetter climate such as at 50 or 100 ka, then groundwater ³⁶Cl/Cl ratios may be expected to be lower than rainfall ratios due to radioactive decay. If recharge last occurred such a long time ago then slow decay of hydraulic head would now be occurring. Given that the hydraulic modelling of Brown et al. (1990) derived groundwater travel times of several million years through the Amadeus Basin within the Hermannsburg Sandstone, it is likely, if significant increase in head during such recharge events is assumed, that these heads could take several hundreds of thousands of years to dissipate. Thus, groundwater features such as springs and seeps can be expected to have the potential to last without recharge through the arid periods associated with the Quaternary glacial cycles.

For instance based on Equation 7.2, groundwater at bore RN 4508 with a ³⁶Cl/Cl ratio of 130 by 10⁻¹⁵ may have been recharged at around 400 ka if the rainfall ratio of 325 by 10⁻¹⁵ recorded by Keywood (1995) at Ti Tree is taken as a likely input ratio. Even if the ³⁶Cl/Cl ratio measured from shallow groundwater at West Spring or bore RN 14952 is taken as a more reliable estimate of input ratio (i.e. Mazor, 1992), groundwater at RN 4805 is still calculated to have been recharged around 300 ka. Determination of the exact timing of recharge to this aquifer, given the complex flow paths, lack of data points and inherent uncertainty in estimating input ratio at time of recharge, is, however, speculative.

In other arid areas, such as Egypt, past Quaternary higher water table (pluvial) events have been determined by the uranium-series disequilibrium method dating of travertine spring deposits (e.g. Crombie et al., 1997). It is possible that older higher calcrete deposits at Palm Valley will yield similar information on the timing of past wetter
climates if dated. There are, however, some pitfalls in dating calcrete with this method: assumption of initial activities can be crucial (Sturchio and Binz, 1988) and, most importantly, calcrete must be assumed to have remained a closed system to U and Th since formation for an accurate age determination (e.g. Williams et al., 1993; Sturchio and Binz, 1988). Given the ready potential of calcrete to remobilise during repeated episodes of dissolution and precipitation (Arakel, 1991), processes that can lead to the cement of old calcrete being made of components of different ages (e.g. Geyh and Eitel, 1998), the timing of past pluvial events by applying uranium-series disequilibrium method may be no more, or perhaps even less, certain than the inferences drawn from ³⁶Cl decay dating of groundwater.

This range of groundwater ages and the complex flow path geometry discussed above indicate that the shallow groundwater supply thought to maintain the unique flora at Palm Valley is in part likely to be hundreds of thousands of years old. If Palm Valley has been a floral refuge during the arid phases of the Pleistocene glacial cycles, as speculated earlier, then fossil groundwater moving through the Hermannsburg Sandstone may be the key to the continued survival of the unique stand of remnant flora at Palm Valley.

Such a scenario, at least within the Hermannsburg Sandstone, provides a framework that offers indirect evidence for conclusions drawn by other studies in central Australia. The survival of remnant aquatic fauna noted by Davis (1995, 1997) in this region and evidence of human occupation 22 ka in central Australia - which would require permanent waters to be viable (Smith, 1987, 1989) - benefit from the identification of a flow system that is maintained even during hyper arid periods such as glacial maxima.

9.2.4. Conclusion

The computed age and proposed flow path mechanism of groundwaters within the Hermannsburg Sandstone indicate that systems dependent on natural groundwater flows such as Palm Valley should be sustainable. As flow may be maintained by slow decay of hydraulic head after interglacial pluvial recharge for considerable periods, modest developments such as the Hermannsburg Community water supply should also be sustainable.

10. Conclusion

10.1. Main conclusions

This study has examined and characterised the hydrogeology of Palm Valley and applied the ³⁶Cl decay dating method. The background and theory of this method have been reviewed and the results obtained assessed in context against other previous studies. The culmination of this work has allowed the development of a conceptual model that goes a long way towards outlining the link between groundwater and the presence of a flora refuge at Palm Valley. Like many other groundwater studies the complexity and scale of the physical and biological environments encountered by water as it moves through the aquifer system has prevented an exact understanding of all facets of the hydrologic cycle being gained. The data obtained and interpretations made do, however, provide a sound basis for any further work to build on. To demonstrate how the objectives of this study have been achieved or attempted a summary of pertinent conclusions developed throughout the thesis is presented below for each of the four objectives presented in Chapter 1.

Objective 1) Evaluate the hydrogeology, hydrochemistry and isotope chemistry of the aquifer underlying Palm Valley:

The biology (genetics and physiology) are not yet fully understood for the Livistona mariae though they do appear very closely related, if not the same species, to some tropical dwelling Livistona sp. palms in northern Australia. Exact proof of groundwater dependence is not yet available for this or any other Livistona species. Groundwater dependence is, however, inferred by many people due to the climate and limited distribution of these and many other desert dwelling palms elsewhere in the world.

Groundwater movement is very slow within the Hermannsburg Sandstone and hydraulically it can be conceptualised as a low permeability double porosity aquifer where water moves through both fractures and sandstone matrix. Groundwater is low salinity and bicarbonate rich. Slight variations in cation chemistry seem to indicate slight differences in reactions experienced along different flow paths. The stable isotopes of water data (¹⁸O and ²H) indicate recharge has been rapid and via an indirect mechanism. Radiocarbon results range form modern to beyond the limit of detection.

Plots of ³⁶Cl/Cl v. Cl and ³⁶Cl show a decay trend and no apparent gain (or loss) of chloride within the aquifer. The ³⁶Cl/Cl ratios vary from around known modern atmospheric values to ratios consistent with over 300 ka of radioactive decay.

Objective 2) review the background theory and previous applications of 36 Cl decay dating of groundwater systems:

There is much consideration needed before using ³⁶Cl results as a decay dating tool. There are some uncertainties in assigning input values of ³⁶Cl/Cl ratios, not least because natural fallout of ³⁶Cl to the earth is yet to be categorically defined. Shallow groundwater ³⁶Cl/Cl are sometimes used as a proxy for input ratio when atmospheric ³⁶Cl fallout data are absent. Evidence of anthropogenic radioactive contamination and deviation from expected atmospheric Cl/Br ratios are the most appropriate criteria to test shallow water sample integrity in sparsely populated arid regions. While changes in ³⁶Cl fallout have occurred in the past, the nature of the variation in the Earths magnetic field mean that is unlikely that fallout has ever been less than at present; hence it is unlikely that ³⁶Cl decay method would ever underestimate age because a lower than present atmospheric input ratio operated at the time of recharge.

The affects of epigene ³⁶Cl production can usually be ignored for groundwater studies as the rate of build up and release is slow compared to the flux of groundwater. Hypogene production is usually only small (<10 by 10-15) when compared to atmospheric ³⁶Cl/Cl ratios. Only when exceptionally high rock U and Th contents are suspected or measured groundwater ratios approach hypogene production levels is it an important consideration. There is some evidence that bomb produced ³⁶Cl is still being released into the environment via methyl chloride production from plants. Based on known global methyl halide budgets the significance of this effect seems slight though it remains possible in areas of heavy vegetation such as temperate forests, accordingly it is not a consideration in central Australia. Uncertainties on the provenance of subsurface chlorides, often manifested as an observed inverse relationship between ³⁶Cl/Cl ratio and chloride concentration along flow path, has complicated previous ³⁶Cl decay dating of groundwater studies. The main problem is that addition of subsurface chloride makes decay age determination ambiguous as un-testable assumptions on the rate and amount of chloride addition need to be made.

Objective 3) apply the ³⁶*Cl decay dating method at Palm Valley and critically assess results in the context of objectives 1) and 2):*

The clear ³⁶Cl decay trend and indication of great groundwater age obtained for this study is significant. The application of the ³⁶Cl decay dating technique has provided an unambiguous set of results, an unusual occurrence when compared to other studies. This suggests some lessons may be gained from this work. The review of previous ³⁶Cl decay studies show the crucial factor to unequivocally apply ³⁶Cl decay dating is to ensure that no addition of chloride occurs along groundwater flow paths after recharge. At Palm Valley, this condition is apparently met due the considerable thickness and monotonous lithology of the Hermannsburg Sandstone. In addition, the very low extraction to aquifer storage ratio which exist for this aquifer mean that near natural hydraulic equilibrium conditions exist. These results highlight the importance of ensuring that groundwater which has only flowed through the same lithological unit should be compared for explicit ³⁶Cl decay dating. The amount of aquifer stress may also be a factor to consider. Given these results it may be that other studies which have usually used existing water production bores drilled through sedimentary succession of differing lithology have suffered because of a mixed lithological setting. For instance, as noted by Love et al. (2000) in the GAB, possible poor bore construction and/or lack of certainty on lithology type and aquifer horizon sampled may lead to problems with sample integrity in confined aquifer settings. In addition, the juxtaposition of different lithology types such as clays and sandstones means that mixing of chlorides derived from aquifer, aquitards and intra aquifer clay/shale horizons can be expected. Given the likelihood of mixed provenance of chloride where age and hence also ³⁶Cl content is anticipated, it is little wonder that many studies have produced ³⁶Cl/Cl against Cl or ³⁶Cl plots that do not provide unambiguous decay trends.

Determination of ³⁶Cl/Cl ratios has been likened to identifying one particular sugar grain from a room full of sugar. Consequently, it stands to reason that ³⁶Cl decay dating of groundwater is likely to be unusually sensitive to contamination by outside aquifer sources of chloride. Consequently, all efforts to minimise potential ambiguity on the provenance of chlorides including maintaining sample integrity whenever possible seem justified. In many previous study sites however, ensuring such integrity of samples may prove very expensive and logistically challenging (e.g. electric logging, setting packers etc).

In an effort to limit these uncertainties some of the most recent applications of this method have tried to obtain complementary age data from other techniques such as ⁸¹Kr and ⁴He (e.g.; Lehmann et al., 2003; Sturchio et al., 2004). While these other techniques also have their own problems of application and interpretation, the continued employment and refinement of many different dating techniques in conjunction with ³⁶Cl sampling offers the chance to reduce the ambiguity of dating groundwater in future studies. It is worth noting that even this study relied upon ¹⁴C, stable C, H, and O isotopes, major-ion geochemistry, and hydraulic characteristics in combination with hydrologic modelling to support the conclusion that ³⁶Cl decay dating is feasible at Palm Valley. This point illustrates the maxim that as much data that it is possible to gather should be used in interpreting groundwater flow systems.

Objective 4) Determine the feasibility that groundwater maintains the flora refuge at Palm Valley:

The clear ³⁶Cl decay trend and indication of great groundwater age obtained for this study has also helped meet the objective of determining the feasibility that groundwater maintains the flora refuge at Palm Valley. The distribution of several vintages of groundwater discharging contemporaneously in close proximity allows a conceptual hydrogeologic model to be developed. This model, based on the standard mathematically derived Toth groundwater flow theory, shows these data are consistent with modern and mid to late Pleistocene groundwater discharging at Palm Valley. This model could represent recharge form several previous Quaternary pluvial events now discharges at Palm Valley though direct linkage to these climate drivers is yet to be proved. Whatever the nature of recharge, episodic or continuos, this conceptual model

demonstrates a flow system able to sustain dependent flora during previous Quaternary hyper arid phases. It is therefore feasible that groundwater maintains the palms at Palm Valley and that it has operated as a flora refuge since at least the mid Pleistocene.

10.2. Recommendations

Hydrogeology is a complex science, not only do groundwater flow and chemistry within an aquifer need to be characterised but ideally the mechanism, chemistry, spatial and temporal nature of all flows into and out of an aquifer also need to be measured for a complete understanding of a groundwater system. This study, like many other groundwater studies, suffers from the fact that not all these parameters have been determined. The task of dating groundwater is also a complex science – water of several vintages maybe mixed within an aquifer and different isotopes may indicate differing groundwater ages from the same sample (e.g. Bethke and Johnson, 2002). As this study concentrated on the radio-chlorine decay dating method I have included some recommendations on its application along with some specific points that should improve the understanding of the local hydrogeology and ecology.

The hydraulic understanding of Palm Valley is based on sparse data points and little or no monitoring data. While there is no pressing justification to disturb the National Park for installation of monitoring bores, the opportunity to do so during other future work projects such as shelter or road building should be considered. In fact as the Hermannsburg Sandstone is an important aquifer elsewhere in the central Australian region, a well funded designated study of the hydrogeology of the entire aquifer, would be of benefit for the management and allocation of water resources in the Northern Territory.

The link to groundwater and Palm survival is only circumstantial, irrefutable proof by way of studies linking tree physiology and groundwater use are desirable. The relict nature of the palms is also only circumstantial and yet to be proved. The work on genetic comparisons with near relatives should continue so that a quantitative estimate on the period of isolation for this palm can be made. Such an estimate may provide some corroboration of the conclusions drawn from the hydrogeologic conceptual model developed. These points indicate that a more comprehensive understanding of the Palm Valley flora refuge will only be achieved by a multi disciplinary study. It is therefore recommended that future studies of probable groundwater dependant ecosystems such as Palm Valley in central Australia enlist a range of scientific disciplines to work on a collaborative basis. A start point for further work at Palm Valley could be the comparison of deuterium levels in xylem water of *Livistona mariae*, sampled after a long period (months) without rain, with the deuterium values of groundwater in the underlying Hermannsburg Sandstone. This may allow, in a similar manner to that outlined in the study by Lamontagne et al. (2005) for other tree species in northern Australia, an appreciation of the degree of groundwater dependence (at least at the time of sampling) of the *Livistona mariae*.

The flows into and out of the Hermannsburg Sandstone and the climate events that drive the hydrologic cycle are largely unknown or proven. Further work is needed to create a better knowledge of the water balances of this aquifer. Such information may allow the development of a functioning numerical model to aid the understanding and management of this aquifer. It is worth noting, however, that even after much study and expenditure some major exploited groundwater resources in arid central Australia such as the Alice Springs town supply and the Great Artesian Basin are yet to have water fluxes characterised to the extent that fully calibrated dynamic numerical models can be constructed. This suggests that the hydrogeology of large aquifers such as the Hermannsburg Sandstone in the arid zone is complex and therefore requires much further effort to be fully understood.

The atmospheric ³⁶Cl/Cl ratios for this area have only been measured over a two-year period in early 1990s (Keywood, 1995). Knowledge of this crucial factor in the dating of groundwater would benefit from a long-term collection programme so that the relationship of ³⁶Cl/Cl ratio to the full suite of possible climatic conditions could be examined. For example the ³⁶Cl/Cl ratio of precipitation during the anomalous `big wet' years of 1974 and 2000 in central Australia would be of interest as recharge may be significant for some aquifers only during these events.

Rainfall and dry fallout chemistry data are also meagre in central Australia; most future groundwater studies would benefit from a systematic collection of these data.

The chemical and isotopic data that forms the basis of this study is largely obtained form only one round of sampling. A better understanding of the system maybe gained if repeat sampling of groundwater along with some data from rain and dry fallout were obtained.

It is worth analysing the Hermannsburg Sandstone to obtain some solid phase chemical data as an aid to any future geochemical studies such as reactive transport modelling using the PHREEQC (Parkhurst and Appelo, 1999) computer code.

The past hydrologic regimes that have existed at Palm Valley should be further examined – consideration of dating aquifer precipitate deposits such as the calcrete lenses noted in the area above the water table should form a part of any further groundwater studies at Palm Valley.

While the ideal of adequately accounting for all sources and sinks of chloride has long been recognised for successful ³⁶Cl decay dating, it has rarely been achieved. This may in part be due to the expense and logistical difficulties associated with maintaining sample integrity in large deep sedimentary basins. This study highlights that unambiguous ³⁶Cl decay results do occur in some natural systems; hence future studies should plan and budget for much effort in ensuring that only one lithological unit is sampled. In many instances, however, this may not feasible. Better control on mixing end members by way of sampling solid phases and aquitard pore waters for halide signatures and ³⁶Cl/Cl ratios along with aquifer waters is recommended to add credence to interpretations of groundwater ages necessarily based on allowing for mixing of dead chlorides along flow paths etc. Where possible the use of complementary techniques such as ⁸¹Kr and ⁴He should also be considered.

11. References

Acuna, J.A. and Yortos, Y.C.,1995. Applications of fractal geometry to the study of networks of fractures and their pressure transient. Water Resources Research, 31, 527-540.

Alekin, O.A., 1970. Fundamentals of hydrochemistry. Hydrometeoizdatelstvo. Leningrad, 444 p.

Alley, W. M., Reilly, T.E. and Franke, O. L., 1999. Sustainability of groundwater resources. United States Geological Survey, Circular 1186, 79p.

Allison, G.B., 1981. The use of natural isotopes for measurement of groundwater recharge. Proceedings of the Groundwater Recharge Conference, Townsville, 1980, Australian Water Resources Council Conference Series No 3, Australian Government Publishing Service, Canberra, 203-214.

Allison, G.B., 1982, The relationship between ¹⁸O and deuterium in water in sand columns undergoing evaporation: Journal of Hydrology, v. 55, p. 163-169.

Allison, G.B., 1988. A review of some of the physical, chemical and isotopic techniques available for estimating groundwater recharge. *In* Simmers I. (ed), Estimation of Natural Groundwater Recharge, North Atlantic Treaty Organisation, Advanced Science Institutes series C, Volume 222, 510 pp. (Proceedings of the NATO advanced research workshop, Antalya, Turkey, March 1987) D. Reidel Publishing Company Dordrecht

Ambroggi, R.P., 1977. Underground Reservoirs to control the water cycle. Scientific American, Vol. 236, (5), 21-27.

Andrews, J.N., Fontes, J.-C., Michelot, J.-L. and Elmore, D., 1986. In-situ neutron flux, ³⁶Cl production and groundwater evolution in crystalline rocks at Stripa, Sweden. Earth and Planetary Science Letters, 77, 49-58.

Andrews, J.N., Florkowski, T., Lehmann, B.E. and Loosli, H.H., 1991. Underground production of radionuclides in the Milk River aquifer, Alberta, Canada. Applied Geochemistry, 6, 425-434.

Andrews, J.N. and Fontes, J.-Ch., 1992. Importance of in situ production of chlorine-36, argon-36 and carbon-14 in hydrology and hydrogeochemistry. In Isotope Techniques in Water Resources Development, 1991. International Atomic Energy Agency, Vienna, pp. 245-269.

Andrews, J.N. and Fontes, J.-Ch., 1993. Comment on "Chlorine 36 dating of very old groundwater, 3, further results on the Great Artesian Basin, Australia" by T. Torgersen et al. Water Resources Research, 29, 1871-1874.

Andrews, J.N., Edmunds, W.M., Smedley, P.L., Fontes, J-Ch., Fifield, L.K. and Allan, G.L. 1994. Chlorine-36 in groundwater as a palaeoclimatic indicator: the East Midlands Triassic sandstone aquifer (UK). Earth and Planetary Science Letters, 122, 159-171.

APHA, 1998. American Public Health Association: Standard Methods for the Examination of Water and Wastewater, 20th ed. Washington DC.

Aquachem, 1999. Aqueous geochemical analysis, plotting and modeling software, version 3.7, Waterloo Hydrogeologic Inc.

Aquilina, L., Ladouche, B., Doerfliger, N., Seidel, J.L., Bakalowicz, M., Dupuy, C. and Le Strat, P. 2002. Origin, evolution and residence time of saline thermal fluids (Balaruc springs, southern France): Implications for fluid transfer across the continental shelf. Chemical Geology, 192, 1-21.

Arakel, A.V., 1991. Evolution of Quaternary duricrusts in Karinga Creek drainage system, central Australian groundwater discharge zone. Australian Journal of Earth Sciences, 38, 333-347.

Bajjali, W. and Abu-Jaber, N., 2001. Climatological signals of paleogroundwater in Jordan. Journal of Hydrology, 243, 133-147.

Balderer, W. and Synal, H.-A., 1996. Application of the chlorine-36 method for the characterization of the groundwater circulation in tectonically active areas: examples from northwestern Anatolia/Turkey. Terra Nova, 8, 324-333.

Balderer, W., Synal, A., Deak, J. and Stichler, W., (1997). Application of the chlorine-36 method for the investigation of the infiltration conditions and evolution of the deep groundwaters of multilayer aquifer systems of the great Hungarian plain. *In*, Isotope techniques in the study of environmental change: Study of past and current environmental changes in the hydrosphere and the atmosphere, International Atomic Energy Agency, Vienna: 761-766.

Barker, J.A, 1988. A generalized radial flow model for hydraulic tests in fractured rock. Water Resources Research, 24, 1796-1804.

Barker, J.A., 1991. The reciprocity principle and an analytical solution for Darcian flow in a network. Water Resources Research, 27, 743-746.

Barnes, C.J., Jacobson, G. and Smith, G.D., 1992. The origin of high-nitrate groundwaters in the Australian arid zone. Journal of Hydrology 137, 181-197.

Bates, R.L. and Jackson, J.A., (eds) 1987. Glossary of geology. American Geological Institute, Alexandrea, Virginia, USA, 788 pp.

Beasley, T.M., Cecil, L.D., Sharma, P., Kubik, P.W., Fehn, U., Mann, L.J. and Gove, H.E., 1993. Chlorine-36 in the snake river plain aquifer at the Idaho national engineering laboratory: origin and implications. Ground Water, 31, 302-310.

Beck, J.W., Richards, D.A., Edwards, R.L., Silverman, B.W., Smart, P.L., Donahue, D.J., Hererra-Osterheld, S., Burr, G.S., Calsoyas, L., Jull, A.J.T. and Biddulph, D., 2001. Extremely large variations of atmospheric ¹⁴C concentration during the last glacial period. Science, 292, 2453-2458.

Bentley, H.W., Phillips, F.M., Davis, S.N., Gifford, S., Elmore, D., Tubbs, L.E. and Gove, H.E., 1982. Thermonuclear ³⁶Cl pulse in natural water. Nature, 300, 737-740.

Bentley, H.W., Phillips, F.M. and Davis, S.N., 1986a. Chlorine-36 in the terrestrial environment. *In* Fritz, P. and Fontes, J.C. (eds.), Handbook of Environmental Isotope Geochemistry, Vol. 2. Elsevier, Amsterdam, pp. 427-480.

Bentley, H.W., Phillips, F.M., Davis, S.N., Habermehl, M.A., Airey, P.L., Calf, G.E., Elmore, D., Gove, H.E. & Torgersen, T., 1986b. Chlorine 36 dating of very old groundwater 1. The Great Artesian Basin, Australia. Water Resources Research, 22, 1991-2001

Berkowitz, B., 2002. Characterizing flow and transport in fractured geological media: A review. Advances in Water Resources, 25, 861-884.

Bernadiner, M.G. and Protopapas, A.L., 1994. Progress on the theory of flow in geologic media with threshold gradient. Journal of Environmental Science and Health. 29, 249-275.

Bethke, C.M., Torgersen, T. and Park, J., 2000. The 'age' of very old groundwater: Insights from reactive transport models. Journal of Geochemical Exploration, 69-70: 1-4.

Bethke, C.M. and T.M. Johnson, 2002, Ground water age. Ground Water, 40, 337-339.

Bird, J.R., Davie, R.F., Chivas, A.R., Fifield, L.K. and Ophel, T.R., 1991. ³⁶Cl production and distribution in Australia. Palaeogeography, Palaeoclimatology, Palaeoecology, 84, 299-307.

Blinov, A., Massonet, S. Sachsenhauser, H., Stan-Sion, C., Lazarev, V., Beer, J., Synal, H.-A., Kaba, M., Masarik, J. and Nolte, E., 2000. An excess of ³⁶Cl in modern atmospheric precipitatipion. Nuclear Instruments and Methods in Physics Research Section B: Beam Interactions with Materials and Atoms, 172, 537-544.

BMR 1970. Bureau of Mineral Resources. Amadeus Basin (East), 1:500,000 geological map. Bureau of Mineral Resources, Canberra.

Boehmer, W.K., 1993. Secondary type aquifers and the value of pumping tests for the evaluation of groundwater potential. In Banks, S. and Banks, D. (Eds) Hydrogeology of Hard rocks. Memoires of the XXIVth congress, part 1, International Association of hydrogeologists, Oslo, Norway, 159-168.

Boehmer, W.K. and Boonstra, J., 1987. Analysis of drawdown in the country rock of composite dyke aquifers. Journal of Hydrology 94, 199-214.

BOM, Bureau of meterology http://www.bom.gov.au

Boonstra, J. and Boehmer, W.K., 1986. Analysis of data from aquifer and well tests in intrusive dykes. Journal of Hydrology 88, 301-317.

Bourke, M.C., 1998. Channel adjustment to extreme floods in arid central Australia. Proceedings of the Institute of Australian Geographers and New Zealand Geographical Society, Second Joint Conference, Hobart, Australia, 1997, Massey University Printery, Palmerston North, New Zealand. pp 488 – 492.

Bowman, D.M.J.S. 2002, Personal communication, 18th September, 2002, Northern Territory University, Darwin.

Bredehoeft, J. 1997. Safe yield and the water budget myth. Ground Water, 35 (No.6), p 929

Brown, D.M., Lloyd, J.W. and Jacobson, G., 1990. Hydrogeological model for Amadeus Basin aquifers, central Australia. Australian Journal of Earth Sciences, v. 37, p. 215-226.

Bureau of Rural Sciences, 2001. (http://www.dest.gov.au/radwaste/publications.htm) Australia's national radioactive waste repository – Phase 3 site assessment and stage 3 drilling: Stage 3 Assessment Report, Appendix C2, groundwater recharge considerations, 8p.

Burton, W.C., Plummer, L.N., Busenburg, E., Lindsey, B.D. and Gburek, W.J., 2002. Influence of fracture anisotropy on ground water ages and chemistry, Valley and Ridge Province, Pennsylvania. Ground Water, 40 (3), 242-257.

Cacas, M.C., Ledoux, E., de Marsily, G., Tillie, B., Barbreau, A., Calmels, P., Gailard, B., Margrita, R., Durand, E., Feuga, B. and Peaudecerf, P., 1990. Flow and transport in fractured rocks: an in situ experiment in the Fanay-Augeres Mine and its interpretation with a discrete fracture network model. In Water Resources in Mountainous Regions, Memoires of the 22nd Congress of the International Association of Hydrogeologists, Volume XXII, Lausanne, Switzerland, 1990, 13-38.

Calf, G.E., 1978. An investigation of recharge to the Namoi Valley aquifers using environmental isotopes. Australian Journal of Soil Sciences, 16, 197-207.

Calf, G.E., 1979. The isotope hydrology of the Mereenie Sandstone aquifer, Alice Springs, Northern Territory, Australia. Journal of Hydrology, 38, 343-355.

Calf, G.E., McDonald, P.S. and Jacobson, G., 1991. Recharge mechanism and groundwater age in the Ti Tree Basin, Northern Territory. Australian Journal of Earth Sciences 38, 299-306.

Campbell, K., Wolfsberg, A., Fabryka-Martin, J. and Sweetkind, D., 2003. Chlorine-36 data at Yucca Mountain: Statistical tests of conceptual models for unsaturated-zone flow. Journal of Contaminant Hydrology, 62-3: 43-61.

Carothers, W.W., Kharaka, Y.K., 1980. Stable isotopes of HCO₃- in oil field waters implications for the origin of CO₂. Geochimica et Cosmochimica Acta 44, 323–332.

Cartwright, I.,. Weaver, T. R., Fulton, S., Nichol, C., Reid M., and Cheng X., 2004. Hydrogeochemical and isotopic constraints on the origins of dryland salinity, Murray Basin, Victoria, Australia. Applied Geochemistry, 19, 1233-1254. CCNT, 1992, Finke Gorge National Park concept plan, Conservation Commission of the Northern Territory, Alice Springs.

Cerling, T.E., Pederson, B.L. and Von Damm, K.L., 1989. Sodium-calcium ion exchange in weathering of shales: implications for global weathering budgets. Geology, 17, 552-554.

Chen, X. Y. and Barton, C. E., 1991. Onset of aridity and dune building in central Australia: sedimentological and magnetostratigraphic evidence from Lake Amadeus, central Australia. Palaeogeography., Palaeoclimatology, Palaeoecology, v. 84, p.55-73.

Chewings, C., 1931. A delineation of the Precambrian plateau in central and north Australia, with notes on the impingent sedimentary formations. Transactions of the Royal society of South Australia, 55, 1-11.

Childs, J. (compiler). 1990. Groundwater Source Review, Hermannsburg. NT Water Resources Branch, Alice Springs, Report # 20/1990, 20 pp.

Chmiel, G., Fritz, S.J. and Elmore, D., 2003. Control of ³⁶Cl production in carbonaceous shales by phosphate minerals. Geochimica et Cosmochimica Acta, 67, 2377-2395.

Clark, I.D. and Fritz, P., 1997. Environmental Isotopes in Hydrogeology. CRC Press Florida, 320 p.

Collon, P., Kutschera, W., Loosli, H.H., Lehmann, B.E., Purtschert, R., Love, A., Sampson, L., Anthony, D., Cole, D., Davids, B., Morrissey, D.J., Sherrill, B.M., Steiner, M., Pardo, R.C. and Paul, M., 2000. ⁸¹Kr in the Great Artesian Basin, Australia: a new method for dating very old groundwater. Earth and Planetary Science Letters, 182, 103-113.

Colman, S.M., Peck, J.A., Karabanov, E.B., Carter, S.J., Bradbury, J.P., King, J.W. and Williams, D.F., 1995, Continental climate response to orbital forcing from biogenic silica records in Lake Baikal: Nature, v. 378, p. 769-771.

Commander, D.P., Fifield, L.K., Thorpe, P.M., Davie, R.F., Bird, J.R. and Turner, J.V., 1994. Chlorine-36 and carbon-14 measurements on hypersaline groundwater in Tertiary palaeochannels near Kalgoorlie, Western Australia. Western Australia Geological Survey, Report 37, Professional Papers, 53-60.

Cook, P.J., 1968. (compiler) Henbury, N.T. 1:250,000 geological series – explanatory notes. Sheet SG/53-1. Bureau of Mineral Resources, Canberra.

Cook, P.G., 2003. A guide to regional groundwater flow in fractured rock aquifers. CSIRO, Seaview Press, South Australia, 116 p.

Cook, P.G., Jolly, I.D., Leaney, F.W., Walker, G.R., Allan, G.L., Fifield, L.K. and Allison, G.B., 1994. Unsaturated zone tritium and chlorine-36 profiles from southern Australia: their use as tracers of soil water movement. Water Resources Research, 30, 1709-1719.

Cook, P.G., Hatton, T.J., Pidsley, D., Herczeg, A.L., Held, A., O'Grady, A.P., Eamus, D., 1998. Water balance of a tropical woodland ecosystem, northern Australia: a combination of micro-meteorological, soil physical and groundwater chemical approaches. Journal of Hydrology 210, 161–177.

Cook, P.G. and Herczeg, A. L., (Eds.), 2000. Environmental tracers in subsurface hydrology. Kluwer Academic Publications, Boston, 529p.

Cooper, H.H., Jr. and Jacob, C.E., 1946. A generalized graphical method for evaluating formation constants and summarizing well field history. Transactions of the American Geophysical Union, 27, 526-534.

Corcho Alvarado, J.A., Purtschert, R., Hinsby, K., Troldborg, L., Hofer, M., Kipfer, R., Aeschbach-Hertig, W. and Arno-Synal, H., 2005. ³⁶Cl in modern groundwater dated by a multi tracer approach (³H/³He, SF₆, CFC-12 and ⁸⁵Kr): a case study in quaternary sand aquifers in the Odense Pilot River Basin, Denmark. Applied Geochemistry, 20, 599-609.

Cornett, R.J., Andrews, H.R, Chant, L.A., Davies, W.G., Greiner, B.F., Imahori, Y., Koslowsky, V.T., Kotzer, T., Milton, J.C.D. and Milton, G.M., 1997. Is ³⁶Cl from weapons' test fallout still cycling in the atmosphere? Nuclear Instruments and Methods in Physics Research Section B: Beam Interactions with Materials and Atoms, 123, 378-381.

Cresswell, R., Wischusen, J., Jacobson, G. and Fifield, K., 1999a. Assessment of recharge to groundwater systems in the arid southwestern part of Northern Territory, Australia, using chlorine-36. Hydrogeology Journal, 7(4): 393-404.

Cresswell, R.G., Jacobson, G., Wischusen, J. and Fifield, L.K., 1999b. Ancient groundwaters in the Amadeus Basin, Central Australia: evidence from the radio-isotope ³⁶Cl. Journal of Hydrology, 223, 212-220.

Cresswell, R.G., Bauld, J., Jacobson, G., Khadka, M.S., Jha, M.G., Shrestha, M.P. and Regmi, S., 2001. A first estimate of ground water ages for the deep aquifer of the Kathmandu Basin, Nepal, using the radioisotope chlorine-36. Ground Water, 39, 449-457.

Cresswell, R., Hostetler, S. and Jacobson, G., 2002. Rapid episodic recharge in the arid north of South Australia. Proceedings of "Balancing the Groundwater Budget" International Association of Hydrogeologists Conference, Darwin, Northern Territory, CD-ROM.

Croke, J., Magee, J.W. and Price, D.M., 1996, Major episodes of Quaternary activity in the lower Neales River, northwest of Lake Eyre, central Australia: Palaeogeography., Palaeoclimatology, Palaeoecology, v. 124, p.1-15.

Crombie, M.K., Arvidson, R.E., Sturchio, N.C., El Alfy, Z. and Abu Zeid, K., 1997. Age and isotopic constraints on Pleistocene pluvial episodes in the Western Desert, Egypt. Palaeogeography., Palaeoclimatology,. Palaeoecology, v. 130, p.337-355. Dames and Moore, 1982. Environmental assessment and management report, Palm Valley pipeline project.

Dansgaard, W., 1964. Stable isotopes in precipitation. Tellus, 16, 436 – 468.

Davie, R.F., Kellett, J.R., Fifield, L.K., Evans, W.R., Calf, G.E., Bird, J.R., Topham, S. and Ophel, T.R., 1989. Chlorine-36 measurements in the Murray Basin: preliminary results from the Victorian and South Australian Mallee region. Bureau of Mineral Resources, Journal of Geology and Geophysics, 11, 261-272.

Davis, J.A., 1997. Conservation of aquatic invertebrate communities in central Australia. Memoirs of the Museum of Victoria, 56(2), 491-503.

Davis, J.A., 1995, The aquatic ecosystems of the west Macdonnell Ranges, Northern Territory; with particular reference to the aquatic invertebrate fauna: Report to the Parks and Wildlife Commission of the Northern Territory, 50 p.

Davis, S.N., Cecil, D., Zreda, M. and Sharma, P., 1998a. Chlorine-36 and the initial value problem. Hydrogeology Journal, 6, 104-114.

Davis S.N., Whittemore, D.O. and Fabryka-Martin J., 1998b. Uses of chloride/ bromide ratios in studies of potable water. Groundwater, 36, 338-350.

Davis S.N., Fabryka-Martin J., Wolfsberg L., Moysey S., Shaver R., Alexander E.C. and Krothe N., 2000. Chlorine-36 in ground water containing low chloride concentrations. Ground Water. 38, 912-921.

Davis, S.N., Cecil, L.D., Zreda, M. and Moysey, S., 2001. Chlorine-36, bromide, and the origin of spring water. Chemical Geology, 179, 3-16.

Davis, S.N., Moysey, S., Cecil, L.D. and Zreda, M., 2003. Chlorine-36 in groundwater of the United States: empirical data. Hydrogeology Journal, 11, 217-227.

De Wiest, R.J.M., 1965. Geohydrology. John Wiley and Sons, Inc. New York, USA, 366 pp.

Deffeyes, K.S., 1965. Carbonate equilibria: a graphic and algebraic approach. Limnology Oceanography. 10, 412-426.

Dighton, J.C., Leaney, F.W.J., Herczeg, A.L., Allison, G.B. and Hughes, M.W., 1997. A rapid and robust method for the preparation of isotopic ally unaltered hydrogen gas from water for stable isotope mass spectrometry. Sixth Australian-New Zealand Environmental Isotope Conference, Lower Hutt, New Zealand.

Do Rozario and Baird, 1987. The detection and significance of fractures in the Palm Valley gas field. The APEA journal, 27, 264-280.

Do Rozario, R.F., 1991. The Palm Valley Gas Field, Amadeus Basin, central Australia. In Korsch, R.J. and Kennard, J.M. (eds), Geological and geophysical studies of the Amadeus Basin, central Australia. Bureau of Mineral Resources, Australia, Bulletin 236, 477-492.

Domenico, P.A. and Schwartz, F.W., 1988. Physical and chemical hydrogeology. Wiley and Sons, New York. 506p.

Dowe, John, 2005. Personal communication, 14th of November 2005. Australian Centre for Tropical Freshwater Research, James Cook University, Townsville, Qld.

Dowgiallo, J. Nowicki, Z., Beer, J., Bonani, G., Suter, M., Synal, H.A. and Wolfi, W., 1990. ³⁶Cl in ground water of the Mazowsze Basin (Poland). Journal of Hydrology, 118(1-4): 373-385.

Dransfield, John, 2005. Personal communication, 14th November 2005. Royal Botanical Gardens, Kew, England.

Drever, J.I., 1988. The geochemistry of natural waters, 2nd ed. Prentice Hall, New Jersey, 437 p.

Eamus, D., Hutley, L. and O'Grady, A.P 2001. Carbon and water fluxes above a north Australian savanna . Tree Physiology 21: 977-988.

Edmunds, W.M., 1995. Characterisation of groundwaters in semi-arid and arid zones using minor elements. In Nash, H. and McCall, G.J.H. (Eds.) Groundwater quality, special report No. 17. Chapmann Hall, 19-30.

Edmunds, W. M., 1996. Bromine chemistry in British groundwaters. Mineralogical magazine, 60, 275-284.

Edmunds, W.M. and Smedley, P.L., 2000. Residence time indicators in groundwater: The East Midlands Triassic sandstone aquifer. Applied Geochemistry, 15, 737-752.

Edmunds, W.M., Guendouz, A.H., Mamou, A., Moulla, A., Shand, P. and Zouari, K., 2003. Groundwater evolution in the Continental Intercalaire aquifer of southern Algeria and Tunisia: trace element and isotope indicators. Applied Geochemistry 18, 805-822.

Elmore, D., Fulton, B.R., Clover, M.R., Marsden, J.R., Gove, H.E., Naylor, H., Purser, K.H., Kilius, L.K., Beukens, R.P. and Litherland, A.E., 1979. Analysis of ³⁶Cl in environmental water samples using an electrostatic accelerator. Nature, 277, 22-25.

Elmore, D., Tubbs L.E., Newman D., Ma X.Z., Finkel R., Nishiizumi K., Beer J., Oeschgar H. and Andrea M., 1982. The ³⁶Cl bomb pulse measured in a shallow ice core from Dye3, Greenland, Nature, 300, 735-737.

Fabryka-Martin, J., Davis, S.N. and Elmore, D., 1987. Applications of I and Cl in hydrology. Nuclear Instruments & Methods in Physics Research, Section B: Beam Interactions with Materials and, B29, 361-371.

Fabryka-Martin, J., Whittemore, D.O., Davis, S.N., Kubik, P.W. and Sharma, P., 1991. Geochemistry of halogens in the Milk River aquifer, Alberta, Canada. Applied Geochemistry, 6(4): 447-464. Fehn, U., Moran, J.E., Teng, R.T.D. and Rao, U., 1994. Dating and tracing of fluids using ¹²⁹I and ³⁶Cl: results from geothermal fluids, oil field brines and formation waters. Nuclear Instruments and Methods in Physics Research Section B: Beam Interactions with Materials and Atoms, 92, 380-384.

Feige Y., Oltman B.G and Kastner J. (1968) Production rates of neutrons in soils due to natural radioactivity. Journal of Geophysical research, 73, 3135-3142.

Fetter C.W., 1993. Contaminant hydrogeology. Prentice Hall, 458 p.

Fifield, L.K., Ophel, T.R., Bird, J.R., Calf, G.E., Allison, G.B. and Chivas, A.R., 1987. The ³⁶Cl measurement program at the Australian national University. Nuclear Instruments and Methods in Physics Research, 29, 114-119.

Fifield, L.K., Ophel, T.R., Allan, G.L., Bird, J.R. and Davie, R.F., 1990. Accelerator mass spectrometry at the Australian National University's 14UD accelerator: experience and developments. Nuclear Instruments and Methods in Physics Research Section B: Beam Interactions with Materials and Atoms, 52, 233-277.

Fifield, L.K., Allan, G.L., Stone, J.O.H. and Ophel, T.R., 1994. The ANU AMS system and research program. Nuclear Instruments and Methods in Physics Research Section B: Beam Interactions with Materials and Atoms, 92, 85-88

Finkel, R.C., Nishiizumi, K., Elmore, D., Ferraro, R.D. and Gove, H.E., 1980. ³⁶Cl in polar ice, rainwater and seawater. Geophysical Resaerch Letters, 7, 983-986.

Fishel, V.C., 1935. Further tests of permeability with low hydraulic gradients. American Geophysical Union Transactions, 1935 part 2, 499-503.

Fisher, R.S. and Mullican, W.F., 1997. Hydrochemical evolution of sodium-sulfate and sodium-chloride groundwater beneath the northern Chihuahuan Desert, Trans-Pecos, Texas, USA. Hydrogeology Journal, 5, 4-16.

Fontes, J. Ch. and Garnier, J.M., 1979. Determination of the initial ¹⁴C activity of total dissolved carbon: A review of existing models and a new approach. Water Resources Research, 15, 399-413.

Fontes, J.C., 1980. Environmental isotopes in groundwater hydrology. Chapter 3 In: Fritz, P. and Fontes, J.C. (Eds.) Handbook of Environmental Isotope Chemistry. 1, Elsevier, Amsterdam, The Netherlands, 75-140.

Fontes, J.-Ch., Brissaud., I and Michelot., J.L., 1984. Hydrological implications of deep production of chlorine-36. Nuclear Instruments and Methods in Physics Research, Section B: Beam Interactions with Materials and Atoms, 5, 303-307

Fontes, J.-Ch. and Andrews, J.N., 1993a. Comment on "The ³⁶Cl ages of the brines in the Magadi-Natron basin, East Africa" by A. Kaufman et al. Geochimica et Cosmochimica Acta, 57, 1607-1612

Fontes, J.-Ch. and Andrews, J.N., 1993b. Comment on "Reinterpretation of ³⁶Cl data: physical processes, hydraulic interconnections and age estimates in groundwater systems" by E. Mazor. Applied Geochemistry, 8, 663-666.

Fontes, J.-Ch. and Andrews, J.N., 1994. Accelerator mass spectrometry in hydrology. Nuclear Instruments and Methods in Physics Research Section B: Beam Interactions with Materials and Atoms, 92, 367-375.

Foster, S.S.D., 1988. Quantification of groundwater recharge in arid regions: a practical view for resource development and management. In Simmers I (ed), Estimation of Natural Groundwater Recharge; NATO, Advanced Science Institutes series C, Volume 222, D. Reidel Publishing Company, Dordrecht, Holland, 239-270.

Foster, S.S.D., Garduno, G., Evans, R., Olson, D., Tian, Y., Zhang, W. and Han, Z., 2004. Quaternary Aquifer of the North China Plain—assessing and achieving groundwater resource sustainability. Hydrogeology Journal, 12, 81-93.

Freeze, R.A. and Cherry, J.A., 1979, Groundwater: Prentice-Hall, New Jersey, USA, 604 p.

Firtz, P. and Fontes, J-Ch. (Eds.). 1986. Handbook of environmental isotope geochemistry, Volume 2, The terrestrial environment, B. Elsevier, Amsterdam, The Netherlands, 577 p.

Fröhlich, K, Ivanovich, M., Hendry, M.J., Andrews, J.N., Davis, S.N., Drimmie, R.J., Fabryka-Martin, J., Florkowski, T., Fritz, P., Lehmann, B., Loosli, H.H. and Nolte, E., 1991. Application of isotopic methods to dating of very old groundwaters: Milk River aquifer, Alberta, Canada. Applied Geochemistry, 6, 465-472.

Gascoyne, M., Sharma, P. and Kubik, P.W., 1994. Origin of groundwater salinity in the Lac du Bonnet granite, southeastern Manitoba, from ³⁶Cl measurements. Nuclear Instruments and Methods in Physics Research Section B: Beam Interactions with Materials and Atoms, 92, 389-392.

Gat, J.R., 1980. The isotopes of hydrogen and oxygen in precipitation. Chapter 1 In: Fritz, P. and Fontes, J.C. (Eds.) Handbook of Environmental Isotope Chemistry. 1, Elsevier, Amsterdam, 21-47.

Gat, J.R., 1981. Isotopic Fractionation. In: Stable isotope hydrology, Deuterium and Oxygen -18 in the water cycle. Technical Reports Series No. 210, International Atomic Energy Agency, Vienna, p 21-34.

Geyh, M.A., 2000. An overview of ¹⁴C analysis in the study of groundwater. Radiocarbon, 42, 99-114.

Geyh, M.A. and Eitel, B., 1998. Radiometric dating of young and old calcrete. Radiocarbon, 40, 795-802.

Goguel, R., 1969, Direct spectrophotometric determination of sulfate in natural waters by formation of the ferric sulfate complex, Analytical Chemistry 1969, 41, 1034-1038.

Goni, G.B., Fellman, E. and Edmunds, W.M., 2001. Rainfall geochemistry in the Sahel region of northern Nigeria. Atmospheric Environment, 35, 4331-4339.

Gosse, J.C. and Phillips, F.M., 2001. Terrestrial in situ cosmogenic nuclides: theory and application. Quaternary Science Reviews, 20, 1475-1560.

Gray, 1999. Aboriginal Land rights (Northern Territory) Act 1976. The Palm Valley Land Claim No. 48. Report and recommendations by Justice Gray. Aboriginal and Toress straight Islander commission, Australian Government, Canberrra. 184 p.

Guendouz, A.H., Mamou, A., Moulla, A., Shand, P. and Zouari, K., 2003. Groundwater evolution in the Continental Intercalaire aquifer of southern Algeria and Tunisia: trace element and isotope indicators. Applied geochemistry 18, 805-822.)

Guyodo, Y. and Valet, J-P., 1999. Global changes in intensity of the Earth's magnetic field during the past 800 kyr. Nature, 399, 249-252.

Hainsworth, L.J., Mignerey, A.C., Helz, G.R., Sharma, P. and Kubik, P.W., 1994. Modern chlorine-36 deposition in southern Maryland, U.S.A. Nuclear Instruments and Methods in Physics Research Section B: Beam Interactions with Materials and Atoms, 92, 345-349.

Hamm, S.,-Y. and Bidaux, P., 1996. Dual-porosity fractal models for transient flow analysis in fractured rocks. Water Resources Research, 32, 2733-2745.

Harrington, G.A., 1999. Recharge mechanisms and chemical evolution in an arid groundwater system. Central Australia. PhD thesis, Flinders University, South Australia, 293p.

Harrington, G.A., Herczeg, A.L., 2000. Estimating groundwater ¹⁴C ages in the arid Ti-Tree Basin, Central Australia: use of ⁸⁷Sr/⁸⁶Sr to constrain sources of inorganic carbon. Proceedings International Symposium on Isotope Techniques in Water Resources Management and Development. International Atomic Energy Agency (IAEA), Vienna, pp. 272–273. Harrington, G.A., Cook, P.G. and Herczeg, A.L., 2002. Spatial and temporal variability of ground water recharge in central Australia: a tracer approach. Ground Water, 40, 518-528.

Harrington, G.A., Herczeg, A.L., 2003. The importance of silicate weathering of a sedimentary aquifer in arid central Australia indicated by very high ⁸⁷Sr/⁸⁶Sr ratios. Chemical Geology, 199, 281-292.

Hazel, C.P., 1975. Groundwater hydraulics. Australian Water Resources Council, Groundwater School, Adelaide, 242 pp.

Hem, J.D., 1985, Study and interpretation of the chemical characteristics of natural water: United States Geological Survey, Water Supply Paper 2254, 3rd Edition, 263p.

Hendrickx, J.M.H. and Walker, G.R., 1997. Recharge from precipitation. In Simmers, I. (Ed.), Recharge from Phreatic Aquifers in (semi) Arid Areas. International Association of Hydrogeologists, International contributions to Hydrogeology 19. AA Balkema, Rotterdam. 19-111.

Hendry, M.J. and Schwartz, F.W., 1988. An alternative view on the origin of chemical and isotopic patterns in groundwater for the Milk River aquifer, Canada. Water Resources Research, 24, 1747-1763.

Hendy, C.H., 1971. The isotopic geochemistry of speleothems: I. The calculation of the effects of different modes of formation on the isotopic composition of speleothems and their applicability as palaeoclimatic indicators. Geochimica et Cosmochimica Acta 35, 801–824.

Herczeg, A.L., Torgersen, T., Chivas, A.R. and Habermehl, M.A., 1991. Geochemistry of ground waters from the Great Artesian Basin, Australia. Journal of Hydrology, 126, 225-245.

Herczeg, A.L., Leaney, F.W.J., Stadler, M.F., Allan, G.L. and Fifield, L.K., 1997. Chemical and isotopic indicators of point-source recharge to a karst aquifer, South Australia. Journal of Hydrology, 192, 271-299.

Herut, B., Starinsky, A., Katz, A., Paul, M., Boaretto, E. and Berkovits, D., 1992. ³⁶Cl in chloride-rich rainwater, Israel. Earth and Planetary Science Letters, 109, 179-183.

Hesse, P.P., Magee, J.W. and van der Kaars, S., 2004. Late quaternary climates of the Australian arid zone: a review. Quaternary International, 118-119, 87-102

Hostetler, S.D., 2000. Age of groundwater, Alice Springs, N.T. Bureau of Rural Sciences report to the NT department of Lands, Planning and Environment, November 2000. Report # A1186, 18 p.

Hostetler, S.D., 2002. Recharge to the Amadeus Basin around Alice Springs, central Australia. Proceedings of "Balancing the Groundwater Budget" International Association of Hydrogeologists Conference, May 2002, Darwin, Northern Territory, CD-ROM.

Hutton, J.T., 1983 Soluble ions in rainwater collected near Alice Springs N.T., and their relation to locally derived atmospheric dust. Transactions of the Royal Society of South Australia, 107 part 2, 138)

IAEA/WMO (2001). International Atomic Energy Agency/ World Meteorological Organization, Global Network of Isotopes in Precipitation. Alice Springs station 1962 – 1966. The GNIP Database. Accessible at: http://isohis.iaea.org.

Ingerson R. and Pearson F.J., 1964. Estimation of age and rate of motion of groundwater by the ¹⁴C method. In, Miyake, Y. and Koyama, T. (Eds.). Recent Research in the Field of Hydrosphere, Atmosphere and Nuclear Chemistry, Marusen, Tokyo, 263-283.

Ingraham, N.L. and Matthews, R.A., 1988. Fog drip as a source of groundwater recharge in northern Kenya. Water resources Research, 24, 1406-1410.

Isagi, yuji. 2003. Personal communication, 31 October, 2003. Faculty of intergrated arts and sciences, Hiroshima University, Japan.

Jacobson, G., Calf, G.E., Jankowski, J. and McDonald, P.S., 1989, Groundwater chemistry and palaeorecharge in the Amadeus Basin, central Australia: Journal of Hydrology, v. 109, p. 237-266.

Jacobson, G. and Lau, J.E., 1991. The Mereenie Sandstone aquifer - a vital resource for central Australia. In Proceedings of the International Conference on Groundwater in Large Sedimentary Basins, Perth 1990. Australian Water Resources Conference Series No 20, Australian Government Printer, Canberra, 110-118.

Jacobson, G, Jankowski, J. and Abell, R.S., 1991. Groundwater and surface water interaction at Lake George, New South Wales. BMR Journal of Australian Geology and Geophysics, 12, 161-190.

Jacobson, G., Barnes, C.J., Fifield, L.K. & Cresswell, R.G., 1994. The time factor in arid-zone groundwater recharge. Water Down-under 94: 25th Congress of the International Association of Hydrogeologists. Adelaide 21-25 November. The Institution of Engineers, Australia, Preprints of Papers, 2B, 471-478.

Jacobson, G, Cresswell, R and Hostetler, S., 2000. The age of groundwaters at Uluru, central Australia – Report to Uluru – Kata Tjuta National Park, April 2000. 30 p.

Jankowski, J. and Jacobson, G., 1989. Hydrochemical evolution of regional groundwaters to playa brines in central Australia. Journal of Hydrology 108, 123-173.

Johnson, B.J., Miller, G.H., Fogel, M.L., Magee, J.W., Cagan, M.K. and Chivas, A.R., 1999. 65,000 years of vegetation change in central Australia and the summer monsoon. Science, 284, 1150-1152.

Jolly, P.B., Prowse, G.W. and Chin, D.N., 1991. The Amadeus Basin Mereenie Sandstone aquifer - regional modelling based on the diagenetic history of the Mereenie Sandstone. In Proceedings of the International Conference on Groundwater in Large Sedimentary Basins, Perth, 1990. Australian Water Resources Conference Series No 20, Australian Government Printer, Canberra, 340-354.

Jones, B.G., 1972, Upper Devonian to Lower Carboniferous stratigraphy of the Pertnjara Group, Amadeus Basin, central Australia: Journal of the Geological Society of Australia, v. 19, p. 229-249.

Jones, B.G., 1991. Fluvial and lacustrine facies in the Middle to Late Devonian Pertnjara Group, Amadeus Basin, Northern Territory, and their relation to tectonic events and climate. *In* Korsch, R.J. and Kennard, J.M. (Editors), Geological and geophysical studies in the Amadeus Basin, central Australia. Bureau of Mineral Resources, Australia, Bulletin 236, 333-347.

Jones, D.L., 2002. Cycads of the world: ancient plants in today's landscape. Reed New Holland, 456 p.

Jourde, H., Cornation, F., Pistre, S. and Bidaux, 2002. Flow behavior in a dual fracture network. Journal of Hydrology. 266, 99 – 119.

Kalin, R.M., 2000. Radiocarbon dating of groundwater systems In Cook, P.G. and Herczeg, A.L. (Eds.) Environmental tracers in subsurface hydrology. Kluwer Academic Publishing, Boston, 111-144

Kaufman, A., Margaritz, M., Paul, M., Hillaire-Marcel, C., Hollos, G., Boaretto, E. and Taeb, M., 1990. The ³⁶C1 ages of the brines in the Magadi-Natron basin, East Africa. Geochimica et Cosmochimica Acta, 54, 2827-2833.

Kaufman, A., Margaritz, M., Paul, M. and Hillaire-Marcel, C., 1993. Reply to comment by J.-Ch. Fontes and J.N. Andrews on "the ³⁶Cl ages of the brines in the Magadi-Natron basin, East Africa". Geochimica et Cosmochimica Acta, 57, 1613-1617.

Kellett, J.R., Evans, W.R., Allan, G.L., Davie, R.F. and Fifield, L.K., 1990. A study of stable chloride and chlorine-36 in the Murray Group Limestone, Western Murray Basin.

In Proceedings of the International Conference on Groundwater in Large Sedimentary Basins, Perth, 1990. Australian Water Resources Conference Series No 20, Australian Government Printer, Canberra, 403-419.

Kellett, J.R., Evans, W.R., Allan, G.L. and Fifield, L.K., 1993. Reinterpretation of ³⁶Cl data: physical processes, hydraulic interconnections and age estimates in groundwater systems - discussion. Applied Geochemistry, 8, 653-658.

Kershaw A. P. and Nanson G.C., 1993, The last full glacial cycle in the Australian region: Global and Planetary change, v. 7, p. 1-9.

Keywood, M.D., 1995, Origins and sources of atmospheric precipitation from Australia: Chlorine-36 and major ion chemistry: PhD thesis, Australian National University, 207p.

Keywood, M.D., Chivas, A.R., Fifield, L.K., Cresswell, R.G. and Ayers, G.P., 1997. The accession of chloride to the western half of the Australian continent. Australian Journal of Soil Research, 35, 1177-1189.

Keywood, M.D., Fifield, L.K., Chivas, A.R., and Cresswell, R.G., 1998. Fallout of chlorine 36 to the Earth's surface in the southern hemisphere. Journal of Geophysical Research, 103, 8281-8286

Knies, D.L., Elmore, D., Sharma, P., Vogt, S., Li, R., Lipschutz, M.E., Petty, G., Farrell, J., Monaghan, M.C., Fritz, S. and Agee, E., 1994. ⁷Be, ¹⁰Be, and ³⁶Cl in precipitation. Nuclear Instruments and Methods in Physics Research, Section B: Beam Interactions with Materials and Atoms. 92, 340-344.

Korsch, R.J. and Kennard, J.M. (eds.), 1991. Geological and geophysical studies of the Amadeus Basin, central Australia. Bureau of Mineral Resources, Australia, Bulletin 236.

Krusemann, G.P. and de Ridder, N.A., 1970. Analysis and evaluation of pumping test data. International Institute for Land Reclamation and Improvement, Bulletin 11, Wageningen, The Netherlands, 200 pp.

Krusemann, G.P. and de Ridder, N.A., 1991. Analysis and evaluation of pumping test data. 2nd edition, International Institute for Land Reclamation and Improvement, Publication 47, Wageningen, The Netherlands, 377 pp.

Kruseman, G.P., 1997. Recharge from intermittent flow, In Simmers, I., (Ed.) 1997. Recharge of phreatic aquifers in (semi-) arid areas. International Association of Hydrogeologists, 19, A.A. Balkema, Rotterdam, 145-184.

Laing, A.C.M., 1996. Comment on "Hydraulic continuity in large sedimentary basins, by J. Toth". Hydrogeology Journal, 4, 96.

Lal, P. and Peters. B., 1967. Cosmic ray produced radioactivity on the Earth. In Handbuch der Physik, 46, 551-612. Springer-Verlag, Berlin.

Lamontagne, S., Cook, P.G., O'Grady, A. and Eamus, D. 2005 Groundwater use by vegetation in a tropical savanna riparian zone (Daly River, Australia). Journal of Hydrology 310, 280-293.

Larsen, F., Owen, R., Dahlin, T., Mangeya, P. and Barmen, G., 2002. A preliminary analysis of the groundwater recharge to the Karoo formations, mid-Zambezi basin, Zimbabwe. Physics and Chemistry of the Earth, 27, 765-772.

Latz, P.K. 1975. Notes on the relict palm Livistona mariae in central Australia. Transactions of the Royal Society of South Australia 99: 189-196

Latz P. K. 1995 Bushfires and bushtucker: Aboriginal plant use in central Australia. Institute for Aboriginal Development Press, Alice Springs, NT, 400 p.

Latz, P. K., 1996. Knowledge of relict plants in central Australia: a measure of botanical research during the last 100 years. In Morton, S.R. and Mulvaney, D.J. (Eds.) Exploring central Australia: society, the environment and the 1894 Horn expedition. Surrey Beatty & Sons, Chipping Norton., 225-229.

Lau, J.E., 1989. Loggging of diamond drill core from Roe Creek borefield Alice Springs.. NT Water Resources Branch report #1989/49, 140p.

Lau, J. E. and Jacobson, G., 1991, Aquifer characteristics and groundwater resources of the Amadeus Basin. In Korsch, R.J. and Kennard, J.M. (eds.), Geological and geophysical studies of the Amadeus Basin, central Australia. Bureau of Mineral Resources, Australia, Bulletin 236, p.563-580.

Lauriol, B. and Clark, I.D., 1993. An approach to determine the origin and age of massive ice blockages in two Arctic caves. Permafrost and Periglacial Processes, 4, 77-85.

Le Gal La Salle, C., Fontes, J-Ch., Andrews, J. N., Schroeter, P, Karbo, A, and Fifield, K. L., 1995. Old groundwater circulation in the Iullemeden basin (Niger): Preliminary results of an isotopic study, in, Application of tracers in arid zone hydrology, Proceedings of the Vienna Symposium, 1994, IAHS Publication, 1995, Pages 129-139

Le Gal La Salle, C., Marlin, C., Leduc, C., Taupin. J.D., Massault, M. and Favreau, G., 2001. Renewal rate estimation of groundwater based on radioactive (³H, ¹⁴C) in an unconfined aquifer in a semi-arid area, Iullemeden Basin, Niger. Journal of Hydrology 254, 145-156.

Leaney, F.W. and Allison, G.B., 1986. Carbon-14 and stable isotope data for an area in the Murray Basin: its use in estimating recharge. Journal of Hydrology, 88, 129-145.

Leaney, F.W., Herczeg, A.L. & Dighton, J.C., 1994. New developments for the direct CO2 absorption method for radiocarbon analysis. Quaternary Geochronology. (Quaternary Science Review) 13, 171-178.

Leaney, F.W., Herczeg, A.L. and Walker, G.R., 2003. Salinization of a fresh palaeoground water resource by enhanced recharge. Ground Water, 41, 84-92.

Lehmann, B.E., Loosli, H.H. and Pearson, P.J., Jr., 1990. Limit to the use of ⁴He and ³⁶Cl for groundwater dating. *In* Water Resources in Mountainous Regions, Memoires of

the 22nd Congress of the International Association of Hydrogeologists, Volume XXII, Lausanne, Switzerland, 1990, 356-363.

Lehmann, B.E., Love, A., Putschert, R., Collon, P., Loosli, H.H., Kutschera, W., Beyerle, U., Aeschbach-Hertig, W., Kipfer, R., Frape, S.K., Herczeg, A., Moran, J., Tolstikhin, I.N. and Gröning, M., 2003. A comparison of groundwater dating with ⁸¹Kr, ³⁶Cl and ⁴He in four wells of the Great Artesian Basin, Australia. Earth and Planetary Science Letters, 211, 237-250.

Lerner, D.N., Issar, A.S. and Simmers, I., 1990. Groundwater recharge, a guide to understanding and estimating natural recharge. International Association of Hydrogeologists - International Contributions to Hydrogeology, Volume 8, Heise Hannover, Germany, 345 pp.

Lindsay, J.F. and Korsch, R.J., 1991. The evolution of the Amadeus Basin, central Australia. In Korsch, R.J. and Kennard, J.M. (Editors), Geological and geophysical studies in the Amadeus Basin, central Australia. Bureau of Mineral Resources, Australia, Bulletin 236, 429-462.

Lippmann, J., Stute, M., Torgersen, T., Moser, D. P., Hall, J. A., Lin, L., Borcsik, M., Bellamy R. E. S. and Onstott, T. C. , 2003. Dating ultra-deep mine waters with noble gases and ³⁶Cl, Witwatersrand Basin, South Africa, Geochimica et Cosmochimica Acta, 67, (23), 4597-4619.

Liu, B., Phillips, F., Hoines, S., Campbell, A.R. and Sharma, P., 1995. Water movement in desert soil traced by hydrogen and oxygen isotopes, chloride, and chlorine-36, southern Arizona. Journal of Hydrology, 168, 91-110

Lloyd, J.W. and Heathcote, J.A., 1985. Natural inorganic hydrochemistry in relation to groundwater: an introduction. Oxford University Press, 296 p.

Lloyd J.W. and Jacobson G., 1987. :The hydrogeology of the Amadeus basin, central Australia. Journal of Hydrology, 93, 1-24.

Lockhart, L.B., Jr., Baus, R.A., Patterson, R.L., Jr. and Saunders, A.W., Jr., 1959. Contamination of the air by radioactivity from the 1958 nuclear tests in the Pacific. Science, 130, 161-162.

Lods, G. and Gouze, P., 2004. WTFM, software for well test analysis in fractured media combining fractional flow with double porosity and leakance approaches. Computers and Geosciences, 30, 937-947.

Love, A.J., Herczeg, A.L., Sampson, L., Cresswell, R.G. and Fifield, L.K., 2000. Sources of chloride and implications for Cl dating of old groundwater, southwestern Great Artesian Basin, Australia. Water Resources Research, 36, 1561-1574.

Low, W., Cassanet, M, Moore, D. and Reilly, T., 2004. Palm Valley Gas Field; environmental impact assessment of the proposed PV 11 well site and access road, November 28 to December 2, 2003. Report by W.A. Low Ecological Services to Magellan Petroleum,

Mcqueen, A.D. and Knott, G.G. 1982. Groundwater in the North Eastern part of the Amadeus Basin. NT Water Resources Branch, Report 82/12, 75 pp.

Machta, L. and List, R.J., 1959. Fallout from nuclear weapons tests. Advances in Geophysics. 6, 273.

Mackie, C.D., 1982. Multi rate testing in fractured formations. In papers of the Groundwater in Fractured Rock Conference 1982, Australian Water Resources Conference Series 5, Australian Government Publishing Service, Canberra, 139-150.

Mackie, C.D., 1989. Aquifer evaluation - 2. In Knight M.J. (Ed) Proceedings of the 10th Australian Water Resources Council Groundwater School, Chapter 16. University of New South Wales, Centre for Groundwater Management and Hydrogeology, Short Course Publication 1/89, 11 pp.

Magaritz, M., Kaufman, A., Michael, P., Boaretto, E. and Hollos, G., 1990. New method to determine regional evapotranspiration. Water Resources Research, 26, 1759 - 1762

Magee, J.W., Bowler, J.M., Miller, G.H. and Williams, D.L.G., 1995. Stratigraphy, sedimentology, chronology and palaeohydrology of Quaternary lacustrine deposits at Madigan Gulf, Lake Eyre, South Australia. Palaeogeography, Palaeoclimatology, Palaeoecology, 113, 3-42.

Magellan 1966, Orange #1, Well completion report, Magellan Petroleum. Mines and Energy, Northern Territory, report PR 66/19.

Magellan, 2003. Magellan petroleum. Well proposal and drilling program, Palm Valley 11.

Mah, A., Taylor G.R., Lennox, P. and Balia, L., 1995. Lineament analysis of Landsat Thematic Mapper images, Northern Territory, Australia. Photogrammetric Engineering and Remote sensing, 61, (6) 761-773.

Mahara, Y., Igarashi, T., Hasegawa, T., Miyakawa, K., Tanaka, Y. and Kiho, K., 2001. Dynamic changes in hydrogeochemical conditions caused by tunnel excavation at the Aspo Hard Rock Laboratory (HRL), Sweden. Applied Geochemistry, 16(3): 291-315.

Matthai, S.K. and Belayneh, M., 2004. Fluid flow partitioning between fractures and a permeable rock matrix. Geophysical Research Letters, 31, L07602.

Masciopinto, C., *In Press*. Pumping-well data for conditioning the realization of the fracture aperture field in groundwater flow models. Journal of Hydrology. 19pp.

Mazor, E., 1991. Applied chemical and isotopic groundwater hydrology. Open University Press, Milton Keynes, U.K., 274 pp.

Mazor, E., 1992. Reinterpretation of ³⁶Cl data: physical processes, hydraulic interconnections and age estimates in groundwater systems. Applied Geochemistry, 7(4): 351-360.

Mazor, E., 1993a. Chlorine-36 data and basic concepts of hydrology – comment on F.M. Phillip's comment, with special reference to the great Artesian Basin. Applied Geochemistry, 8(6): 649-651

Mazor, E., 1993b. Some basic principles of ³⁶Cl hydrology: a reply to the discussion by Kellett, Evans, Allan and Fifield. Applied Geochemistry, 8(6): 659-662

Mazor, E., 1993c. The need for auxiliary data to address the long list of processes recruited to interpret ³⁶Cl data in groundwater systems--a reply to the comment by J.-C. Fontes and J.N. Andrews. Applied Geochemistry, 8(6): 667-669.

Mazor, E., 1995. Stagnant aquifer concept Part 1. Large-scale artesian systems -- Great Artesian Basin, Australia. Journal of Hydrology, 173(1-4): 219-240.

Mazor, E., 1996. Comment on "Hydraulic continuity in large sedimentary basins, by J. Toth". Hydrogeology Journal, 4, 98-102.

Mazor, E., 1997, Chemical and isotopic groundwater hydrology: the applied approach. M. Dekker, New York, 413 p.

Mazor, E. and Nativ, R., 1994. Stagnant groundwater stored in isolated aquifers: implications related to hydraulic calculations and isotopic dating-Reply. Journal of Hydrology, 154(1-4): 409-418.

Meinzner, O.E. and Fishel, V.C., 1934. Tests of permeability with low hydraulic gradients. American Geophysical Union Transactions, 1934, 405-409.

Miller, G.H., Magee, J.W. and Tull, J.T., 1997. Low-latitude glacial cooling in the Southern Hemisphere from amino-acid racemization in emu eggshells. Nature, 385, 241-244.

Milne-Home, W.A., 1988. Interpretation of pumping tests in fractured aquifers: from theory to routine field analysis. In Hitchon, B. and Bachu, S. (Editors) Proceedings, Fourth Canadian/American Conference on Hydrogeology, Banff Alberta Canada 1988, National Water Well Association, Dublin, Ohio, U.S.A., 177-184

Milton, G.M., Andrews, H.R., Causey, S.E., Chant. L.A., Cornett, R.J., Davies, W.G., Greiner, B.F., Koslowsky, V.T., Imahori, Y., Kramer, S.J., McKay, J.W. and Milton, J.C.D., 1994. Clhorine-36 dispersion in the Chalk River area. Nuclear Instruments and Methods in Physics Research Section B-Beam Interactions with Materials and Atoms, 92, 376-379.

Milton, J.C.D., Milton, G.M., Andrews, H.R., Chant. L.A., Cornett, R.J.J., Davies, W.G., Greiner, B.F., Imahori, Y., Koslowsky, V.T., Kotzer, T., Kramer, S.J. and McKay, J.W., 1997. A new interpretation of the distribution of bomb-produced chlorine-36 in the environment, with special reference to the Laurentian Great Lakes. Nuclear Instruments & Methods in Physics Research Section B-Beam Interactions with Materials and Atoms, 123, 382-386.

Milton, G.M., Milton, J.C.D., Schiff, S., Cook, P., Kotzer, T.G. and Cecil, L.D., 2003 Evidence for chlorine recycling - hydrosphere, biosphere, atmosphere - In a forested wet zone on the Canadian Shield. Applied Geochemistry, 18(7): 1027-1042.

Mook, W.G., Bommerson, J.C., Staverman, W.H., 1974. Carbon isotope fractionation between dissolved bicarbonate and gaseous carbon dioxide. Earth and Planetary Science Letters, 22, 169–176.

Moran, J.E. and Rose, T.P., 2003. A chlorine-36 study of regional groundwater flow and vertical transport in southern Nevada. Environmental Geology, 43(5): 592-605.

Morton, S.R., Short, J. and Barker, R.D., 1995. Refugia for biological diversity in arid and semi-arid Australia. Biodiversity Series, Paper No. 4. Department of the Environment, Sport and Territories, Canberra, 171 pp.
Moysey, S., Davis, S.N., Zreda, M. and Cecil, L.D., 2003. The distribution of meteoric ³⁶Cl/Cl in the United States: a comparison of models. Hydrogeology Journal, 11: 615-627.

Murad, A.A. and Krishnamurthy, R.V., 2004. Factors controlling groundwater quality in the Eastern Arab Emirates: a chemical and isotopic approach. Journal of Hydrology, 286, 227-235

Murphy, E.M., Ginn, T.R. and Phillips, J.L., 1996. Geochemical estimates of paleorecharge in the Pasco Basin: evaluation of the chloride mass balance technique. Water Resources Research, 32(9): 2853-2868.

Murray, L.R. and Siebert, B.D., 1962. Nitrate in underground waters of central Australia. The Australian Journal of Science, 125, 22-23.

Nanson, G. C., Price, D.M. and Short, S.A., 1992, Wetting and drying of Australia over the past 300 ka: Geology, 20, p. 791-794.

Nanson, G. C., Chen, X.Y. and Price, D.M., 1995. Aeolian and fluvial evidence of changing climate and wind patterns during the past 100 ka in the western Simpson Desert, Australia. Palaeogeography., Palaeoclimatology, Palaeoecology, 113, 87-102.

Nanson, G.C., Callen, R.G. and Price, D.M., 1998. Hydroclimatic interpretation of Quarternary shorelines on South Australian playas et al. (1998). Palaeogeography, palaeoclimatology, palaeoecology, 144, 281-305

Nastev, M., Savard, M. M., Lapcevic, P., Lefebvre R. and Martel R. 2004. Hydraulic properties and scale effects investigation in regional rock aquifers, south-western Quebec, Canada. Hydrogeology Journal 12(3) 257-269.

Nesbitt, Andrew, 2005. Personal communication, Development officer, Northern Territory Department of Business, Industry and Resource Development, Alice Springs, NT. Nolte, E., Krauthan, P., Korschinek, G., Maloszewski, P., Fritz, P. and Wolf, M., 1991. Measurements and interpretations of ³⁶Cl in groundwater, Milk River aquifer, Alberta, Canada. Applied Geochemistry, 6(4): 435-445.

Norris, A.E., Wolfsberg, K., Gifford, S.K., Bentley, H.W. and Elmore, D., 1987. Infilltration at Yucca Mountain, Nevada, traced by ³⁶Cl. Nuclear Instruments and Methods in Physics Research Section B: Beam Interactions with Materials and Atoms, 29, 376-379.

Novakowski, K.S. and Bickerton, G.S., 1997. Borehole measurement of the hydraulic properties of low-permeability rock. Water Resources Research, 33, 2509-2517.

NT Government bore file RN 16018, Department of Infrastructure, Planning and Environment, Alice Springs, NT

NT Government, Water Resource Branch file No. 60.16 Department of Infrastructure, Planning and Environment, Alice Springs, NT

Neuzil, C.E., 1986. Groundwater flow in low-permeability environments. Water Resources Research, 22, 1163-1195.

Oaks, Jr., R.Q., Deckelman, J.A., Conrad, K.T., Hamp, L.P., Phillips, J.O. and Stewart, A.J., 1991. Sedimentation and tectonics in the northeastern and central Amadeus Basin, central Australia. In Korsch, R.J. and Kennard, J.M. (eds), Geological and geophysical studies of the Amadeus Basin, central Australia. Bureau of Mineral Resources, Australia, Bulletin 236, 73-90.

Ophori, D.U. and Toth, J., 1990. Relationships in regional groundwater discharge to streams: an analysis by numerical simulation. Journal of Hydrology, 119, 215-244.

Orscheg, C.K. and Parsons, R.F. (1996) Ecology of Livistona Australis (R.Br.) Maritus (arecaceae) at its southern limit, south eastern Australia Part 1 distribution and genetic variation. Mooreana, 6, 8-17.

Pacontinental, 1984. Amadeus Basin – Northern Territory, OP's 175, 178. Composite photogeological interpretation and seismic control, Palm Valley 1: 100 000 Sheet. Photogeology by Australian Photogeological consultants for Pancontinental Petroluem Limited.

Park, J., Bethke, C.M., Torgersen, T. and Johnson, T.M., 2002. Transport modelling applied to the interpretation of groundwater 36Cl age. Water Resources Research, 38, (5),1-1 to 1-15 (1043, 10.1029/2001WR000399.).

Parkhurst, D. L. and Appelo, C.A.J., 1999. User's guide to PHREEQC (version2)- A computer program for speciation, batch reaction, one dimensional transport, and inverse geochemical calculations. Water-Resources Investigations Report 99-4259. United States Geological Survey, 312 p.

Pearson. F.J. Jnr. and Swarzenki, V.W., 1974. ¹⁴C evidence for the origin of arid area region groundwater, Northeastern Province Kenya, Proceedings Symposium on Isotope Techniques in Groundwater Hydrology. International Atomic Energy Agency, Vienna, 1, 95-108.

Penny, E., Lee, M.-K. and Morton, C., 2003. Groundwater and microbial processes of Alabama coastal plain aquifers. Water Resources Research, 39, (1320) HWC **2**, 1-17.

Perry, R.A., Mabbutt, J.A., Litchfield, W.H., Quinlan, T., Lazarides, M., Jones, N.O., Slayter, R.O., Stewart, G.A., Bateman, W., and Ryan, G.R., 1962. General report on lands of the Alice Springs area, Northern Territory, 1956-57. Commonwealth Scientific and Industrial Research Organization, Australia, Land Research Series No. 6, 300 p.

Phillips, F.M., Bentley, H.W. and Elmore, D., 1986a. Chlorine-36 dating of old ground water in sedimentary basins. Proceedings of the third Canadian/American conference on hydrogeology, Banff, Alberta, Canada, National Water Well Association, Dublin, Ohio, USA, 143-150.

Phillips, F.M., Bentley, H.W., Davis, S.N., Elmore, D. and Swanick, G.B., 1986b. Chlorine-36 dating of very old groundwater 2. Milk River aquifer, Alberta, Canada. Water Resources Research, 22, 2003-2016.

Phillips, F.M., 1993. Comment on "Reinterpretation of ³⁶Cl data: physical processes, hydraulic interconnections and age estimates in groundwater systems" by E. Mazor. Applied Geochemistry, 8, 643-647.

Phillips, F.M., 2000. Chlorine-36. *In* Cook, P.G. and Herczeg., A.L. (Eds.) Environmental Tracers in Subsurface Hydrology. Kluwer Academic Publishers, Boston. 299 – 348.

Phillips, F.M., Stone, W.D. and Fabryka-Martin, J.T., 2001. An improved approach to calculating low-energy cosmic-ray neutron fluxes near the land/atmosphere interface. Chemical Geology, 175, 689-701.

Pickup, G., Allan, G. and Baker, V.R., 1988. History, palaeochannels and palaeofloods of the Finke River, central Australia. In Warner (ed.), Fluvial geomorphology of Australia, Academic Press, Sydney, 177-200.

Pidsley, D., Kerlie E., Williams, D., Hatton, T. and Myers, B., 1994. Role of transpiration in the recharge-discharge process of an aquifer system underlying open eucalypt forest in wet-dry north Australian tropics. 25th Congress of The International Association of Hydrogeologists / International Hydrology and Water Resources Symposium of The Institution of Engineers, Australia, Adelaide, 21-25 November. The Institute of Engineers, Australia, Preprints of Papers, 491-496.

Plummer, L.N., 1993. Stable isotope enrichment in paleowaters of the southeast Atlantic coastal plain, United States. Science, 262, 2016-2020.

Plummer, M.A., Phillips, F.M., Fabryka-Martin, J., Turin, H.J., Wigand, P.E. and Sharma, P., 1997. Chlorine-36 in fossil rat urine: An archive of cosmogenic nuclide deposition during the past 40,000 years. Science, 277, 538-541.

Purdy, C.B., Mignerey, A.C., Helz, G.R., Drummond, D.D., Kubik, P.W., Elmore, D. and Hemmick, T., 1987. ³⁶Cl: A tracer in groundwater in the Aquia Formation of Southern Maryland. Nuclear Instruments and Methods in Physics Research Section B: Beam Interactions with Materials and Atoms, 29, 372-375.

Purdy, C.B., Helz, G.R., Mignerey, A.C., Kubik, P.W., Elmore, D., Sharma, P. and Hemmick, T., 1996. Aquia aquifer dissolved Cl⁻ and ³⁶Cl/Cl: Implications for flow velocities. Water Resources Research, 32, 1163-1171.

Radke, B.M., Ferguson, J., Cresswell, R.G., Ransley, T.R. and Habermehl, M.A., 2000. Hydrochemistry and implied hydrodynamics of the Cadna-owie-Hooray Aquifer Great Artesian Basin. Bureau of Rural Sciences, Canberra, 229p.

Ranford, L.C. and Mikolajczak, A.S., 1963. (Compilers) Henbury, Northern Territory, SG 53-1, 1:250 000 Sheet Geology Map. Bureau of Mineral Resources, Canberra. First edition.

Ranford, L.C. and Cook, P.J., 1964. The geology of the Henbury 1:250,000 sheet area, Amadeus Basin, Northern Territory. Bureau of Mineral Resources, Record 1964/40. Canberra, 62 p.

Rao, U., Fhen, U., Teng, R.T.D. and Goff, F., 1996. Sources of chloride in hydrothermal fluids from the Valles caldera, New Mexico: a ³⁶Cl study. Journal of Volcanology and Geothermal Research, 72, 59-70.

Rozanski, K., Araguas-Araguas, L., Gonfiantini, R.,1993. Isotopic patterns in modern global precipitation. In: Swart, P.K., Lohmann, K.C., McKenzie, J., Savin, S. (Eds.) Climate change in continental isotopic records. American Geophysical Union Monograph Series, 78, 1-36.

Read, R.E. and Paul, R.J., 2000: Rocky Hill - Ooraminna Groundwater Investigation 1998 and 1999. DIPE, unpublished report, report number 16/1998

Read, R.E. and Paul, R.J., 2000. Rocky Hill – Ooraminna groundwater investigation 2000. NT Water Resources Branch Report # 05/2002 A, 85 p.

Rhew, R.C., Miller, B.R., Weiss, R.F., 2000. Natural methyl bromide and methyl chloride emissions from coastal salt marshes. Nature 403, 292 – 295.

Rhew, R. C., Aydin, M. and Saltzman, E. S., 2003.Measuring terrestrial fluxes of methyl chloride and methyl bromide using a stable isotope tracer technique, Geophysical Research Letters, Vol 30, 2103, ASC 7, 1-5.

Rodd, A.N., 1998. Revision of Livistona (Arecaceae) in Australia. Telopea, 8, 49-153.

Salas, G., 1987. Drilling at six Hermannsburg Outstations, 1987, (Undarana, Ipolera, Antjakwera, Lyiltjera, Kwala and Intjarrtnama). Northern territory Water Resources, report # 29/1987, 17p.

Salomons, W., Mook, W.G., 1986. Isotope geochemistry of carbonates in the weathering zone. In: Fritz, P., Fontes, J.C. (Eds.), Handbook of Environmental Isotope Geochemistry, The Terrestrial Environment, B, vol. 2. Elsevier, Amsterdam, pp. 239–269.

Scanlon, B.R., Kubik, P.W., Sharma, P., Richter, B.C. and Gove, H.E., 1990. Bomb Cl-36 Analysis in the Characterization of Unsaturated Flow at a Proposed Radioactive-Waste Disposal-Facility, Chihuahuan Desert, Texas. Nuclear Instruments & Methods in Physics Research Section B-Beam Interactions with Materials and Atoms, 52(3-4): 489-492.

Schaeffer O.A., Thompson, S.O. and Lark, N.L., 1960. Chlorine-36 radioactivity in rain. Journal of Geophysical Research. 65, 4013-4016.

Scheffel, C., Blinov, A., Massonet, S., Sachsenhauser, H., Stan-Sion, C., Beer, J., Synal, A., Kubik, P.W., Kaba, M. and Nolte, E., 1999. ³⁶Cl in modern atmospheric precipitation. Geophysical Research Letters, 26(10): 1401-1404.

Schoeller, H., 1959. Arid zone hydrology: Recent Developments. United Nations Educational, Scientific and Cultural Organisation, Paris, 125 pp

Schofield, S. and Jankowski, J., 2004. Hydrochemistry and isotopic composition of Na-HCO₃-rich groundwaters from the Ballimore region, central New South Wales, Australia. Chemical Geology 211, 111-134.

Sharma, I.K., Jones, D.L., Forster, P.I., 2004. Genetic differentiation and phenetic relatedness among seven species of the *Macrozamia plurinervia* complex (Zamiaceae). Biochemical Systematics and Ecology 32, 313–327.

Shekaforoush, S.S., 2000. Groundwater chemistry and hydrogeochemical processes in fractured bedrock aquifers beneath dryland salinity occurrences in South Eastern Australia. Unpublished PhD thesis, University of New South Wales, Sydney. 348 p.

Simpson, C.J., Wilford J.R., Macias, L.F. and Korsch, R.J., 1991. Remote sensing of natural hydrocarbon seepage atr the Palm Valley Gas Field, Amadeus Basin, central Australia. *In* Korsch, R.J. and Kennard, J.M. (Editors), Geological and geophysical studies in the Amadeus Basin, central Australia. Bureau of Mineral Resources, Australia, Bulletin 236, 541-562.

Simpson, H.J. and Herczeg, A.L., 1994. Delivery of marine chloride in precipitation and removal by rivers in the Murray-Darling Basin, Australia. Journal of Hydrology, 154(1-4): 323-350.

Singhal, B.B.S. andb Gupta, R.P., 1999. Applied Hydrogeology of Fractured Rocks, Kluwer Academic Publishers, Dordrecht, 400p.

Smith, M.A. 1987. Pleistocene occupation in arid central Australia. Nature, 328, 710-711.

Smith, M.A. 1989. The case for a resident population in the central Australian ranges during full glacial aridity. Archaeology in Oceania, 24, 93-105.

Socki, R.A., Karlsson, H.R. and Gibson, E.K. Jr., 1992. Extraction technique for the determination of oxygen-18 in water using pre-evacuated glass vials. Analytical Chemistry, 64, 829-831.

Songsheng, J., Shan, J., Hong, G., Shubin, D., Zhanru, C., Qifeng, G. and Yunlong, Z., 1994. Determination of ³⁶Cl in the groundwaters and ores around a uranium deposit. . Nuclear Instruments and Methods in Physics Research Section B-Beam Interactions with Materials and Atoms, 92, 385-388.

Stead, G., 1982. Aboriginal Issues, an 18 page appendix in, Dames and Moore, 1982. Environmental assessment and management report, Palm Valley pipeline project.

Stumm, W. amd Morgan, J.J., 1996. Aquatic chemistry. Wiley-interscience, New York (3rd Edition) 1022 pp.

Sturchio, N.C. and Binz, C.M., 1988. Uranium-series age determination of calcite veins, VC-1 drill core, Valles caldera, New Mexico. Journal of Geophysical Research, 93, 6097-6102.

Sturchio, N.C., Du, X., Purtschert, R., Lehmann, B. E., Sultan, M., Patterson, L. J., Lu, Z.-T., Müller, P., Bigler, T., Bailey, K., O'Connor, T. P., Young, L., Lorenzo, R., Becker, R., El Alfy, Z., El Kaliouby, B., Dawood, Y. and Abdallah, A. M. A., 2004. One million year old groundwater in the Sahara revealed by krypton-81 and chlorine-36. Geophysical Research Letters, 31, L05503, 4pp.

Suter, M., Beer, J., Bonani, G., Hoffmann, H.J., Michel, D., Oeschger, H., Synal, H.A. and Wölfli, W., 1987. ³⁶Cl studies at the ETH/SIN-AMS facility. Nuclear Instruments and Methods in Physics Research Section B-Beam Interactions with Materials and Atoms, 29, 211-215.

Synal, H.-A., Beer, J., Bonani, G., Suter, M. and Wölfli, W., 1990. Atmospheric transport of bomb-produced ³⁶Cl. Nuclear Instruments & Methods in Physics Research Section B-Beam Interactions with Materials and Atoms, 52, 438-488.

Tamers, M.A., 1967. Radiocarbon ages of groundwater in an arid zone unconfined aquifer. In: Isotope Techniques in the Hydrological Cycle, American Geophysical Union, Monograph 11, 143-152.

Theis, C.V. 1940. The source of water derived from wells. Civil Engineering, 10, 277 – 280.

Thigpen, J. B., 1973, Fracture analysis Palm Valley, Gardiner and James Range anticlines: Magellan Petroleum. NT Mines and Energy Alice Springs report PR 73/14.

Thorpe, P.M. and Davidson, W.A., 1990. Groundwater age and hydrodynamics of the confined aquifers, Perth, Western Australia. *In* Proceedings of the International Conference on Groundwater in Large Sedimentary Basins, Perth, 1990. Australian Water Resources Conference Series No 20, Australian Government Printer, Canberra, 420-436

Tickell, S.J., 1997. Mapping dryland-salinity hazard, Northern Territory, Australia. Hydrogeology Journal, 5, 109-117.

Todd, D.K., 1959. Groundwater Hydrology. Willey, New York. 336 p

Torgersen, T., Habermehl, M.A., Phillips, F.M., Elmore, D., Kubik, P., Geoffrey-Jones, B., Hemmick, T. and Gove, H.E., 1991. Chlorine 36 dating of very old groundwater 3. Further studies in the Great Artesian Basin, Australia. Water Resources Research, 27(12): 3201-3213.

Torgersen, T. and Phillips, F.M., 1993. Reply (to Andrews, J.N. and Fontes, J.-Ch., 1993). Water Resources Research, 29, 1875-1877.

Toth, J., 1963, A theoretical analysis of groundwater flow in small drainage basins: Journal of Geophysical research, v. 68, p. 4793-4812.

Toth J. 1990. Hydraulic continuity in large sedimentary basins. In Proceedings of the international conference on groundwater in large sedimentary basins. Australian Water

Resources Council conference series no. 20. Australian Government publishing Service., Canberra, 2-15.

Toth, J., 1995. Hydraulic continuity in large sedimentary basins: Hydrogeology Journal, v.3, p. 4-16.

Tric, E., Valet, J.-P., Tucholka, P., Paterne, M., Labeyrie, L., Guichard, F., Tauxe, L. and Fontugne, M., 1991. Paleointensity of the geomagnetic field during the last 80,000 years. Journal of Geophysical Research, 97, 9337-9351.

Troisi, S. and Vurro, M., 1988. Experimental tests to study the hydrodynamic analogies between fissured and porous media. In; Karst Hydrogeology and Karst Environment Protection, Proceedings of the 21st Congress of the International Association of Hydrogeologists, Volume XXI part 2, Gulin, China, 1988, 565-571.

Tsang, Y.W., Tsang, C.F., Hale, F.V. and Dverstorp, B., 1996. Tracer transport in a stochastic continuum model of fractured media. Water Resources Research, 32, 3077-3092

Turner, J.V., Bradd, J.M. and Waite, T.D., 1992. Conjunctive use of isotopic techniques to elucidate solute concentration and flow processes in dryland salinized catchments. Isotope techniques in water resources development 1991. Proceedings symposium, Vienna, 1991, Published by IAEA: 33-59.

Turner, J.V., Rosen, M.R., Fifield, L.K. and Allan, G.L., 1995. Chlorine-36 in hypersaline palaeochannel groundwaters of Western Australia. *In* Applications of Tracers in Arid Zone Hydrology. Proceedings of the Vienna symposium, August, 1994. IAHS Publication No. 232. (International Association of Hydrological Sciences): 15-33.

U.N.E.S.C.O., 1984. Groundwater in hard rocks. Studies and reports in hydrology #33, United Nations Educational, Scientific and Cultural Organisation, Paris, 228 pp.

Vogel., J.C., 1967. Investigations of groundwater flow with radiocarbon. In: Isotopes in hydrology. (Symposium proceedings, Vienna 1966), IAEA, Vienna, 355-369.

Vogt, S., Elmore, D. and Fritz, S.J., 1994. ³⁶Cl in shallow, perched aquifers from central Indiana. Nuclear Instruments & Methods in Physics Research Section B-Beam Interactions with Materials and Atoms, 92, 398-403.

Walker, G.R. and Cook, P.G., 1991. The importance of considering diffusion when using carbon-14 to estimate recharge to an unconfined aquifer. Journal of hydrology, 128, 41-48.

Warren, R.G. and Shaw, R.D., 1995. Hermannsburg, SF 53-13, 1:250 000 geological map series explanatory notes, Northern Territory Department of Mines and Energy and Australian Geological Survey Organisation, 81 p.

Wells, A. T., Forman, D.J., Ranford, L.C. and Cook, P.J., 1970, Geology of the Amadeus Basin, central Australia: Bureau of Mineral Resources Australia, Bulletin 100, 222p.

White, M.E., 1994, After the greening: The browning of Australia: Kangaroo Press, Sydney, Australia, 288 p.

White, P.J. and Broadley, M.R., 2001. Chloride in soils and its uptake and movement within the plant: A review. Annals of Botany, 88, 967-988.

Williams, M.A.J., Dunkerley, D.L., De Decker, P., Kershaw, A.P. and Stokes, T., 1993, Quaternary Environments: Edward Arnold, UK, 329p.

Wischusen, J.D.H., 1994a Sustainability of a Hard Rock Aquifer at Kintore, Gibson Desert, preprints of papers of the 25th World Congress of the International Association of Hydrogeologists (IAH), Adelaide. Institute of Engineers Australia, publication 94/10, 343-349

Wischusen, J.D.H., 1994b. Areyonga bore rehabilitation. Northern Territory Water Resources, Report # 40/1993, 36p.

Wischusen, J.D.H., 1995a. Palm Valley, production bore RN 12024: Bore workover 1994. Northern Territory Water Resources, Report # 05/1995, 10p.

Wischusen, J.D.H., 1995b. Hydrogeology and sustainable yield of a hard rock aquifer at Kintore, Gibson Desert, central Australia. Master of Science thesis, Monash University, Melbourne, 256p.

Wischusen, J.D.H., 1998. Hydrogeology of the Yuendumu – Papunya – Kintore region, Northern Territory. Australian Geological Survey Organisation Record 1998/31, 44 pp.

Wischusen, J.D.H., Fifield, L.K. and Cresswell, R.G., 2004. The hydrogeology of Palm Valley, central Australia; a Pleistocene flora refuge? Journal of Hydrology, 293, 20-46.

Yechieli, Y., Ronen, D. and Kaufman, A., 1996. The source and age of groundwater brines in the Dead Sea area, as deduced from ³⁶Cl and ¹⁴C. Geochimica et Cosmochimica Acta, 60(11): 1909-1916.

Yokouchi, Y., Nojiri, Y., Barrie, L.A., Toom-Sauntry, D., Machida, T., Inuzuka, Y., Akimoto, H., Li, H.-J., Fujinuma, Y., Aoki, S., 2000. A strong source of methyl chloride to the atmosphere from tropical coastal land. Nature 403, 295 – 298.

Zhou, L., Liu, C.-F., Jiang, S. and Gao, S., 2001. A study of ³⁶Cl age in Quaternary groundwater of Hebei plain, China. Science in China, Series E-Technological Sciences, 44, 11-15 supplement.

Zuber, A., Weise, S.M., Osenbruck, K., Pajnowska, H. and Grabczak, J., 2000. Age and recharge pattern of water in the Oligocene of the Mazovian basin (Poland) as indicated by environmental tracers. Journal of Hydrology, 233(1-4): 174-188.

Zuppi, G.M. and Sacchi, E., 2004. Hydrogeology as a climate recorder: Sahara-Sahel (North Africa) and the Po Plain (Northern Italy). Global and Planetary Change, 40, 79-91